

IntechOpen

Emerging Pollutants Some Strategies for the Quality Preservation of Our Environment

Edited by Sonia Soloneski and Marcelo L. Larramendy





EMERGING POLLUTANTS - SOME STRATEGIES FOR THE QUALITY PRESERVATION OF OUR ENVIRONMENT

Edited by Sonia Soloneski and Marcelo L. Larramendy

Emerging Pollutants - Some Strategies for the Quality Preservation of Our Environment

http://dx.doi.org/10.5772/intechopen.71801 Edited by Sonia Soloneski and Marcelo L. Larramendy

Contributors

Agata Matarazzo, Kula Senapati, Cécile Duclairoir Poc, Ségolène Depayras, Nicole Orange, Marc Feuilloley, Hermann J. Heipieper, Tatiana Kondakova, Philiswa Nomngongo, Anele Mpupa, Nilce Ortiz

© The Editor(s) and the Author(s) 2018

The rights of the editor(s) and the author(s) have been asserted in accordance with the Copyright, Designs and Patents Act 1988. All rights to the book as a whole are reserved by INTECHOPEN LIMITED. The book as a whole (compilation) cannot be reproduced, distributed or used for commercial or non-commercial purposes without INTECHOPEN LIMITED's written permission. Enquiries concerning the use of the book should be directed to INTECHOPEN LIMITED rights and permissions department (permissions@intechopen.com). Violations are liable to prosecution under the governing Copyright Law.

CC BY

Individual chapters of this publication are distributed under the terms of the Creative Commons Attribution 3.0 Unported License which permits commercial use, distribution and reproduction of the individual chapters, provided the original author(s) and source publication are appropriately acknowledged. If so indicated, certain images may not be included under the Creative Commons license. In such cases users will need to obtain permission from the license holder to reproduce the material. More details and guidelines concerning content reuse and adaptation can be foundat http://www.intechopen.com/copyright-policy.html.

Notice

Statements and opinions expressed in the chapters are these of the individual contributors and not necessarily those of the editors or publisher. No responsibility is accepted for the accuracy of information contained in the published chapters. The publisher assumes no responsibility for any damage or injury to persons or property arising out of the use of any materials, instructions, methods or ideas contained in the book.

First published in London, United Kingdom, 2018 by IntechOpen eBook (PDF) Published by IntechOpen, 2019 IntechOpen is the global imprint of INTECHOPEN LIMITED, registered in England and Wales, registration number: 11086078, The Shard, 25th floor, 32 London Bridge Street London, SE19SG – United Kingdom Printed in Croatia

British Library Cataloguing-in-Publication Data A catalogue record for this book is available from the British Library

Additional hard and PDF copies can be obtained from orders@intechopen.com

Emerging Pollutants - Some Strategies for the Quality Preservation of Our Environment Edited by Sonia Soloneski and Marcelo L. Larramendy

p. cm.

Print ISBN 978-1-78923-384-1 Online ISBN 978-1-78923-385-8 eBook (PDF) ISBN 978-1-83881-613-1

We are IntechOpen, the world's leading publisher of Open Access books Built by scientists, for scientists

3,550+

112,000+

International authors and editors

115M+

151 Countries delivered to Our authors are among the Top 1%

most cited scientists

12.2%

Contributors from top 500 universities



WEB OF SCIENCE

Selection of our books indexed in the Book Citation Index in Web of Science™ Core Collection (BKCI)

Interested in publishing with us? Contact book.department@intechopen.com

Numbers displayed above are based on latest data collected. For more information visit www.intechopen.com



Meet the editors



Sonia Soloneski received her PhD degree in Natural Sciences and is an Assistant professor of Molecular Cell Biology at the Faculty of Natural Sciences and Museum of La Plata, National University of La Plata, Argentina. She became a member of the National Scientific and Technological Research Council (CONICET) of Argentina in genetic toxicology field. Presently, she is a member

of the Latin American Association of Environmental Mutagenesis, Teratogenesis and Carcinogenesis (ALAMCTA), the Society of Environmental Toxicology and Chemistry, the Argentinean Society of Toxicology (ATA), and the Argentinean Society of Genetics (SAG). She is authored of more than 320 scientific publications in the field, including scientific publications in research papers, reviewed journals and conferences worldwide. She is a referent for on issues related to genetic toxicology, mutagenesis and ecotoxicology field.



Marcelo L. Larramendy, PhD, serves as a Professor of Molecular Cell Biology at the School of Natural Sciences and Museum, National University of La Plata, Argentina. He is appointed as a Senior Researcher of the National Scientific and Technological Research Council of Argentina. He is a former member of the Executive Committee of the Latin American Association of En-

vironmental Mutagenesis, Teratogenesis and Carcinogenesis. He is the author of more than 470 contributions, including scientific publications, research communications, and conferences worldwide. He is a recipient of several national and international awards. Prof. Larramendy is a regular lecturer at the international A. Hollaender Courses organized by the IAEMS and a former guest scientist at NIH (USA) and University of Helsinki (Finland). He is an expert in genetic toxicology and is, or has been, a referee for more than 30 international scientific journals. He is a member of the International Panel of Experts at the International Agency for Research on Cancer (IARC, WHO, Lyon, France) in 2015 for the evaluation of DDT, 2,4-D and lindane. Presently, he is the Head of the Laboratory of Molecular Cytogenetics and Genotoxicology at UNLP.

Contents

Preface XI

Chapter 1	Rough Set Applied to Air Pollution: A New Approach to
	Manage Pollutions in High Risk Rate Industrial Areas 1
	Agata Matarazzo

Chapter 2 The Hidden Face of Nitrogen Oxides Species: From Toxic Effects to Potential Cure? 19 Ségolène Depayras, Tatiana Kondakova, Hermann Josef Heipieper,

Marc GJ Feuilloley, Nicole Orange and Cécile Duclairoir-Poc

Chapter 3 Multivariate-Assisted Solid Phase Extraction Procedure for Simultaneous Preconcentration and Assessment of UV-Filters in Wastewater Prior to UV-Vis Spectrophotometric Determination 45 Anele Mpupa and Philiswa N. Nomngongo

Chapter 4 Study on Magnetic Materials for Removal of Water Pollutants 61 Manoj Sharma, Pankaj Kalita, Kula Kamal Senapati and Ankit Garg

Chapter 5 EPs Antibiotics: Photodecomposition and Biocarbon Adsorption 79 Nilce Ortiz

Preface

Environmental contamination is a serious problem for our society worldwide, which continuously needs new monitoring and implementation of strategies for the preservation of environmental quality on the best plausible mode. Although a wise terminology with no strict definition is given for emerging pollutants, they can be defined as xenobiotic, synthetic or natural. In addition, different microbial communities that are not usually monitored or regulated have the possibility to introduce, by themselves, into the different environments and, then, exert unfavorable ecological and/or human health consequences.

Emerging pollutants are closely associated with multiple man-made activities such as industry, transport, food production and urbanization, which are necessary for the development of our modern society. In several cases, discharge of emerging pollutants into the environment has likely occurred for an extensive period of time, but detection, identification and quantification may not have been performed until new strategies and methodologies for recognition were developed. In other cases, synthesis of new chemicals or novel applications are forms of use for known contaminants, and final disposal of preexisting chemicals, not necessarily toxic, can generate new sources of emerging pollutants in multiple environmental compartments.

Few regulatory agencies have proposed methods for establishing provisional safety levels of emerging pollutants. According to the United Nations Educational, Scientific, and Cultural Organization (UNESCO), these xenobiotics include mainly "...compounds found in pharmaceuticals, personal care products, pesticides, industrial and household products, metals, surfactants, industrial additives and solvents". The US Environment Protection Agency (US EPA) states that an emerging pollutant "... is a chemical or material that is characterized by a perceived, potential, or real threat to human health or the environment or a lack of published health standards....". However, today there is no international consensus on the term "emerging pollutants" and it remains problematic. It is not easy to conclude which type of xenobiotic, belonging to different classes, should or should not be included in the list of these contaminants because they represent a changing reality, dependent on perspective as well as timing. Emerging pollutants are currently not included in international legislation mostly because there is not enough information on toxicity, behavior and ecotoxicological effects to establish threshold values. In general, the chemical structure of emerging pollutants comprises diverse, and origin, and they can be released from either point or diffuse pollution sources, i.e., effluents of wastewater treatment plants from urban or industrial areas and atmospheric deposition or from crop and animal activities. Despite the rate of scientific publications covering the problematic of "emerging pollutants" input into the environment, the frequency of occurrence, the last fate and the deleterious effects, the knowledge of the real effects of this particular type of xenobiotic remains fragmentary.

Although a vast literature is available on emerging pollutants, this book contains important investigations into the diverse chemical hazards encountered in both anthropogenic and natural environments and provides valuable information about the genotoxicity of several xenobiotics that can negatively influence the health of humans and ecosystems.

This book begins with a chapter presenting an overview of the current advantages and disadvantages of the employ of the rough set theory as a mathematical approach employed for data analysis in the field of air pollution management including the main methodological and operational contributions as an integrated approach for climate risk evaluation as well as management when developing air quality management strategies. The second chapter provides a review of data concerning the employment of exogenous and endogenous nitrogen oxide species implicated in a large diversity of vital functions as well as their use in the medical field against pathogens' infection highlighting the versatility of these nitrosative compounds. The third chapter describes a study employing a simple, fast, and inexpensive method combining solid-phase extraction and ultraviolet spectrophotometry for simultaneous preconcentration and determination of benzophenone and sulisobenzone compounds in wastewater samples. The fourth chapter is focused on the removal of pollutants from wastewater samples by employing magnetic nanomaterials, particularly affected by heavy metals and dyestuffs. Lastly, this book includes a final chapter presenting an approach combining solar photodecomposition and biocarbon adsorption for removing a mixture of antibiotics, e.g., enrofloxacin, oxytetracycline, cephalexin and amoxicillin from the polluted surface water.

The editors of *Emerging Pollutants - Some Strategies for the Quality Preservation of Our Environment* are enormously grateful to all the contributing authors for sharing their knowledge and insight into this book project. They have made an extensive effort to arrange the information included in every chapter. This book is designed to provide some strategies for the preservation of our environmental quality focusing on the different categories of environmental pollutants and their negative consequences on living organisms. The contributions made by the specialists in this field of research are gratefully acknowledged. We hope that the information presented in this book will continue to meet the expectations and needs of all those interested in the different aspects of the ecotoxicology field. The publication of this book is of high importance for those scientists, chemists, technologists, as well as engineers belonging to academia, governmental and nongovernmental institutions and environmental entities, who make use of these different investigations to understand both basic and applied aspects of how the known and newly emerging pollutants can affect our society and to guide them in the future investigations.

Sonia Soloneski, PhD, and Marcelo L. Larramendy, PhD

School of Natural Sciences and Museum National University of La Plata La Plata, Argentina

Rough Set Applied to Air Pollution: A New Approach to Manage Pollutions in High Risk Rate Industrial Areas

Agata Matarazzo

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/intechopen.75630

Abstract

This study presents a rough set application, using together the ideas of classical rough set approach, based on the indiscernibility relation and the dominance-based rough set approach (DRSA), to air micro-pollution management in an industrial site with a high environmental risk rate, such as the industrial area of Syracuse, located in the South of Italy (Sicily). This new data analysis tool has been applied to different decision problems in various fields with considerable success, since it is able to deal both with quantitative and with qualitative data and the results are expressed in terms of decision rules understandable by the decision-maker. In this chapter, some issue related to multi-attribute sorting (i.e. preference-ordered classification) of air pollution risk is presented, considering some meteorological variables, both qualitative and quantitative as attributes, and criteria describing the different objects (pollution occurrences) to be classified, that is, different levels of sulfur oxides (SOx), nitrogen oxides (NOx), and methane (CH_{a}) as pollution indicators. The most significant results obtained from this particular application are presented and discussed: examples of 'if, ... then' decision rules, attribute relevance as output of the data analysis also in terms of exchangeable or indispensable attributes/criteria, of qualitative substitution effect and interaction between them.

Keywords: industrial areas, air pollution, meteorological attributes, rough set approach

1. Introduction

Air pollution in a region depends mainly on the emission of pollutants and on local meteorological conditions. The probability of air pollution occurrences may be estimated by simple

IntechOpen

© 2018 The Author(s). Licensee IntechOpen. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.

atmospheric dispersion models with proper meteorological data and predefined typical air pollution sources [1, 2].

A lot of studies, however, do not give enough information about the possible relationships between sampling and meteorological parameters, as well as their optimal correspondence formal tools in order to enable modeling and determination of patterns which are characteristic of the investigated area. Proposing conceptual models enables decision-makers at many levels to assess and manage air quality as a whole, rather than on a pollutant-by-pollutant concentration. By developing a holistic approach to air quality, it is possible to evaluate its extensive benefits of more effective developments in existing air-quality features, thereby avoiding the growth of air-quality ceilings, and to consider air quality within its wider meteorological context [3, 4]. By establishing why and when air pollution occasions may occur across a region, strategies should be designed and implemented so as to deal with such episodes. The possibility of forecasting pollutant concentration near the ground with high spatial detail offers the opportunity of constantly monitoring and managing the territory. Air-quality modeling procedures can forecast the behavior and the effects of the substances emitted from identified sources, particularly using data from meteorological instruments. These models can supply the distribution of pollutant concentrations on the ground, and are used for thermoelectric power plant management, being very useful in the case of exceptional events, such as when a highly dangerous pollutant escapes [5].

This study analyses the main relationships between air micro-pollution and meteorological conditions of the area surrounding Siracusa, a city located in Sicily. This was done by measuring air samples from a receiving station near a small town called Melilli, a Sicilian industrial area with a high environmental risk rate [6, 7].

This station has been chosen because it allows the production of a complete picture with respect to the amount of micro-pollution data and meteorological variables descriptions [8]. Then the most reliable parameters for the phenomena of the dispersion of micro-pollutants were identified and also the various critical scenarios were checked, so that all available air pollution sources were considered [9–11]. In particular, a specially designed model, with forecasting abilities of air pollution, has been developed, working independently from the knowledge of the local sources [12]. This monitoring model uses temperature and wind vertical profiles, measured by Radar Analysis Support System (RASS, a radar manufacturer-independent system for evaluating the different elements of a radar by connecting to signals) and SOnic Detection And Ranging (SODAR, a meteorological instrument used as a wind profiler to measure the scattering of sound waves by atmospheric turbulence) and concentration data from ground stations. The local values are correlated with the characteristics of the thermal profile and the direction and intensity of the wind at a selected altitude. On the basis of stored and statistically analyzed data, the model is able to forecast the pollution in the area surrounding the ground station [13] and to give useful information about the management of its main sources.

From the methodological point of view, the proposed approach is in the framework of multicriteria decision analysis, where a lot of different points of views, often conflicting one other, are explicitly considered together to support effective decisions. The utility or, better, the necessity of a multicriteria evaluation in public policies has been recently underlined by Munda [14]. The novel method of data analysis applied to the study of air micro-pollution management, the rough set approach (RSA), considers objects described by a lot of both qualitative and/ or quantitative attributes and criteria (that is their 'profile'). In this context, inconsistencies between descriptions and risk classes assignments need not be removed prior to the analysis, therefore giving useful information about the quality of the inferred decision rules; moreover, the RSA also allows for highlighting the attributes which most contribute to air pollution among those taken into account for the assessment, giving too some useful information about the management of pollution.

Furthermore, this method is able to identify redundant attributes. This concerns the elimination of superfluous data from the data table, without deteriorating the quality of the results, that is, obtaining the same information of that inferred from the original table, therefore permitting enormous savings in data collection. Additionally, the rough set theory also shows a posteriori the relative importance of the considered attributes and criteria, without requiring a priori any elicitation or assessment of technical parameters (such as importance weights, trade-off, etc.), which are often very difficult to provide and never easily understandable by decision-makers.

The results hereby obtained are just an example of the RSA application, in order to understand how and why it is possible to apply this approach to environmental problems.

This chapter contains other five sections; Section 2 explains the basic principles of rough set theory and its main methodological features; Section 3 shows air micro-pollution analyzed data; Section 4 presents the main decision rules obtained; Section 5 discusses the interpretations of the results from the methodological and operational points of view; and lastly, Section 6 concludes this chapter.

2. The rough set theory

The rough set theory (RST), introduced by Pawlak [15–17], has proved to be an excellent tool for data analysis, even in the presence of inconsistencies and ambiguities. The main idea of the RSA is that every object in the universe U (data to be analyzed) is associated a certain amount of information (data, knowledge), expressed by means of some attributes used for their description (e.g. if the objects are air pollution observed by monitoring stations, attributes may be air temperature, the relative humidity index, direction and wind speed, quantities of some micro-pollutants, etc.). Objects having the same description [18] in terms of these attributes are called indiscernible (similar); the indiscernible objects, called elementary sets or granules of knowledge, which therefore result in information granulation. If set U is divided in some classes, objects indiscernible should belong to the same class to be consistent with the indiscernibility principle.

From the universe U, any subset X can be expressed either precisely (as a union of elementary sets) or approximately. In the latter case, the subset X may be characterized by two ordinary sets, called the lower and upper approximations. The lower approximation of X is composed

of all the elementary sets included in X (whose elements, therefore, certainly belong to X), while the upper approximation of X consists of all the elementary sets which have a nonempty intersection with X (whose elements, therefore, may belong to X). A rough set is defined by means of these two approximations, which coincide in the case of an ordinary set. The difference between the two approximations represents the boundary region, whose elements cannot be characterized with certainty as belonging or not to X. The information about objects from the boundary region is, therefore, inconsistent or ambiguous.

The original RSA based on the indiscernibility relation (usually called classical rough set approach) is not, however, able to deal with preference ordered attribute domains (so-called criteria) and preference ordered decision classes (sorting problem), very often crucial for application to real problems in the field of multicriteria decision analysis.

To be able to deal with criteria and ordered decision classes, Greco et al. [19-25] have proposed an extension of the original rough set theory, called dominance-based rough set approach (DRSA), mainly based on the substitution of the indiscernibility relation by a dominance relation: object a dominates object b, if and only if a is at least as good as b with respect to all considered conditional criteria. In a similar way, the decision attribute *d* makes a partition of U into a finite number of preference ordered classes, $Cl = \{Clt, t = 1, ..., n\}$, each $x \in U$ belonging to one and only one class, Clt∈Cl. We can therefore state a basic consistency principle with respect to the dominance relation: if object a dominates object b with respect to a set of criteria and *b* belong to class Clt, *a* should belong at least to class Clt (upward union of Clt). Otherwise, there is an inconsistency with respect to the dominance principle. Therefore, xbelongs to the lower approximation of any subset X of U if all objects dominating x belong to at least the same class of x, that is, x belongs to Clt or better without any ambiguity; x belongs to the upper approximation of X if among the objects dominated by x there is at least an object y belonging to Clt or better. In a similar way, it is possible to define lower and upper approximation of downward union of classes. Also in DRSA, the difference between the two approximations represents the boundary region.

The objects from U can be split into some decision classes by decisional criterion d, obtaining a decision table (DT), where each object x is described using some independent variables, called conditional attributes/criteria, and each object is assigned to a class of this partition, considered as a dependent variable. The quality of classification expresses the ratio between the objects which have been correctly classified and the total number of the elements of the DT, it lies between 0 (any object is not correctly classified) and 1 (all the objects of the universe are correctly classified), and therefore it can measure the goodness of the classification.

Besides, the classification quality may be unaltered if certain conditioned attributes are eliminated because they are superfluous. The minimal sets of the attributes which maintain the same classification quality of the entire table are called reducts. The intersection among all the reducts generates the core (the set of the most important attributes, which consequently cannot be eliminated without deteriorating the quality of the classification). Therefore, the attributes belonging to the core are indispensable, while the attributes belonging to the reducts are exchangeable with one another; the others are actually superfluous. The relations existing among conditional attributes/criteria and decisional classes in the multicriteria sorting problem are expressed by decision rules. These are logical statements of the type 'if..., then...', where the antecedent (condition part) specifies values assumed by one or more condition attributes/criteria and the consequence specifies an assignment to one or more decision classes. If there is only one possible consequence, then the rule is said to be certain, otherwise, it is said to be approximate or ambiguous. An object $x \in U$ supports decision rule r if its description is matching both the condition part and the decision part of the rule; certain rules are supported only by objects from the lower approximation of the corresponding decision classes.

Procedures for the generation of decision rules from a decision are complex tasks, and a number of procedures have been proposed to solve it [18, 25–27].

The existing induction algorithms use one of the following strategies:

- the generation of a minimal set of rules covering all objects from a decision table;
- the generation of an exhaustive set of rules consisting of all possible rules for a decision table; and
- the generation of a set of 'strong' decision rules, even partly discriminant, covering relatively many objects from the decision table (but not necessarily all of them).

In this chapter to infer the rules, the jMAF software has been used, that is available for free in the Internet: RSES – Rough Set Exploration System, http://logic.mimuw.edu.pl/~rses, ROSE – ROugh Set data Explorer http://idss.cs.put.poznan.pl/site/rose.html, –jMAF, java Multicriteria and Multi-attribute Analysis Framework http://www.cs.put.poznan.pl/jblaszczynski/ Site/jRS.html, and jRank – ranking generator using Dominance-based Rough Set Approach http://www.cs.put.poznan.pl/mszelag/Software/jRank/jRank.html [28].

The rules inferred by DRSA can use also the 'at least' and 'at most' terms in their conditional and decisional parts. All these rules are expressed in a natural language, simple to understand the studied phenomenon and for decision support [22]. This means that the proposed approach actually is also able to explain the reasons of a particular pollution situation, moreover showing the real examples of these (traceability of decisions), and is able to support the management in preventing pollution damages, presenting them the situations where some critical events are most probable. Moreover, parameters like the support (the number of the objects which satisfy both the conditional part and the decisional part of the rule) and the confidence (the ratio between support and the number of the objects which satisfy only the conditional part of the rule, expressed in percentage) help the decision-maker in their choice of the most relevant rules.

We can summarize the main characteristics of the rough set approach as follows. With respect to input information (object description), both quantitative and qualitative data can be considered, even if they present some inconsistencies. With reference to output, information about the relevance of attributes and the quality of approximation can be acquired, and the final results are expressed in the form of 'if..., then...' decision rules, which are sentences that decision-makers find easier to understand [29–31] and using only the most relevant attributes/criteria (i.e. some reduct).

In the case of air pollution problem at hand, for example, we can consider some different decision classes of pollution according to an increasing level of some micro-pollutants (SOx, NOx,...). Since some meteorological variables (conditional attributes/criteria) present a monotonic relationship with the degree of pollution (e.g. the air temperature, the degree of humidity) and other no (e.g. wind direction, etc.), it is very important from both the operational and methodological points of view to take into consideration and to exploit in the appropriate way in the description of the objects and in the rule induction attributes and criteria distinctly. Therefore, we have to consider the indiscernibility relation with respect to the former, the dominance relation with respect to the latter, and the assignment to ordered classes with respect to the decision.

Greco et al. [26] proposed an approach for this kind of real-life multicriteria problems. This can be easily modeled by introducing some appropriate thresholds to discretize the conditional attributes and to characterize different levels of air pollution, for the decision classes. No discretization is required with respect to criteria, using the DRSA.

Consequently, the rough sets could be very efficiently applied in the case of uncertainty derived from the granularity of information. Actually, granules of condition attributes/criteria (objects having the same descriptions or respectively belonging to the same dominating/dominated sets) are used to approximate granules of decision (assignment to some decision classes).

The RSA is therefore very different with respect to the fuzzy sets, where the linguistic imprecision due to the use of natural language is mainly considered, and the membership function aims at indicating in what degree each object belongs to a particular class. Of course, the two approaches are not mutually exclusive, but they can actually be used in a complementary way [32–34]. Using a terminology from image representation, we could say that rough sets are related to the number of pixels of an image (its resolution), while the fuzzy sets represent the number of gray levels between black and white. At an operational level, the implementation of fuzzy sets always requires the definition and specification of particular membership functions, one for each attribute, not easy to specify analytically. Therefore, both classical rough set approach and fuzzy sets are sensitive to the specification of these values and both interesting and useful sensitivity and robustness analysis are actually useful and recommended by moving the level of the thresholds and other parameters [30, 35, 36]. It is not the case of DRSA, where actually no parameter should be elicited, but only some example of decision (from the past experiences of from expert knowledge) is needed to model the preference of the decision-maker.

3. Data description

Air micro-pollution-analyzed data come from an air monitoring network, working since 1975, covering an industrial area of 500 km², including the towns of Priolo–Melilli–Siracusa, situated in the province of Siracusa, in the region of Sicily. This industrial area was declared 'a

high environmental risk rate place' by the Law 349/86 and covers six surrounding towns (Augusta, Priolo, Melilli, Siracusa, Floridia, Solarino); the landscape is very varied and is formed by sandy hills, mountains, and plains near the coast [37, 38].

In this territory, a lot of chemical plants, energy production industries, and oil refineries are found, as well as members of a private organization, the industrial trust for environmental safety (CIPA, Consorzio Industriale Protezione Ambiente–Environmental Protection Industrial Consortium). In its operative center, CIPA assembles and works out different micro-pollution parameters and various meteorological variables, measured by 12 different monitoring stations. Data collection and processing is useful in statistical analysis and in upgrading air pollution management in order to avoid the air-tested-exceeding threshold qualities, previously established [38–42].

This chapter studies monitoring station in Melilli only, because in this place, data concerning air micro-pollution quantities and weather conditions, present at the moment of pollution sample construction, are thoroughly collected. In fact, in Melilli, monitoring station hourly quantities of some micro-pollutants, such as sulfur oxide, nitric oxide, non-methanic hydro-carbon, ozone, sulfonyl hydrogen, and different meteorological conditions present at the moment of their observations, such as air temperature, relative humidity index, wind direction, and speed are observed and stored. Some previous studies show the evident correlation between these environmental variables and the quantity of air micro-pollution found in the samples. Because of the complete data present in the samples studied, levels of four micro-pollutants (SOx, CH_4 , NMHC, NOx) in correlation with the meteorological variables previously mentioned [42, 43] are analyzed in this chapter.

Data recovered from the Melilli monitoring station during 2 weeks, more precisely 1 week in January and 1 in August 2010, have been studied, in order to observe differences of analysis results also on the basis of the different seasons of the year. Daily available recorded 'objects' described both by meteorological variables (condition attributes) and by micro-pollution quantity (decision attribute) have been considered. More than 1000 data records have been analyzed, as an example to which the RSA could be applied.

The selected condition attributes/criteria (descriptors) considered in this analysis are the hour of observation (attribute), wind speed (criterion) and wind direction (attribute), air temperature (criterion), and the relative humidity index (criterion), whereas the levels of the aforementioned micro-pollutants are the decision classes. The descriptors have been chosen because in previous studies [36, 42] they looked like some very important factors, at a local level, influencing air micro-pollution quantity.

4. Results

In spite of the fact that data samples used are restricted to a relatively short period of time (each one only 2 weeks), their analysis allowed us to obtain some interesting results, both from methodological and from operational points of view, which give an idea of the knowledge extraction (in terms of decision rules) from available data using the considered approach and the possibility to use this new method to improve air pollution management. As mentioned before, the final results are expressed in the form of 'if..., then...' decision rules, using at any time a particular (relevant) subset of attributes (reducts), according to the season and the micro-pollutant considered at each time.

In the following sections of this chapter, just some examples of decision rules obtained in our study are presented, useful for understanding and describing concisely pollution effects caused by particular combinations of conditional attribute/criteria values. Such rules, as mentioned before, are very useful in explaining the main reasons of some particular pollution events and can be also used in forecast analysis and for decision support too. The rules were chosen as the most representative among those with the highest degree of confidence, indicating the relative frequency of antecedent ('if' part of the rule) also matching the consequent ('then' part of the same rule) of the considered rule. Apart from the analysis of SOx, $CH_{4'}$ and NMHC with respect to the January observations (**Tables 1–4**), the considered rules have a confidence equal to one, that means that all the objects match both the antecedent and the consequent in each rule [37, 41, 43].

These decision rules are presented in the form of tables which are very easy to read, showing in the first column the number of the rule and in the other columns the values of the conditional attributes/criteria characterizing that rule. These values are expressed as intervals with respect to attributes (corresponding to the partition of their domain) and as real numbers ('vertices' of dominance cones) with respect to criteria. The last columns of **Tables 1–3** display the confidence of each rule; in the other tables, the confidence of the rules is one. In particular, **Tables 1–4** show results from Melilli Monitoring Station during the month of January, and **Tables 5–7** show the results in Melilli Monitoring Station during the month of August. Each table concerns a different micro-pollutant.

The threshold interval values for the conditional attribute were chosen as following: hour: 0, 1, ... 23 and wind direction: N (North), S (South), E (East), W (West), as main direction $\pm 45^{\circ}$. On the contrary, the criteria values were automatically determined by the method applied for

Rule	Hour	Air temperature	Humidity	Wind speed	Wind direction	Wind speed and direction	Confidence
1		≥11.55					0.54
2					Е		0.54
3						81.9–137.7	0.46
4	13–16				Е		0.58
5			≥52.3		Е		0.58
6			≥52.3			81.9–117.9	0.58
7					E	81.9–117.9	0.58

Table 1. Melilli monitoring station, January 2010 SOx; threshold = 70 µg/Nm³.

Rough Set Applied to Air Pollution: A New Approach to Manage Pollutions in High Risk Rate... 9 http://dx.doi.org/10.5772/intechopen.75630

Rule	Hour	Air temperature	Humidity	Wind speed	Wind direction	Wind speed and direction	Confidence
8	13–17	≥9.5					0.21
9	14–17				Ν		0.21
10	13–17					134.2–190.2	0.21

Table 2. Melilli monitoring station, January 2010 CH₄; threshold = 950 µg/Nm³.

Rule	Hour	Air temperature	Humidity	Wind speed	Wind direction	Wind speed and direction	Confidence
11		≥11					0.36
12					Е		0.40
13						107.3–149.7	0.37
14					Е	107.3–149.7	0.42

Table 3. Melilli monitoring station, January 2010 NMHC, threshold = 90 µg/Nm³.

Rule	Hour	Air temperature	Humidity	Wind speed	Wind direction	Wind speed and direction
15					Е	
16						66.4–111.5
17	11–14	≥10.5				
18	10–21		≥75.5			
19		≥9.7				18.8–117.9
20			≥49.6		Е	
21			≥ 53.8			67.3–111.5
22				≤3.4	Е	

Table 4. Melilli monitoring station, January 2010 NOx, threshold = $20 \mu g/\text{Nm}^3$, confidence = 1.

them (DRSA). Since the dominance approach is also used for the decisional attribute, pollution is reached 'by definition' depending whether or not the observed value of micro-pollutant is at least equal to the threshold value defined by law and indicated in each table. They are usually some threshold values that it is not allowed to exceed more than three times a year.

In each table, the conditional attributes/criteria are the following: Attribute 1: hour of observation; Criterion 1: air temperature (°C); Criterion 2: air relative humidity index (%); Criterion 3: wind speed (m/s); Attribute 2: wind direction; Attribute 6: wind direction (degrees) with respect to wind speed (measured by SODAR).

Rule	Hour	Air temperature	Humidity	Wind speed	Wind direction	Wind speed and direction	Risk Class
23			≥72.8				В-С
24				≤4.8	Е		А
25		≥31.3	≥54	≤3.9			А
26			≥37.2	≥3.3			В-С
27			≥33	≥3.7			В-С
28			≥37		Ν		С
29			≥31.9		Ν		С

 Table 5. Melilli monitoring station, august 2010 SOx, confidence = 1.

Rule	Hour	Air temperature	Humidity	Wind speed	Wind direction	Wind speed and direction
30	14–20		≥37.2			
31	6–14					132.5–269.1
32		≥26.3	≥37.1			
33		≥25.9	≥37.6, ≤42.9			

Table 6. Melilli monitoring station, August 2010 CH₄ threshold =845 μ g/Nm³, confidence = 1.

Rule	Hour	Air temperature	Humidity	Wind speed	Wind direction	Wind speed and direction
34					Е	
35						88.9–103.8
36				≥3.7	Е	
37					Е	90.5–103

Table 7. Melilli monitoring station, August 2010 NOx, threshold = $20 \mu g/\text{Nm}^3$, confidence = 1.

With respect to the decision, in **Table 5**, we consider three decisional classes, coded as following according to the law: A (Emergency), B (Alarm), C (Alert). The corresponding decision rules are expressed in terms of 'at least (\geq)' or 'at most (\leq)'; therefore, for example, B–C means 'at most Alarm' and A means '(at least) Emergency'. In all the other tables, the decision rules indicate if the considered threshold values are overtaken or not.

The rules in all the above tables represent only a few of several rules obtained by applying the RSA and they are presented here just as examples of easily understandable samples of the results of this analysis. All these rules are of the type 'At least' with respect to the decision, in the sense that if the antecedence is verified, the level of the corresponding micro-pollution is greater than the threshold value. We observe that the selected rules involve in the conditional part only few attributes/criteria each time.

In the following lines, some examples of how reading the decisional rules are presented from **Tables 1–7**. Rule 18 (**Table 4**): between hours 10.00 and 21.00, if the air relative humidity index is at least 75.5%, then the NOx is at least 20 μ g/Nm³, with a confidence of 1. Rule 33 (**Table 6**): if the air temperature is at least 25.9°C and air relative humidity index lies in the interval 37.6–42.9%, then the CH₄ level is at least 845 μ g/Nm³ with a confidence of 1.

5. Discussions

The decision rules concerning CH_4 and NMHC of January 2010 (**Tables 2** and **3**) have a very low confidence level; this means that the considered attributes are not sufficient to explain the phenomenon. Perhaps some attributes are missing and therefore, in order to improve this result, it would be useful to consider further attributes. This is another important methodological feature of the rough set approach, underlining that sometimes more information is needed to better describe some object in order to be able to arrive at well-founded conclusions, that is with a high degree of confidence. On the other hand, the same analysis regarding the observations of August gives very interesting results; we can observe the particular relevance of the degree of humidity in the CH_4 level (**Table 6**) and the crucial role of the wind direction and speed in NOx analysis (**Table 7**). We can also observe that sometimes (e.g. rules 12, 15, 23) it is possible to explain a result using only one attribute, that is with very short and simple decision rules. It should be remembered that a general property of the rough set approach is one that uses all conditional attributes; instead of only attributes from a reduct, we can obtain more concise rules, that is with a greater variety a fewer number of attributes in the conditional part of the rules.

With respect to SOx, we used a greater value for the threshold in January than in August both in order to present the relative pollution level more clearly and to obtain an acceptable confidence degree for same decision rules. Both the analyses of NOx (**Tables 4** and **7**) give excellent results in terms of confidence with respect to all the decision rules obtained, where we can observe a greater relevance of the attributes hour of observation, air temperature, and humidity degree in the analysis of the January data, while a crucial role in the August results is played by winds.

Actually, a first idea about the relevance of the conditional attributes can be directly revealed by the presence frequency of each conditional attribute in the decisional rules, as shown in the **Tables 1–7** (see, e.g. how important the conditional attribute for air relative humidity is in **Table 5**). Some more sophisticatedly important indices can also be computed, for example, according to the Shapley value in the cooperative games in the framework of game theory; the main idea is to compute the contribution to the quality of results by adding another attribute/criterion in the conditional part of the rules, in other words, a degree of the involvement of each attribute in all coalitions of attributes, measuring therefore also the interaction (synergy or redundancy) between the considered attributes [20] (Greco, S. et al., 2001 b). It should be observed that this kind of importance is therefore an output of the analysis within the rough set approach, and not an input information, as usually happens when we use other approaches, as, for example, weighted sum or outranking methods for the comparative evaluation of some objects.

Moreover, the results also show interesting interpretations in terms of a particular kind of trade-off. From the analysis of the couples of decision rules (26,27) (32,33) we can easily observe some cases of trade-off between the values of the couple of attributes in the classical meaning of 'compensation'. From rules (32,33), there is a relationship between air temperature and degree of humidity, in the sense that the different values of air temperature can be compensated by the different degrees of humidity obtaining the same results in terms of pollution. A similar relationship can be observed in the pair of decision rules (26,27) with respect to the degree of humidity and wind speed. This means that a certain capacity of compensation is allowed (trade-off) between the performances of a couple of attributes: a better value on one attribute is able to compensate the worst value on the other and vice-versa.

By observing the following couples of rules (5,6) (9,10) (20,21), we can see that it is possible to obtain the same results in terms of level of pollution considering the combination of one (fixed) attribute/criterion and step by step another one associated with it (example of exchangeable attributes/criteria). In other words, the same decision could be described and explained by different rules, where at each time are present different combinations (in this case, couple) of attributes/criteria that, therefore, are able to describe the same phenomenon independently one another. So, for example, from the couple of rules (5,6), it can be observed that the same result in terms of level of pollution, with the same degree of confidence, is the consequence of the degree of humidity associated with the wind speed and wind direction between 81.9 and 117.9 (rule 6).

Another similar observation can be made comparing rules 6 and 7, where again the phenomenon of exchangeable attributes can be observed that in this case are the air humidity and the wind direction. This means that using the RSA the same effect in the pollution class assignment can be obtained as a result of a combination of an attribute/criterion value each time with other different attributes/criteria, as a particular very interesting 'qualitative substitution effect' between different attributes/criteria. The exchangeable role played independently by some conditional attributes/criteria in combination with a given level of another conditional attribute/criterion (in the previous example, the degree of humidity or the wind direction and speed) results therefore in the assignment of an object to the same decision class of pollution.

With respect to the operational aspect of this approach, it is important to emphasize how obtained results can be used to capably support the decision-maker to manage the pollution risk. Actually, the information given by decision rules can help to understand the main reasons of a pollution event, giving us the explanation of this (its 'traceability') but also for preventing or forecasting dangerous situations, very probable when meteorological conditions similar to those described by the obtained decision rules are approaching (air temperature, humidity degree, wind direction, ...).

Another very interesting result using this approach concerns the information we can receive by so-called non-activated rules in improving or in deteriorating the results of a decision. See, for example, rules from **Table 5** and at levels of air relative humidity index. It can be observed that if this value is smaller than 31.9, the SOx will never be at a level higher than the threshold of 10/gr/Nm³. These rules, therefore, are able to give us useful information about 'critical values' of the conditional criterion air of relative humidity.

More generally, we can say that using this approach we are able to detect some threshold values of one or more condition criteria that can be considered as boundary values to be reached or to be avoided and the combination of two or more attributes/criteria that can be really dangerous for the air pollution. Of course, the meteorological variables cannot be changed by decision-makers. But the rules inferred using the rough set approach can be actually used as guidelines for forecasting in some areas particular cases of pollution events (e.g. emergency, alarm, alert), consequently giving people useful information and suggestions concerning the probable danger of air pollution.

6. Conclusions

The aim of this chapter is to give a first idea of the possibilities offered by rough sets data analysis in the field of air pollution management. In the following, we summarize its main methodological and operational contributions of this exemplary application.

From the methodological point of view, the RSA allows us to take into consideration quantitative and qualitative data, without being in need of their arbitrary transformations.

The relevance of each subset of attributes/criteria is an output of the analysis, and not an input, and therefore does not require elicitation of a priori subjective weights.

It is possible to underline the role of each attribute/criterion in terms of reducts and core; the attributes belonging to the core are indispensable, while the attributes belonging to the reducts are exchangeable with one another; the others are actually superfluous or redundant.

The significance of the results can be measured by peculiar indicators (quality, strength, support, confidence, etc.).

The results are presented in the form of 'if... then...' logical statements, decision rules expressed in simple language and very understandable for the decision-makers.

It is not necessary to remove a priori some inconsistency in the data to be analyzed, but—on the contrary—also these inconsistencies are an important piece of information about the degree of uncertainty of the decision rules inferred (certain or approximate rules, degree of confidence, etc.).

From the operational point of view, the decision rules inferred can be used immediately for managerial purposes as guidelines for preventing or warning people about the risk of air pollution (emergency, alarm, or alert situations), when the weather conditions match or are similar to those shown on the tables and to other rules not included in this chapter.

The decision rules are also able to explain the reasons of particular pollution occurrences, describing the consequences of different meteorological scenarios and their giving a traceability of possible decisions. Moreover, the results obtained point out other relevant profiles of the phenomenon considered. They clearly show, for example, the more or less important role played by each meteorological variable in the assignment of the actions to different pollution classes, the fundamental relationships between the antecedent (attributes and conditional criteria), and the consequent (ordered decision class). Furthermore, they provide interesting information about the semantic importance of quantitative and qualitative trade-offs between attribute/criteria, that is the role of combination of different levels and/or different pollutants considered together, showing therefore also the main interaction among some meteorological factors. Finally, using the RSA it is also possible to detect some particularly interesting threshold values, of one or more condition criteria, that can be considered as boundary values to be reached or to be avoided.

The decision rules, in fact, could be the basis for the development of air-quality management strategies under the impacts of climate change, that is fundamentally a risk valuation and risk management process involving priority assessment of the impacts of climate change and associated uncertainties, including determination of air-quality targets, the selection of potential management options, and identification of effective air-quality management strategies through decision-support models.

The simple application of the method presented in this chapter shows how it can effectively help decision-makers in making appropriate responses to climate change, since it provides an integrated approach for climate risk assessment and management when developing air-quality management strategies. The risk-based decision-making framework can also be applied to develop climate-responsive management strategies for the other environmental dimensions and appraise costs and benefits of future environmental management policies.

Like any study, this could be improved and a more in-depth study can be carried out. For example, the original database could be enlarged, both in time limits and with reference to the variables considered. If we take into consideration data concerning different years or places, and analyze them by using the same methodology, we can, for instance, eliminate the peculiar effects related occasionally to atypical weather conditions. Moreover, if we extend the analysis to other meteorological variables we could obtain decisional rules which are sometimes easier, more intuitive, and more precise than those obtained by using a smaller number of descriptors.

Author details

Agata Matarazzo

Address all correspondence to: amatara@unict.it

Department of Economics and Business, University of Catania, Catania, Italy

References

 Chen KS, Ho YT, Lai CH, Chou Y-M. Photochemical modeling and analysis of meteorological parameters during ozone episodes in Kaohsiung, Taiwan. Atmospheric Environment. 2003;37:1811-1823

- [2] Vukovich FM, Sherwell J. An examination of the relationship between certain meteorological parameters and surface ozone variations in the Baltimore-Washington corridor. Atmospheric Environment. 2003;**37**:971-981
- [3] Bojkov RD. Surface ozone during the second half of the nineteenth century. Journal of Applied Meteorology and Climatology. 1986;25:343-352
- [4] Premec K. Ultraviolet solar radiation in the northern Adriatic. Croatian Meteorological Journal. 2002;**3**:79-88
- [5] Xue M, Droegemeir V, Wong V. The Advanced Regional Prediction System (ARPS)—A multi-scale nonhydrostatic atmospheric simulation and prediction model. Part I: Model dynamics and verification. Meteorology and Atmospheric Physics. 2000;75(3/4):161-193
- [6] ARPA (Agenzia Regionale per la Protezione dell'Ambiente) (2002). Siracusa, Environmental Report 2002. Siracusa, 2002
- [7] Decreto Assessoriale (Territorio e Ambiente) 14 giugno 2006. Approvazione del piano di azione con gli interventi di prevenzione dell'inquinamento atmosferico dell'area ad elevato rischio di crisi ambientale della provincia di Siracusa. Gazzetta Ufficiale della Regione 2006.25.2041
- [8] Armendia A. A study of correlation between atmospheric stability and atmospheric pollution in Salamanca (Spain). In: Proc. 8th World Clean Air Congress "Man and His Ecosystem", The Hague, Holland, 11-15 September 1989; 1989
- [9] Di Lorenzo A. New technologies in air pollution prevention and management: A case study. In: Proc 12th World Clean Air & Environment Congress and Exibition "Greening the New Millennium", vol. 2; 26-31 August 2001; Seoul Korea. 2001
- [10] Hsin-Chung L. The influence of meteorological conditions to the statistical characters of air pollutants.In: Proc. 12th World Clean Air & Environment Congress and Exibition, "Greening the New Millennium", vol. 1; 26-31 August 2001; Seoul, Korea. 2001
- [11] Seinfeld JH. Atmosperic Chemistry and Physics of Air Pollution. New York: Wiley; 1986
- [12] Biamino W, Gambadoro A, Trivero P, Zerbo G. An experimental approach to forecasting the atmospheric pollution in a complex industrial area. In: Proc. 12th World Clean Air & Environment Congress and Exhibition; 26-31 Aug. 2001; Seoul. 2001
- [13] Tinarelli G, Anfossi D, Brusasca G, Ferrero E, Giostra U, Morselli MG, Moussafir J, Tampieri F, Trombetti F. Lagrangian particle simulation of tracer dispersion in the lee of a schematic two-dimensional hill. Journal of Applied Meteorology. 1994;33(6):744-756
- [14] Munda G. Beyond welfare economics: Some methodological issues. Journal of Economic Methodology. 2016;23(2):186-202
- [15] Pawlak Z, Slowinski R. Rough set approach to multi-attribute decision analysis. European Journal of Operational Research. 1994;**72**(1994):443-459
- [16] Pawlak Z. Rough sets. International Journal of Information and Computer Science. 1982;11:341-356

- [17] Pawlak Z. Rough Sets. Theoretical Aspects of Reasoning about Data. Dordrecht: Kluwer; 1991
- [18] Greco S, Matarazzo B, Slowinski R. Granular computing for reasoning about ordered data: the dominance-based rough set approach. Chapter 15. In: Pedrycz W, Skowron A, Kreinovich V, editors. Handbook of Granular Computing. Chichester: Wiley; 2008. pp. 347-373
- [19] Greco S, Matarazzo B, Slowinski R. Rough sets theory for multicriteria decision analysis. European Journal of Operational Research. 2001;129:1-47
- [20] Greco S, Matarazzo B, Slowinski R. Assessment of a value of information using rough sets and fuzzy measures. In: Chocjan J, Leski J, editors. Fuzzy Sets and Their Applications. Dordrecht: Silesian University of Technology Press; 2001. pp. 185-193
- [21] Greco S, Matarazzo B, Slowinski R. Rough sets methodology for sorting problems in presence of multiple attributes and criteria. European Journal of Operational Research. 2002;138:110-123
- [22] Greco S, Matarazzo B, Slowinski R. Decision rule approach. In: Figueira J, Greco S, Ehrgott M, editors. Multiple-Criteria Decision Analysis. State of the Art Surveys. New York, NY: Springer International Series in Operations Research and Management Science; 2005. pp. 507-561
- [23] Greco S, Inuiguchi M, Slowinski R. Fuzzy rough sets and multiple-premise gradual decision rules. International Journal of Approximate Reasoning. 2006;41:179-211
- [24] Greco S, Matarazzo B, Slowinski R. Dominance-based rough set approach to decision involving multiple decision makers. In: Greco S, Hata Y, Hirano S, Inuiguchi M, Miyamoto S, Nguyen HS, Slowinski R, editors. Rough Sets and Current Trends in Computing (RSCTC 2006). Springer LNCS 4259, Berlin; 2006. pp. 306-317
- [25] Greco S, Matarazzo B, Slowinski R. Dominance-based rough set approach as a proper way of handling graduality in rough set theory. Transactions on Rough Sets, Springer LNCS 4400, Berlin; 2007. pp. 36-52
- [26] Greco S, Matarazzo B, Slowinski R. On joint use of indiscernibility, similarity and dominance in rough approximation of decision classes. Research Report RA-012/98. Institute of Computing Science, Poznan University of Technology, Poznan, November 1998. In: Proc. 5th International Conference of the Decision Science Institute; July 4-7, 1999; Athens, Greece. 1998. pp. 1380-1382
- [27] Greco S, Matarazzo B, Slowinski R. The use of rough sets and fuzzy sets in MCDM. Chapter 14. In: Gal T, Stewart T, Hanne T, editors. Advances in Multiple Criteria Decision Making. London: Kluwer; 1999. pp. 14.1-14.59
- [28] Blaszcynski J, Greco S, Matarazzo B, Slowinski R, Szelag M. jMAF–Dominance-based rough set data analysis framework. In: Rough Sets and Intelligent Systems–Professor Zdzislaw Pawlak in Memoriam Intelligent Systems Reference Library. Berlin Heidelberg: Springer Verlag; 2013. pp. 185-209

- [29] Clasadonte MT, Matarazzo A, Pappalardo N. Rough set analysis for urban air pollution control system. In: Book of Abstract of 22nd European Conference on Operational Research—EURO XXII; July 8-11, 2007; Prague. 2007
- [30] Clasadonte MT, Matarazzo A, Zerbo A. Analisi della concentrazione di PM10 in zona ad alto rischio industriale. PM10 analysis in a High-risk Environmental Area, La termotecnica", 6, Luglio-Agosto 2004;2004. pp. 44-48
- [31] Clasadonte MT, Matarazzo A, Pappalardo N, Collura F, Cortina G, Toscano FP, Zerbo A. Rough set analysis applied to the study of air pollutants in urban areas. In: Proc 13th World Clean Air and Environmental Protection; 22-27 August 2004; London, UK. 2004. pp. 324-330
- [32] Greco S, Matarazzo B, Slowinski R. Dominance-based rough set approach to decision under uncertainty and time preference. Annals of Operations Research. 2010a;**176**:41-75
- [33] Greco S, Matarazzo B, Slowinski R. Dominance-based rough set approach to interactive evolutionary multiobjective optimization. In: Greco S, Marques Pereira RA, Squillante M, Yager RR, Kacprzyk J, editors. Preferences and Decisions: Models and Applications. Studies in Fuzziness 257. Berlin: Springer; 2010. pp. 225-260
- [34] Greco S, Matarazzo B, Slowinski R, Stefanowski J. An algorithm for induction of decision rules consistent with the dominance principle. In: Ziarko W, Yao Y, editors. Rough Sets and Current Trend in Computing (RSCTC'2000). Berlin Heidelberg New York: Springer Verlag; 2000. pp. 304-313
- [35] Matarazzo A. Air micro-pollutants analyzed through rough set approach. Progress in Industrial Ecology. 2016;10:382-897
- [36] Matarazzo A, Clasadonte MT, Pappalardo N. An application of dominance based rough set approach to analyse factors affecting interannual variability of ozone concentrations in urban areas. In: Book of Abstract of 22nd European Conference on Operational Research—EURO XXII; July 8-11, 2007; Prague. 2007. p. 168
- [37] Matarazzo A, Clasadonte MT, Zerbo A, Pappalardo N. Innovative tecniche multicriteriali applicative all'analisi dei BTX in atmosfera. New Multicriteria Tecnhiques Applied to BTX Analysis of Air Pollutants. In: Proc I Congresso Seriale "Tematiche Ambientali – Qualità dell'area Nelle Aree Industriali Ed Urbane"; 16 Dicembre 2004; Udine. 2004
- [38] Slowinski R. Intelligent Decision Support. Handbook of Applications and Advances of the Rought Sets Theory. Dordrecht: Kluwer Academic Publishers; 1992
- [39] Matarazzo A, Cavallaro F, Zerbo A, Ciraolo L. La gestione della qualità dell'aria in zona industriale ad alto rischio ambientale: una applicazione dei rough set all'analisi degli inquinanti. Air Quality Management in a High-risk Environmental Area: An Application of Rough Set Analysis to Air Pollutians. In: Proc XIX Congresso Nazionale di Merceologia, Vol.II; 27-29 settembre 2000; Sassari-Alghero. 2000
- [40] Matarazzo A, Clasadonte MT, Zerbo A. Ozone analysis with dominance based rough set approach in a high-risk environmental area. In: The 14th International Conference "Air

Quality-Assessment and Policy at Local, Regional and Global Scales"; 6-10 October 2003; Dubrovnik, Croatia. 2003. pp. 287-295

- [41] Matarazzo A, Clasadonte MT, Zerbo A. Rough set analysis to manage urban air pollution control system. In: Proc of the NACA IUAPPA Conference "An International Perspective on Air Quality Management Issues in the Developing World with Emphasis on Africa"; 1-3 October 2008; Nelspruit (South Africa). 2008. pp. 1-8
- [42] Matarazzo A, Pappalardo N, Clasadonte MT. La Rough Set Analysis per l'individuazione di Indicatori Sintetici di Performance Ambientale Nel Comparto Aria Dei Centri Urbani, the Rough Set Analysis to Achive Air Quality Performance Synthetic Indicators in Urban Centers. In: Proc Primo Incontro Sui Modelli Matematici Applicati Alle Gestione Ambientale; 26-27 Febbraio 2007; Milazzo. 2007
- [43] Matarazzo A, Clasadonte MT, Ingrao C. The (dominance based) rough set approach applied to air pollution in a high risk rate industrial area. Environmental Engineering and Management Journal. 2018;17(3):1-8

The Hidden Face of Nitrogen Oxides Species: From Toxic Effects to Potential Cure?

Ségolène Depayras, Tatiana Kondakova, Hermann Josef Heipieper, Marc GJ Feuilloley, Nicole Orange and Cécile Duclairoir-Poc

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/intechopen.75822

Abstract

Nitrogen oxide (NOx) species represent ones of the most threatening air pollutants due to their prevalence and harmful impact on the environment and human health. The term NOx gathers mainly nitric oxide (NO) and nitrogen dioxide (NO₂), mostly produced by anthropogenic activities such as transport and industries. Several cellular constituents were already described as NOx targets. These include membranes, proteins, respiratory chain enzymes, lipids, and DNA. Such damages lead to pathologies of lungs, cardio-vascular system, and skin because these organs represent the first barrier toward the environment. On the other hand, NOx is also naturally synthetized by several organisms, playing a mediator role in essential cellular functions. However, few data are yet available on NOx activity toward microorganisms. Here, we review data concerning the double face of NOx, including their use in the medical field against pathogens' infections that highlight the versatility of these compounds.

Keywords: NOx, pollutants, signalization, physiopathology, treatment

1. NOx species: the genesis of a major air pollutant

"Yet, pollution is the largest environmental cause of disease and death in the world today, responsible for an estimated 9 million premature deaths" stated The Lancet Commission in 2017. This phenomenon represents an alarming threat for human health, as a major cause of respiratory and cardiovascular pathologies as well as infertility. Moreover, those atmospheric pollutants have severe impacts on the environment and participate in climatic change,

IntechOpen

© 2018 The Author(s). Licensee IntechOpen. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.

acidification, eutrophication, and ecosystem disturbances. Several international and national organizations (e.g. WHO, EEA, and INERIS) aim at reducing the global emission of pollutants, thanks to environmental policies as Kyoto and Gothenburg protocols signed in 1997 and 1999, respectively. After that, an encouraging decrease in air pollutants' levels was measured between 2000 and 2015 [1].

Atmospheric pollutants can be classified into four families: classical, indoor, and organic or inorganic air pollutants. Among the classical ones, which are the principle in amount, sulfur dioxide (SO₂), particle matter, ozone (O₃), and nitrogen oxide (NOx) species are found [2]. The term NOx refers to a wide range of nitrogen-derived compounds, where nitric oxide (NO) and nitrogen dioxide (NO₂) are predominant [3]. Those compounds can be naturally produced at low level by lightnings [4] and volcanic eruptions [5]. However, NOx is mainly generated by anthropogenic activity (e.g., road transport, energy production, industry, and agriculture) (**Figure 1**) [6].

Even if NO and NO₂ represent the main species of NOx, nitrogen exists in several oxidation states in the environment, from N (–III) to N (+V). NO takes a central place in the series of reactive nitrogen species (RNS) [7]. Oxidation and reduction of NO result in the formation of several RNS, including nitrate or ammonium. NO₂ is formed by the reaction of NO with O₂ and can be dimerized to give N₂O₄. The formed reactive species can, in turn, be involved in a wide range of reactions (**Figure 2**).

Nitrogen is an essential constituent of vital macromolecules (nucleic acids and proteins). This atom also constitutes the dinitrogen (N_2) gas, usually abbreviated nitrogen. This gas represents 78% of atmospheric air and is continuously recycled in our environment. First, the atmospheric nitrogen is fixed in the soil by prokaryotes to form ammonia (NH_3), which could be taken up by plants. NH_3 is then converted into ammonium (NH_4^+). Ammonium could be also



Figure 1. NOx sources linked to anthropogenic activities (according to [1]).

The Hidden Face of Nitrogen Oxides Species: From Toxic Effects to Potential Cure? 21 http://dx.doi.org/10.5772/intechopen.75822



Figure 2. Redox relationships of NOx with other RNS. Black arrow: oxidation and reduction of NO; purple arrow: a wide range of reactions; roman numbers: oxidation state of the nitrogen atom; color gradient: reduction/oxidation state from (-III) to (V).

formed by mineralization of organic matter by decomposers (fungi, worms, and prokaryotes). The second part of the nitrogen cycle is the nitrification of ammonium by prokaryotes leading to nitrite (NO_2^{-}) formation, which is then further oxidized into nitrate (NO_3^{-}) . Finally, nitrate could be assimilated by plants for their growth or converted into N_2 by denitrifying bacteria (**Figure 3**).

Moreover, nitrogen is also the precursor of NOx. Indeed, NOx formation can be separated into four steps. First, NO is formed in the atmosphere in combination with nitrogen, resulting from the global biogeochemical nitrogen cycle, and oxygen, following combustion (natural or anthropic). Then, this highly reactive compound can be oxidized by oxygen (O_2) to form NO₂. Alternatively, nitrogen dioxide can also be directly formed through catalytic ammonia combustion or nitrosyl chloride oxidation. Then, a temperature-dependent equilibrium is establishing between NO₂ and its dimeric form, nitrogen tetroxide ($N_2O_4^-$). Finally, a photochemical reaction between NO₂ and hydroxyl radicals leads to the generation of ozone (O_3^-) (**Figure 4**).

Even if NO and NO_2 are classified as RNS, their characteristics are very different suggesting their diverse behavior and reactivity. For example, the solubility of NO is very low



Figure 3. Nitrogen cycle at the interfaces among air, soil, and ecosystem. (1) Fixation of gaseous nitrogen by prokaryotes leading to ammonia (NH_3) and ammonium (NH_4^+); (2) nitrification of NH_4^+ by prokaryotes resulting in nitrite formation (NO_2^-) subsequently oxidized into nitrate (NO_3^-); (3) denitrifying prokaryotes catalyzing the conversion of nitrate into gaseous N_2 .

 $(5.7 \times 10^2 \text{ g/L})$, and NO₂ is highly reactive in water resulting in the formation of HNO₂ and HNO₃ [8]. Moreover, the half-life time of those compounds highly differs because NO₂ half-life is around 35 h, whereas that of NO is almost impossible to determine because of its high reactivity [9]. The permeability of both compounds is also totally unrelated: NO has a lipophilic behavior with consequently a high-membrane permeability. This can be illustrated by the high-diffusion ability of NO toward the membrane. Conversely, the permeability coefficient of NO, is estimated about 5 cm s⁻¹ suggesting a lower diffusion power toward biological membranes [10]. In spite of these differences, the penetration of NO and NO, inside cells and their high reactivity are at the origin of their pathogenic potential on human health, particularly when physiological elimination thresholds are exceeded. Since skin, respiratory tract, and lungs are the first barrier toward those gases, these organs are evidently the principle targets of these compounds and their derived species [11–13]. However, their diffusion deeper in the organism can lead afterward to other severe effects, for instance, on the cardiovascular and immune systems [12, 14]. Human health effects range from reversible (nausea, breathing difficulties, and asthma symptoms) to irreversible (cardiovascular defects, emphysema, and immunopathologies), including cancer induction in the worst cases (Figure 5) [15].

The Hidden Face of Nitrogen Oxides Species: From Toxic Effects to Potential Cure? 23 http://dx.doi.org/10.5772/intechopen.75822



Figure 4. Atmospheric genesis of NOx (adapted from [3]).



Figure 5. Representation of NOx targets and associated pathologies [11-15].

Thanks to epidemiological and clinical studies, thresholds of NO and $NO_{2'}$, whose chemical structure and conversion factors are represented **Figure 6A**, have been determined for different human health effects (**Figure 6B** and **C**, respectively). However, no guideline values are available for NO (**Figure 6D**) due to its complete and rapid reaction [16]. Thus, in order to limit health issues, international organizations, such as WHO, set up guidelines for atmospheric NO₂ limit value (**Figure 6E**).

NOx is then highly toxic compounds and has a myriad of deleterious impacts on the human physiology when the bearable thresholds are exceeded. However, NOx is also naturally produced by cellular processes in a wide range of living organisms. In this case, NOx can have totally different effects as discussed later.



Figure 6. Key features concerning NOx. (A) Lewis structure of NOx and conversion factors; (B) NO threshold (ppm) according to [9]; (C) NO₂ threshold (ppm) according to [9]; (D) and (E) NO and NO₂ guideline values, respectively, according to [1]. ppm, parts per million; ND, not determined.

2. The double face of NOx in the eukaryotic world

To better understand why NOx impact on so many organs, it is essential to focus on the units forming each part of our body: the eukaryotic cells themselves.

2.1. Exogenous NOx: a hostile intruder inside the eukaryotic cells

Much work was already performed on NOx targets at the cell scale. NO and NO_{γ}, thanks to their chemistry properties mentioned earlier, are highly diffusible through living membranes and exhibit a strong reactivity. It is easy to imagine the potential of such compounds. Since NO is highly reactive in lipophilic media, it can react inside the lipid bilayer, first protection of cells against the environment. Indeed, polyunsaturated fatty acids (PUFAs) are susceptible to NO in the presence of O, and ONOO⁻ [17]. The interaction of these toxic compounds with PUFA double bounds leads to the formation of nitrated FA. This process is considered as a protective strategy because it redirects O₂⁻ and ONOO⁻-mediated cytotoxic reactions to other oxidative pathways. Post-translational modification of several proteins is also observed in the presence of NOx and RNS. For example, tyrosine nitration on phenol residue mediated by peroxynitrite derived from OH and NO, leads to 3-nitrotyrosine formation [18]. This molecule represents a useful biomarker of nitrosative stress in various pathologies such as atherosclerosis [19]. Protein thiol residues are highly susceptible to NO and ONOO⁻ leading to the formation of nitrosothiol (RS-NO) and nitrothiol (RS-ONO), respectively [20]. This reversible modification modulates the activity of several proteins similar to phosphorylation. For example, S-nitrosylated cysteine thiol residue can be denitrosylated by S-nitrosoglutathione reductase or thioredoxin systems [21]. NO and other RNS can also nitrosylate [Fe-S] cluster of transition metal centers, which are essential for protein function. This modification can activate a wide range of enzymes such as the soluble guanylate cyclase (sGC) mentioned below
[22]. On the contrary, this reaction can also alter the protein function, for example, in the case of hemoglobin [23]. NO and other higher RNS products ($ONOO^-$ or NO_2) are able to generate nitration, nitrosation, or deamination reactions on DNA bases leading to mutagenesis [24]. This phenomenon is enhanced by the inhibition of DNA repair triggered by NO [25]. More recently, a role of NO in epigenetic modification was suggested. To be more precise, this simple molecule seems to modulate histone acetylation and methylation through direct and indirect modulations of histone acetyltransferases and deacetylases, lysine demethylases, histone methyltransferases activity, thus modifying the expression of several genes [26]. These pleiotropic activities of NO and derived compounds highlight here the necessity of fine regulation pathways to prevent the development of several diseases.

2.2. Endogenous NO: an essential mediator of cellular signalization

Interestingly, NO can also be produced by several living organisms: plants, animals, and bacteria, thanks to a specific enzyme called nitric oxide synthase (NOS) [27, 28]. The large distribution of this enzyme through the different reigns emphasizes the importance of NO synthesis. Evolutionary studies highlight the necessity for the first living organisms during primitive era to eradicate toxic O₃ present in the paleoatmosphere as a survival strategy. Indeed, the liberation of gaseous NO in extracellular environment could have subsequently neutralized O₃, thus limiting harmful oxidative reactions [29]. In eukaryotic cells, three isoforms of NOS have been described [30]. The neuronal nNOS (NOS-1) and the endothelial eNOS (NOS-3) are constitutively expressed but are only activated through calcium-dependent mechanisms. The third one is the inducible iNOS (NOS-2) expressed in macrophages following infection by pathogens, virus, or tumors. Contrary to NOS-1 and 3, NOS-2 is constitutively functional. Interestingly, these NOSs have high structural similarities with an oxygenase and a reductase domain (**Figure 7**) [27].

To be fully functional, NOS requires to associate with homodimers, and this form of NOS is crucial for the generation of NO [32]. When conditions are favorable (high level of L-arginine



Figure 7. The structural similarities of NOS (adapted from [31]). BH₄, tetrahydrobiopterin; L-arg, L-arginine-binding site; Heme Fe, iron protoporphyrin IX; CaM, calmodulin-binding site; FMN, flavin mononucleotide-binding site; FAD, flavin adenine dinucleotide-binding site; NADPH, nicotinamide adenine dinucleotide phosphate-binding site; *, palmitoylation and myristoylation of eNOS oxygenase domain.

and cofactors like BH₄), this enzyme catalyzes the conversion of L-arginine into L-citrulline using O_2 and leading to NO formation with a ratio of 1:1 [33]. However, if these parameters are not available or if NOS remains monomeric, the enzyme only produces superoxide (O_2^{-}) [32]. Even if NOS isoforms catalyze the same reaction, their distribution is largely related to their respective functions.

2.2.1. The endothelial eNOS and vasodilatation

Endothelial nitric oxide synthase (eNOS) can be activated through calcium-dependent and independent pathways. On the one hand, through activation of specific receptors, such as acetylcholine muscarinic receptors, bradykinin receptors, and H1 histamine receptors, distributed in the endothelial cell membrane, agonists can trigger an increase in intracellular concentration of calcium (Ca²⁺) through the well-known polyphophoinosides pathway. Indeed, activation of those receptors stimulates membrane-associated PLCY and PI3K activation. This results into inositol trisphosphate (IP3) formation, which induces the release of intracellular calcium stock from endoplasmic reticulum [34]. Then, Ca²⁺ binds to calmodulin (CaM), which could later fix on the calmodulin-binding domain of eNOS controlling its enzymatic activity. On the other hand, in response to hormones or growth factors acting on their corresponding receptors, phosphate kinase A (PKA) or B (AKT) pathway can be induced mediating a phosphorylation cascade. The post-translational phosphorylation of eNOS on three specific sites (Ser617 and Ser1179 for AKt, Ser635 and Ser1177 for PKA) enhances the activity of the enzyme [35]. Moreover, the lipidation of eNOS (palmitoylation and myristoylation) could also enhance its activity [36]. This other post-translational modification promotes eNOS association with cell membrane and is stabilized by interaction with membrane chaperone proteins (caveolin-1 and HSP70/90). Both mechanisms are essential for linking upstream signal transduction pathway to eNOS activity in cells [37]. eNOS could then produce a large amount of NO, which latter diffuses freely inside smooth muscles. Herein, NO activates soluble guanylate cyclase (sGC) by reaction with the heme of the enzyme leading to the increase of cyclic guanosine monophosphate level (cGMP). cGMP subsequently activates phosphate kinase G (PKG) favoring a cytosolic Ca²⁺ reuptake into sarcoplasmic reticulum [38]. The decrease of intracellular level of Ca²⁺ leads to the relaxation of smooth muscle. Moreover, NO can inhibit caspases 3 and 8 and thus apoptosis through protein nitrosylation. NO released by endothelium is also reducing platelet aggregation and platelet or leukocyte adhesion (Figure 8) [39, 40].

Another source of NO inside endothelial cells was also described. Indeed, nitrite and nitrate reservoirs could be converted into NO by several enzymes such as cytochrome P450, hemoglobin, myoglobin, and others, under specific conditions [42]. However, several pathways could abolish NO production. Indeed, NO can be eliminated in combination with reactive oxygen species (ROS). eNOS could be inactivated by several phosphatases. The removal of eNOS membrane sequestration by interaction between NOS-interacting protein (NOSIP) and NOS-traffic inducer (NOSTRIN) complex favors eNOS recycling [36]. Since NO plays a central role in this phenomenon, any abnormality altering its production leads to pathogenesis and vascular disorders such as atherosclerosis and hypertension [43].

The Hidden Face of Nitrogen Oxides Species: From Toxic Effects to Potential Cure? 27 http://dx.doi.org/10.5772/intechopen.75822



Figure 8. Physiology of vasodilatation: implication of eNOS and NO (adapted from [41]). eNOS, endothelial nitric oxide synthase; ach, acetylcholine; BK, bradykinin; H, histamine; R, receptor; CaM, calmodulin; PLCY, phospholipase C Y; PI3K, phosphoinositide 3-kinase; PKA, phosphate kinase A; AKT, phosphate kinase B (PKB); sGC, soluble guanylate cyclase; cGMP, cyclic guanosine monophosphate; PKG, phosphate kinase G; A, protein complex of HSP70, HSP90, and caveolin-1; NOSIP, NOS-interacting protein; NOSTRIN, NOS traffic inducer.

2.2.2. The neuronal nNOS: a crucial element in neurotransmission

nNOS, also called NOS-1, was the first isoform discovered in the neuronal tissue. It should be distinguished between two subfamilies: $nNOS\alpha$ and $nNOS\mu$, thanks to their locations and functions. Indeed, $nNOS\alpha$ is found in central nervous system and plays a major role in the neurotransmission at neuronal synapses, whereas $nNOS\mu$ is found in skeletal and cardiac muscle where NO controls muscle contractility and local blood flow.

2.2.2.1. nNOS α in the central nervous system

Glutamate is the major excitatory neurotransmitter in the brain. When deliver into the synapse, this molecule is activating a specific receptor (NMDAR) located inside the postsynaptic membrane. Activated NMDAR thus allows the entry of calcium. Ca²⁺ are able to activate calmodulin by physical interaction [44]. Afterward calmodulin binds to nNOS α , which, in turn, is activated. Interestingly, nNOS α can indirectly interact with the NMDAR through its fixation on an adapter protein PSD95 by their PDZ interaction domain [45]. This allows the high speed of NO synthesis in response to NMDAR activation. Thereafter, NO can diffuse back to the presynaptic area leading to an increase glutamate release in the synaptic cleft and thus neurotransmission. CAPON, a chaperon protein of DexRas, was found altering the level of nNOS α present at the membrane [46]. Indeed, CAPON competes with the PDZ domain of nNOS α and thus promotes its detachment from the NMDAR-PSD95 complex. nNOS α interaction with CAPON-DexRas complex allows the activation of DexRas, a small protein *G*, through s-nitrosylation on its thiol residue [47]. Then, DexRas activates the MAPK cascade and the transcription of several genes [47]. Several shutdown systems are available to limit this process, which could dramatically affect the brain physiology by synapses hyperactivation. Indeed, kinases, such as PKC or PKII, can phosphorylate nNOS α leading to its loss of activity [48]. nNOS α can also be sequestrated by two membrane proteins (caveolins 1 and 3) preventing its interaction with calmodulin (**Figure 9**).

As mentioned above, any disorder affecting NO synthesis can lead to dramatic consequences in the brain. Indeed, a lack of NO could impair the neurotransmission and hence the neuronal plasticity, such as long-term potentiation (LTP) in the hippocampus or long-term depression (LTD) in the cerebellum [50]. NO ensures also the correct irrigation of brain by its vasodilator effect on peripheral blood vessel. The synthesis of NO can be altered by a deficiency or a loss of activity of nNOS. Other regulatory elements, such as the decrease of L-arginine precursor availability or cofactors crucial for the catalytic activity of NOS, contribute also to nNOS inactivation. Conversely, in some several brain pathologies (ischemia, strokes, neurodegenerative disorders including Parkinson's and Huntington's diseases, and amyotrophic lateral sclerosis), a large amount of NO is produced [51]. It is hypothesized that NO or derived nitrogen species interact with various proteins inducing several toxic effects. Thus, pharmacological regulation of NO synthesis offers important strategies for the treatment of neurodegenerative and muscle diseases.

2.2.2.2. nNOSµ in peripheral neurotransmission

On the other hand, $nNOS\mu$ is located in skeletal and cardiac muscle where NO controls muscle contractility and local blood flow essential to support muscular effort. When the excitatory neurotransmitter acetylcholine (Ach) is released at the neuromuscular junction, it is activating specific acetylcholine nicotinic receptors present on muscle cells [52]. Activation of these receptors allows the release in the cell sarcoplasm of calcium from exogenous and/or endogenous



Figure 9. The neurotransmission mediated by NOS α in central nervous system (CNS) (adapted from [49]). Glu, glutamate neurotransmitter; NMDAR, N-methyl p-aspartate type-glutamate receptor; A, PSD95 protein adapter between nNOS α and NMDAR; CaM, calmodulin; PKC, phosphate kinase C; PKII, Ca2+/CaM-dependent protein kinase II; CaN, calcineurin; CAPON, C-terminal PDZ domain ligand of neuronal nitric oxide synthase; MAPK, mitogenactivated protein kinase.

sources. This process is also enhanced by the activation of a voltage-dependent Ca²⁺ channel (CaCn) [53]. The increase of Ca²⁺ from exogenous origin activates ryanodine receptor present on the sarcoplasmic reticulum, which, in turn, delivers a large amount of Ca²⁺. When the Ca²⁺ threshold is reached, it is binding on the calmodulin protein. Calmodulin can then interact with the nNOSµ to activate the generation of NO. Interestingly, nNOSµ interacts through its PDZ domain with adapter proteins (syntrophins) linked to a membrane protein complex called dystrophin [54]. Produced NO could then diffuse back to the motor neuron to enhance the liberation of neurotransmitters or the vasodilatation of blood vessels, thus increasing the local blood flow to support muscle needs during contraction efforts (**Figure 10**) [55].

NO synthesis is also well regulated by calpain (Calp). Indeed, this protein interacts with the PDZ domain $nNOS\mu$ provoking its detachment from the dystrophin/syntrophin membrane complex [56]. Interestingly, in muscular Duchenne dystrophy, the damages caused by dystrophin absence are enhanced in case of simultaneous lack of $nNOS\mu$ [57]. Thus, NO donors may offer a potential avenue for therapy.

2.2.3. The inducible iNOS and the innate immunity

NO also takes an important part in innate immunity. Indeed, NO and derived compounds are cytotoxic weapon against tumor cells, microorganisms, and viruses [58]. NO is generated by inducible nitric oxide synthases (iNOS and NOS2) within macrophages. However, more recently, studies have shown that this enzyme is also expressed in other cell line (vascular endothelial cells, hepatocytes, pulmonary, and colonic epithelial cells) [59]. Unlike nNOS or eNOS, its expression is stimulated by several pathways after activation of specific receptors by endotoxins (LPS) or cytokines (TNF α , IFN Υ , and IL1 β). Briefly, the LPS endotoxins are carried by an LPS-binding (LPB) protein to its specific Toll-like receptor 4 (TLR4). Then, the transduction



Figure 10. nNOS μ activity at the neuromuscular junction (adapted from [49]). Ach, acetylcholine; CaCn, voltagedependent Ca²⁺ channel; AchR, acetylcholine receptor; yellow lightning symbols, membrane depolarization; Ca²⁺, calcium; RyR, ryanodine receptor; L-cit, L-citrulline; L-arg, L-arginine; CaM, calmodulin; A, syntrophin α 1, β 1 and β 2 adapter between dystrophin and nNOS μ ; Calp, calpain.

of the signal leads to the transcription of NF- κ B [60]. This transcription factor is essential for the expression of *iNOS* gene. On the other hand, cytokines produced by infected cells activate pathways responsible for the expression of several transcription factors (NF- κ B, AP1, and IRF1) also involved in the transcription of iNOS [61]. Both pathways are necessary to fully upregulate iNOS expression. Moreover, iNOS is constitutively active and exhibits a high affinity for calcium explaining its Ca²⁺-independent activity. iNOS produces a large amount of NO until substrate depletion [62]. Then, NO acts directly or through derived compounds (mostly ONOO⁻ and NO₂) generated through the reaction of NO with ROS formed by NADPH oxidase. These molecules can be generated inside (phagosome) or outside the cells. Altogether, NO and derivatives could act in synergy on several components within the targeted hostile elements such as proteins (thiols, metal centers, and tyrosine residues), nucleic acids, and lipids [63, 64]. Altogether, these modifications lead to irreversible damages impairing cellular metabolism and ultimately inhibit growth and replication mechanisms and even cell death (**Figure 11**).

After infection, NO generation is abolished by neutralization of iNOS activity. Indeed, iNOS undergoes a posttranslational modification called polyubiquitination leading to its proteasomal degradation. Physiological aggresome, an alternative pathway for protein degradation, is also involved in iNOS inactivation [66]. The degradation of NO can also be modulated by its own interaction with ROS. The depletion of L-arginine precursor also shuts down iNOS activity.

Overexpression of iNOS is observed during chronic infection leading to an excessive level of NO and RNS. This phenomenon can impair the host physiology because these high reactive compounds represent mutagenic sources. Thus, the NO balance is a crucial point to



Figure 11. Implication of iNOS in the innate immunity (adapted from [65]). IFNY, interferon Y; IFNR1/2, IFNY receptors 1 and 2; JAK, Janus kinase; STAT, signal transducer and activator of transcription; IRF1, interferon response factor 1; IL1 β , interleukin 1 β ; IL1R, IL1 β receptor; MAPK, mitogen-activated protein kinase; JNK, C-Jun N-terminal kinase; ERK, extracellular signal-regulated kinase; AP1, activator protein 1; TNF- α , tumor necrosis factor α ; TNFR, tumor necrosis factor receptor; NF- κ B, nuclear factor κ B; TLR4, Toll-like receptor 4; CD14, cluster of differentiation 14; LPS, lipopolysaccharide; LPB, LPS-binding protein; ROS, reactive oxygen species.

control in order to limit its tumorigenic potential. The development of iNOS inhibitors and NO-releasing agents offers interesting therapeutic strategies to struggle against cancer and microbial infections.

3. NOx versus prokaryote: "Catch me if you can"

3.1. NOx at the center of prokaryotic life, virulence, and death

Several environmental, commensal, and pathogenic prokaryotes possess an arsenal of proteins dedicated to the nitrogen metabolism under anaerobic conditions [67]. Indeed, an enzymatic chain is involved in the denitrification pathway responsible for the production of NO through nitrite reductase (NIR) as an intermediate. The outcome of this mechanism is the formation of dinitrogen (N_2). This process is crucial for the growth of bacteria in the absence of oxygen because it is coupled with ATP generation (**Figure 12**).

Interestingly, a shortened NOS-like protein called bNOS was discovered in Gram-positive bacteria [28]. This enzyme is deprived of the classical reductase domain already described in other NOS isoforms. However, bNOS is able to recruit other proteins exhibiting a reductase catalytic activity [69]. Similar to mammalian isoforms, bNOS catalyzes the production of NO using L-arginine precursor. However, NOx is key elements in the innate immunity to avoid infection. Indeed, similar to eukaryotic cells, prokaryotes are also sensitive to NOx activity. Thence, such compounds could alter DNA, proteins, and metal centers leading to severe alterations of bacterial growth and replication.

As previously described, NO could be endogenously produced by bacteria but is also a powerful exogenous antimicrobial agent. This highlights the requirement of systems to manage a bearable threshold of NOx within the cell. First, bacteria have developed NO sensing system



Figure 12. Bacterial nitrogen cycle (adapted from [68]). NO_3^- , nitrate; NO_2^- , nitrite; NO, nitric oxide; N_2O , nitrous oxide; N_2 , dinitrogen; NH₂OH, hydroxylamine; NH₄⁺, ammonia; NAR/NAP, nitrate reductase; NIR, nitrite reductase; NOR, nitric oxide reductase; N₂OR, nitrous oxide reductase.

through non-heme iron cluster conjugated with a wide range of detoxification mechanisms. Several pathways involving reaction of NO with non-heme iron cluster were described, mainly in Escherichia coli [70]. Indeed, NO can lead to a raise of the repression of sulfur assimilation operon (suf) exerted by ferric uptake regulator protein (Fur). This reaction is mediated by Fur nitrosylation through NO-derived S-nitrosoglutathione (GSNO). Suf products are required for the biogenesis of the [Fe–S] cluster, which, in turn, gives a lure to get rid of NOx collateral damages. The flavohemoglobin (hmpA) is another response against the nitrosative stress. This enzyme catalyzes the conversion of NO into nitrous oxide (N₂O) under anaerobic conditions. Interestingly, in the presence of NO, the S-nitrosylation of cysteine residues on MetR and FNR inactivates both the proteins. Thus, the repression exerted by MetR and the fumarate and nitrate reductase regulatory protein (FNR) is raised by *hmpA* allowing its transcription. Furthermore, NO-responsive transcriptional factor (NorR) is directly activated by NO and enhances the expression of NorVW protein catalyzing the reduction of NO into N₂O. Redoxsensitive cysteines also take part in the NO-sensing systems. Indeed, NO-activated superoxide regulon (SoxR) protein leads to the transcription increase of superoxide dismutase (sodA) removing O₂⁻ that may react with NO to form NO₂ and ONOO⁻ [71]. Similar to SoxR, peroxide regulon (OxyR) protein is activated by S-nitrosylation. This protein induces gene expression for protective products against nitrosative stress at least by limiting the S-nitrosylation [72]. For example, the expression of catalase KatA, which could buffer free NO, is activated by OxyR [73]. Altogether, these sensing systems lead to the efficient detoxification by generation of less toxic compounds or scavenger molecules.

Moreover, bacteria also develop repairing system of DNA and [Fe—S] cluster damages related to nitrosative stress. Similar to eukaryotic cells, five DNA repair pathways are available in prokaryotes. Among them, the base excision repair (BER), to repair deaminated base, and the nucleotide excision repair (NER), to remove cross-linking in DNA, are predominant [74]. Concerning the [Fe—S] cluster repair, the iron sulfur cluster S protein (IscS), a cysteine desulfurase, can denytrosylate these protein clusters in the presence of L-cysteine [75]. Moreover, Isc regulator (IscR) protein is able to sense nitrosylated [Fe—S] cluster and thus enhances the formation and/or repair of [Fe—S] cluster [76, 77].

NO could also alter tricarboxylic acid (TCA) metabolic pathway and bacterial respiration through reaction with aconitase and cytochromes of the electron transfer chain, respectively [78]. However, these mechanisms are crucial for the generation of ATP and hence the energetic needs of bacteria [79]. To counteract the inefficiency of this enzyme, some bacteria are able to reprogram their metabolism. For example, *Pseudomonas* spp. uses the citrate lyase, phosphoenolpyruvate carboxylase, and pyruvate phosphate dikinase to convert citrate into pyruvate and ATP [80].

In nonlethal concentrations, NO interplays a signaling role in bacteria through several proteins possessing heme-nitric oxide/oxygen-binding domain and is referred under the name H-NOX proteins. H-NOX exhibits a highly conserved domain among bacteria and also shared in mammals (sGC) [81]. All H-NOX are histidine-ligated protoporphyrin IX hemoprotein able to fix NO on its ferrous iron. The H-NOX-encoding genes are found in operons with diverse bacterial signaling genes [82], whose majority is now divided in two classes: containing (i) histidine

kinase (HK) or (ii) diguanylate cyclase (DGCs) and phosphodiesterase (PDE). H-NOX could modify the function of proteins implicated in virulence such as biofilm formation and motility or even *quorum sensing* [83]. For example, H-NOX alters the function of diguanylate cyclase and enhances the phosphodiesterase activity. Altogether, this leads to the biofilm dispersion through the decrease of c-di-GMP, an essential mediator in this process. On the contrary, a more sophisticated system relative to histidine kinase (HnoK) activity leads to the accumulation of c-di-GMP in response to NO favoring the biofilm adhesion and formation. Another important virulence trait could be also modulated by H-NOX as in *Vibrio* species a cross-talk with QS seems to exist. However, a few studies are yet available.

More recently, an antibiotic resistance modulation was promoted by a bacterial NO₂ gaseous exposure. Indeed, an upregulation in expression of MexEF-OprN efflux pump was observed coupled with an increase in related antibiotic resistance [84]. The link between NO₂ and this efflux pump is yet not established, but the authors suggest here a potential role of this protein in the releasing of NOx outside the cells. Previous studies also highlight a correlation between NO endogenously generated by bNOS and increase aminoglycoside resistance [85]. NO and NO₂ could directly interact with antibiotic leading to a less toxic compound and exerted a repression of bacterial respiration preventing drug uptake.

To resume, bacteria are able to produce NO and RNS. Following endogenous burst or exogenous aggression (pollution, inflammation reactions), a physiological threshold is exceeded. However, bacteria developed an ingenious system to counteract the potential effects of NO and derived products. Various interconnected signaling pathways are implicated in this phenomenon: (i) NO detection, (ii) detoxification, (iii) repairing mechanisms, and (iv) metabolic reprogramming. Finally, these fundamental modifications modulate bacterial community behavior (QS) and virulence traits, such as biofilm and antibiotic resistance. This last point is particularly attractive in medicine. Indeed, the emergence of several antibiotic multiresistant pathogens represents a big threat for human that brings us to the last question of this review: "Does NOx therapy represent the new trends in human health care?"

3.2. The novel therapeutic strategies mediated by NOx

A lot of therapeutic strategies exploiting NOx-mediated pathways has recently emerged, particularly to treat cancer. Here we propose to classify them in three families: (i) NO-donor compounds, materials, or nanoparticles; (ii) modulators of NOS activity; and (iii) last but not least, generators of gaseous NO.

The first family NO donors consist of NO-chelating compounds that can release NO under specific conditions within the organism after administration. As shown below, several NO donors have been developed during the last decade. Their global—direct or not—anticancer effects have been exploited in the case of various carcinoma (**Table 1**).

For further information, we strongly recommend to read the review of Huang et al. [86]. Another extension of the use of NO-donor compounds is their graft on material and nanoparticles. Indeed, prosthetic biomaterials used in medical devices were modified through covalent or noncovalent binding of two NO donors (diazeniumdiolates and S-nitrosothiols).

34 Emerging Pollutants - Some Strategies for the Quality Preservation of Our Environment

NO-donor classes	Representative molecule	NO-release pathway	Physiological impact
Organic nitrates	Glyceril trinitrate (GTN)	Enzymatic biotransformation needed	Anticancer activity
Diazeniumdiolates	NONOates	Spontaneous and selective	Antiproliferative activity
Metal nitrosyl complexes	Sodium nitroprusside (SNP)	Spontaneous	Invasion suppression Radiosensitization
1,2,5-oxadiazole N-oxides	Furoxans	Thiol-dependant	Neuroprotective
S-nitrosothiols	S-nitroso-N-acetylpenicillamine (SNAP)	Metal ions- or superoxide- dependant	Therapeutic potential in cancers
Sydnonimine	3 morpholinosydnonimine (SIN1)	pH-dependant	Therapeutic potential in cancers

Table 1. The different classes of NO donor (adapted from [86]).

Several clinical trials have shown their usefulness in the context of preventing thrombus formation [87]. NO-donating nanoparticle systems represent a new promising tumoricidal agents, thanks to their unique properties: (i) strengthening NO-donor stability, (ii) loading a large amount of NO, and (iii) possibility to trigger NO releasing [88].

The modulation of NOS activity is another interesting strategy. In some pathologies inducing exacerbating inflammation (sepsis), a localized "nitrosative burst" occurs, and inhibition of NOS function is a crucial element. Thus, several L-arginine precursor analogs (N^{ω}-substituted L-arginine) were developed and tested. However, since L-arginine is the common point between all NOS isoforms, these analogs represent nonselective drugs. Moreover, such therapy did not reach clinical studies on human stage despite its effects on canine vascular tone [89]. Thus, investigation on a more selective inhibition of iNOS, crucial for inflammation, was conducted through transgenic animals. Unfortunately, iNOS knockout mice exhibit an increase in mortality following polymicrobial sepsis [90]. More recently, researcher focused on increasing NO delivery within cancers. Thus, thanks to technical progress and scientific knowledge updating, iNOS-based suicide gene therapy was investigated through viral vector use. This treatment exhibits promising tumoricidal effects and appears interesting for its specific and localized iNOS expression on animal models [91].

The third strategy is the direct gaseous delivery of NO in the treatment of various infection diseases related to antibiotic resistant bacteria. Ghaffari et al. provided a considerable work on this topic. First, they developed an ingenious delivery system usefulness for the monitoring of gNO effects on microorganisms (bacteria and fungi) and eukaryotic cells [92]. Then, they reported an effective concentration of gNO up to 200 ppm in continuous 4 h delivery exhibiting bacteriostatic effects on a representative group of microorganisms (*Staphylococcus aureus, Pseudomonas aeruginosa, E. coli, Candida albicans,* and *Group B Streptococcus*). Moreover, no toxic effects were observed on representative mammalian cell lines (dermal fibroblasts) of wound-healing pathology such as leg ulcer or burn injuries [93]. Thus, these promising results placed gNO as a potential topical antibiotic agent [94]. Later, they also investigated the potential use of inhaled gNO in the treatment of pathogenic infection in the case of cystic fibrosis. Indeed, this pathology is a threatening pulmonary disease where microbial infection could be lethal. However, a continuous exposure of gNO at such concentration can lead to severe

impact on the human health, particularly through methemoglobinemia. This phenomenon occurs when hemoglobin losses its vital function of oxygen carrier following the saturation of its iron heme with NO. Thus, an intermittent high-dose short-duration exposure was tested to determine the dose/duration most effective treatment. An adapted lung mammalian model was also used (macrophages and monocytes and pulmonary epithelial cells) to appreciate the potential effects of this cure on human health [95]. This study reached phase I clinical studies using a promising NO treatment at 160 ppm for 30 min three times daily for two periods of 5 days [96]. However, as reviewed by Petit et al., it is important to keep in mind that "administration of inhaled NO is associated with and unavoidable codelivery of NO₂," thanks to the spontaneous high reactivity of NO with O₂ [97]. They also reported that the higher NO doses are used, the faster is the formation rate of NO₂. In contrast, a direct gaseous NO₂ exposure could also be benefit in the struggle against undesirable microorganism because it seems to impact their metabolism, social behavior, and growth [84]. Moreover, a continuous high-dose NO₂ exposure seems to lead in membrane alteration through an increased permeability [98].

4. Conclusion

In summary, this review of the literature shows that NOx is ubiquitous and essential in our lives. We are exposed to these compounds through environmental contaminations, but they also result from physiological processes, infections, diseases, or even drugs. NO and derivatives are implicated in a large diversity of vital functions including brain functions; motricity; cardiac, vascular, and pulmonary functions; and immunity. For these reasons, NOx is key molecules of major biochemical interest. As all active substances, NOx can exert both beneficial and adverse effects with regard to the exposition dose. The physiological adaptation to NOx results from a fine-tuned equilibrium where detoxification plays crucial roles. Myriad of finely regulated processes has emerged in response to nitrosative stress. Our knowledge still remains incomplete, and future studies are necessary to finely precise the synergy of each NOx implemented in nitrosative stress, especially when they are presented in gaseous phase that remains less explored. However, in spite of all previously described mechanisms to counteract the nitrosative stress, abnormal NOx thresholds could trigger a wide range of pathologies and even death. These antagonists, positive and negative effects, of NOx are actually intriguing. So, many researches are now focusing on NOx producing pathways to find the most effective treatments and drugs. More investigations are still needed to better understand the real potential of NOx as antitumor, antibacterial agents, and their safe clinical use.

Acknowledgements

We thank Dr. Annabelle Merieau, Djouhar Souak, and Kévin Guérin for their constructive comments. LMSM is supported by grants from Evreux Portes de Normandie, Région Normandie, French research ministry, and European Union (FEDER). S. Depayras is a recipient of a PhD grant from Région Normandie.

Conflict of interest

We have no conflict of interest to declare.

Author details

Ségolène Depayras¹, Tatiana Kondakova², Hermann Josef Heipieper³, Marc GJ Feuilloley¹, Nicole Orange¹ and Cécile Duclairoir-Poc^{1*}

*Address all correspondence to: cecile.poc@univ-rouen.fr

1 Laboratory of Microbiology Signals and Microenvironment, LMSM EA 4312, University of Rouen, Normandy, France

2 Department of Microbiology, University of Illinois at Urbana-Champaign, Urbana, IL, United States

3 Department Environmental Biotechnology, Helmholtz Centre for Environmental Research – UFZ, Leipzig, Germany

References

- [1] Guerreiro C, González Ortiz A, Frank de Leeuw. Air Quality in Europe–2017 Report. 2017. 80 p. https://www.eea.europa.eu/publications/air-quality-in-europe-2017
- [2] WHO (Europe)-Guidelines for indoor air quality. 2010
- [3] Skalska K, Miller JS, Ledakowicz S. Trends in NOx abatement: A review. Science of the Total Environment. 2010;408(19):3976-3989. DOI: 10.1016/j.scitotenv.2010.06.001
- [4] Schumann U, Huntrieser H. The global lightning-induced nitrogen oxides source. Atmospheric Chemistry and Physics Discussions. 2007;7(1):2623-2818. www.atmos-chemphys.net/7/3823/2007/
- [5] Huebert B, Vitousek P, Sutton J, Elias T, Heath J, Coeppicus S, et al. Volcano fixes nitrogen into plant-available forms. Biogeochemistry. 1999;47(1):111-118. https://link.springer. com/content/pdf/10.1023%2FA%3A1006276011055.pdf
- [6] Harrison RM. Sources of air pollution. In: Air Quality Guideliees. Air Quality Guidelines. 2006. pp. 9-31. https://books.google.fr/books?id=7VbxUdlJE8wC&pg=PA29&lpg=PA29 &dq=who+2006+harrison+air+quality+guideline&source=bl&ots=w225wRR7wa&sig=rg tdu2lV-o_03wIRUiVWcS-FJmY&hl=fr&sa=X&ved=0ahUKEwjk6Nr48tLYAhUGPBQKH S0yBI8Q6AEIXDAG#v=onepage&q=who2006harrisonairq
- [7] Patel RP, Mcandrew J, Sellak H, White CR, Jo H, Freeman BA, et al. Biological aspects of reactive nitrogen species; https://ac.els-cdn.com/S0005272899000286/1-s2.0-S00052728

99000286-main.pdf?_tid=80db350e-f838-11e7-9c4d-00000aacb360&acdnat=1515830910_ 5c2ec2c8feb9dd194a83bb07a938dfe9

- [8] Kinugawa T, Enami S, Yabushita A, Kawasaki M, Hoffmann MR, Colussi AJ. Conversion of gaseous nitrogen dioxide to nitrate and nitrite on aqueous surfactants. Physical Chemistry Chemical Physics. 2011;13(11):5144. http://xlink.rsc.org/?DOI=c0cp01497d
- [9] INERIS. Oxydes d'azote NOx. 2011
- [10] Signorelli S, Möller MN, Coitiño EL, Denicola A. Nitrogen dioxide solubility and permeation in lipid membranes. Archives of Biochemistry and Biophysics. 2011;512(2):190-196. http://dx.doi.org/10.1016/j.abb.2011.06.003
- [11] Chang ALS. Expanding our understanding of human skin aging. The Journal of Investigative Dermatology. 2016;**136**(5):897-899. http://dx.doi.org/10.1016/j.jid.2016.02.020
- [12] Ji X, Han M, Yun Y, Li G, Sang N. Acute nitrogen dioxide (NO₂) exposure enhances airway inflammation via modulating Th1/Th2 differentiation and activating JAK-STAT pathway. Chemosphere. 2015;120:722-728. http://www.sciencedirect.com/science/article/ pii/S0045653514012247
- [13] Laumbach RJ, Kipen HM. Respiratory health effects of air pollution: Update on biomass smoke and traffic pollution. The Journal of Allergy and Clinical Immunology. 2012;129(1):3-11. http://www.ncbi.nlm.nih.gov/pubmed/22196520
- [14] Franchini M, Mannucci PM. Short-term effects of air pollution on cardiovascular diseases: Outcomes and mechanisms. Journal of Thrombosis and Haemostasis. 2007;5(11):2169-2174. http://doi.wiley.com/10.1111/j.1538-7836.2007.02750.x
- [15] Latza U, Gerdes S, Baur X. Effects of nitrogen dioxide on human health: Systematic review of experimental and epidemiological studies conducted between 2002 and 2006. International Journal of Hygiene and Environmental Health. 2009;212(3):271-287
- [16] Hanrahan PL. The plume volume molar ratio method for determining NO₂ /NOx ratios in modeling—Part II: Evaluation studies. Journal of the Air & Waste Management Association. 1999;49(11):1332-1338. http://www.tandfonline.com/doi/abs/10.1080/10473 289.1999.10463961
- [17] Augusto O, Bonini MG, Amanso AM, Linares E, Santos CCX, De Menezes SL. Nitrogen dioxide and carbonate radical anion: Two emerging radicals in biology. Free Radical Biology & Medicine. 2002;32(9):841-859
- [18] Souza JM, Peluffo G, Radi R. Protein tyrosine nitration-functional alteration or just a biomarker? Free Radical Biology & Medicine. 2008;45(4):357-366
- [19] Thomson L. 3-nitrotyrosine modified proteins in atherosclerosis. Disease Markers. 2015;2015(2):8
- [20] Brandes N, Schmitt S, Jakob U. Thiol-based redox switches in eukaryotic proteins. Antioxidants & Redox Signaling. 2009;11(5):997-1014. http://www.liebertonline.com/ doi/abs/10.1089/ars.2008.2285

- [21] Benhar M, Forrester MT, Stamler JS. Protein denitrosylation: Enzymatic mechanisms and cellular functions. Nature Reviews. Molecular Cell Biology. 2009;10(10):721-732. http://dx.doi.org/10.1038/nrm2764
- [22] Beuve A. Thiol-based redox modulation of soluble guanylyl cyclase, the nitric oxide receptor. Antioxidants & Redox Signaling. 2017;26(3):137-149. http://www.ncbi.nlm.nih. gov/pubmed/26906466
- [23] Angelo M, Singel DJ, Stamler JS. An S-nitrosothiol (SNO) synthase function of hemoglobin that utilizes nitrite as a substrate. Proceedings of the National Academy of Sciences. 2006;103(22):8366-8371. https://www.ncbi.nlm.nih.gov/pmc/articles/PMC1482500/pdf/ zpq8366.pdf
- [24] Wiseman H, Halliwell B. Damage to DNA by reactive oxygen and nitrogen species: Role in inflammatory disease and progression to cancer. Biochemical Journal. 1996;**313**(2):17-29
- [25] Jaiswal M, LaRusso NF, Shapiro RA, Billiar TR, Gores GJ. Nitric oxide-mediated inhibition of DNA repair potentiates oxidative DNA damage in cholangiocytes. Gastroenterology. 2001;**120**(1):190-199. http://linkinghub.elsevier.com/retrieve/pii/S0016508501634310
- [26] Socco S, Bovee RC, Palczewski MB, Hickok JR, Thomas DD. Epigenetics: The third pillar of nitric oxide signaling. Pharmacological Research. 2017;121:52-58. http://dx.doi.org/ 10.1016/j.phrs.2017.04.011
- [27] Knowles RG, Moncada S. Nitric oxide synthases in mammals. The Biochemical Journal. 1994;298:249-258. https://www.ncbi.nlm.nih.gov/pmc/articles/PMC1137932/pdf/biochemj00092-0010.pdf
- [28] Sudhamsu J, Crane BR. Bacterial nitric oxide synthases : What are they good for? Trends in Microbiology. 2009;17(5):212-218
- [29] Feelisch M, Martin JF. The early role of nitric oxide in evolution. Trends in Ecology & Evolution. 1995;10(12):496-499. http://www.sciencedirect.com/science/article/pii/S01695 3470089206X?via%3Dihub
- [30] Förstermann U, Sessa WC. Nitric oxide synthases: Regulation and function. European Heart Journal. 2012;33: 829-837. https://www.ncbi.nlm.nih.gov/pmc/articles/PMC3345541/ pdf/ehr304.pdf
- [31] Bruckdorfer R. The basics about nitric oxide. Molecular Aspects of Medicine. 2005;26(1-2 SPEC. ISS):3-31
- [32] Vśquez-Vivar J, Kalyanaraman B, Martásek P. The role of tetrahydrobiopterin in superoxide generation from eNOS: Enzymology and physiological implications. Free Radical Research. 2003;37:121-127. http://www.ncbi.nlm.nih.gov/pubmed/12653200
- [33] Thomas G. Medicinal Chemistry: An Introduction. Wiley; 2013
- [34] Long CJ, Stone TW. The release of endothelium-derived relaxant factor is calcium dependent. Journal of Vascular Research. 1985;22(4):205-208. https://www.karger.com/ Article/FullText/158602

- [35] Rafikov R, Fonseca F V, Kumar S, Pardo D, Darragh C, Elms S, et al. eNOS activation and NO function: Structural motifs responsible for the posttranslational control of endothelial nitric oxide synthase activity. Journal of Endocrinology. 2011;210:271-284. https:// www.ncbi.nlm.nih.gov/pmc/articles/PMC3326601/pdf/nihms334170.pdf
- [36] Ortiz PA, Trafficking GJL. Activation of eNOS in epithelial cells. Acta Physiologica Scandinavica.2003;**179**(2):107-114. http://doi.wiley.com/10.1046/j.1365-201X.2003.01207.x
- [37] Chen W, Xiao H, Rizzo AN, Zhang W, Mai Y, Ye M. Endothelial nitric oxide synthase dimerization is regulated by heat shock protein 90 rather than by phosphorylation. PLoS One. 2014;9(8):11. https://www.ncbi.nlm.nih.gov/pmc/articles/PMC4143281/pdf/ pone.0105479.pdf
- [38] Denninger JW, Marletta MA. Guanylate cyclase and the .NO/cGMP signaling pathway. Biochimica et Biophysica Acta (BBA)—Bioenergetics. 1999;1411(2-3):334-350. http:// www.sciencedirect.com/science/article/pii/S0005272899000249?via%3Dihub
- [39] Riddell DR, Owen JS. Nitric oxide and platelet aggregation. Vitamins and Hormones. 1999;57:25-48. http://www.ncbi.nlm.nih.gov/pubmed/10232045
- [40] Radomski MW, Palmer RM, Moncada S. An L-arginine/nitric oxide pathway present in human platelets regulates aggregation. Proceedings of the National Academy of Sciences. 1990;87(13):5193-5197. http://www.pnas.org/content/87/13/5193.long
- [41] Zhao Y, Vanhoutte PM, Leung SWS. Vascular nitric oxide: Beyond eNOS. Journal of Pharmacological Sciences. 2015;129(2):83-94. http://dx.doi.org/10.1016/j.jphs.2015.09.002
- [42] Chen K, Pittman RN, Popel AS. Nitric oxide in the vasculature: Where does it come from and where does it go? A quantitative perspective. Antioxidants & Redox Signaling. 2008;10(7):1185-1198. http://www.ncbi.nlm.nih.gov/pubmed/18331202
- [43] Arnal JF, Dinh-Xuan AT, Pueyo M, Darblade B, Rami J. Endothelium-derived nitric oxide and vascular physiology and pathology. Cellular and Molecular Life Sciences. 1999;55(8-9):1078-1087. http://www.ncbi.nlm.nih.gov/pubmed/10442089
- [44] Fonnum F. Glutamate: A neurotransmitter in mammalian brain. Journal of Neurochemistry. 1984;42(1):1-11
- [45] Strieker NL, Christopherson KS, Yi BA, Schatz PJ, Raab RW, Dawes G, et al. PDZ domain of neuronal nitric oxide synthase recognizes novel C-terminal peptide sequences. Nature Biotechnology. 1997;15(4):336-342. http://www.nature.com/doifinder/10.1038/ nbt0497-336
- [46] Jaffrey SR, Snowman AM, Eliasson MJL, Cohen NA, Snyder SH. CAPON: A protein associated with neuronal nitric oxide synthase that regulates its interactions with PSD95. Neuron. 1998;20(1):115-124. http://www.ncbi.nlm.nih.gov/pubmed/9459447
- [47] Fang M, Jaffrey SR, Sawa A, Ye K, Luo X, Snyder SH. Dexras1: A G protein specifically coupled to neuronal nitric oxide synthase via CAPON. Neuron. 2000;28(1):183-193. https://www.sciencedirect.com/science/article/pii/S0896627300000957

- [48] Rameaut GA, Chiu LY, Ziff EB. Bidirectional regulation of neuronal nitric-oxide synthase phosphorylation at serine 847 by the N-methyl-D-aspartate receptor. The Journal of Biological Chemistry. 2004;279(14, 14):14307. http://www.ncbi.nlm.nih.gov/ pubmed/14722119
- [49] Mungrue IN, Bredt DS. nNOS at a glance: Implications for brain and brawn. Journal of Cell Science. 2004;117(Pt 13):2627-2629. http://www.ncbi.nlm.nih.gov/pubmed/15169833
- [50] Lüscher C, Malenka RC. NMDA receptor-dependent long-term potentiation and long-term depression (LTP/LTD). Cold Spring Harbor Perspectives in Biology. 2012;4(6):15. https:// www.ncbi.nlm.nih.gov/pmc/articles/PMC3367554/pdf/cshperspect-SYP-a005710.pdf
- [51] Zhang N, Diao Y, Hua R, Wang J, Han S, Li J, et al. Nitric oxide-mediated pathways and its role in the degenerative diseases. Frontiers in Bioscience. 2017;22:824-834. http:// www.ncbi.nlm.nih.gov/pubmed/27814649
- [52] Albuquerque EX, Pereira EF, Castro NG, Alkondon M, Reinhardt S, Schröder H, et al. Nicotinic receptor function in the mammalian central nervous system. Annals of the New York Academy of Sciences. 1995;757:48-72. http://www.ncbi.nlm.nih.gov/ pubmed/7611705
- [53] Iino M. Dynamic regulation of intracellular calcium signals through calcium release channels. Molecular and Cellular Biochemistry. 1999;190(1-2):185-190. http://link.springer. com/10.1023/A:1006951317052
- [54] Wells KE, Torelli S, Lu Q, Brown SC, Partridge T, Muntoni F, et al. Relocalization of neuronal nitric oxide synthase (nNOS) as a marker for complete restoration of the dystrophin associated protein complex in skeletal muscle. Neuromuscular Disorders. 2003;13(1):21-31. http://www.ncbi.nlm.nih.gov/pubmed/12467729
- [55] Hong K-S, Kim K. Skeletal muscle contraction-induced vasodilation in the microcirculation. Journal of Exercise Rehabilitation. 2017;13(5):502-507 http://www.ncbi.nlm.nih. gov/pubmed/29114523
- [56] Lainé R, de Montellano PR. Neuronal nitric oxide synthase isoforms alpha and mu are closely related calpain-sensitive proteins. Molecular Pharmacology. 1998;54(2):305-312. http://www.ncbi.nlm.nih.gov/pubmed/9687572
- [57] Tidball JG, Wehling-Henricks M. Nitric oxide synthase deficiency and the pathophysiology of muscular dystrophy. The Journal of Physiology. 2014;592(21):4627-4638. http:// www.ncbi.nlm.nih.gov/pubmed/25194047
- [58] Wink DA, Hines HB, Cheng RYS, Switzer CH, Flores-Santana W, Vitek MP, et al. Nitric oxide and redox mechanisms in the immune response. Journal of Leukocyte Biology. 2011;89(6):873-891. http://www.ncbi.nlm.nih.gov/pubmed/21233414
- [59] Choi J-Y, Nam S-A, Jin D-C, Kim J, Cha J-H. Expression and cellular localization of inducible nitric oxide synthase in lipopolysaccharide-treated rat kidneys. The Journal of Histochemistry and Cytochemistry. 2012;60(4):301-315. http://www.ncbi.nlm.nih.gov/ pubmed/22260992

- [60] Jones E, Adcock IM, Ahmed BY, Punchard NA. Modulation of LPS stimulated NF-kappaB mediated nitric oxide production by PKCε and JAK2 in RAW macrophages. Journal of Inflammation. 2007;4(23):9. http://www.ncbi.nlm.nih.gov/pubmed/18036230
- [61] Kleinert H, Schwarz PM, Förstermann U. Regulation of the expression of inducible nitric oxide synthase. Biological Chemistry. 2003;384:1343-1364. http://www.ncbi.nlm.nih. gov/pubmed/14669979
- [62] Hickey MJ, Granger DN, Kubes P. Inducible nitric oxide synthase (iNOS) and regulation of leucocyte/endothelial cell interactions: Studies in iNOS-deficient mice. Acta Physiologica Scandinavica. 2001;173:119-126. http://doi.wiley.com/10.1046/j.1365-201X.2001.00892.x
- [63] Fang FC. Antimicrobial reactive oxygen and nitrogen species: Concepts and controversies. Nature Reviews. Microbiology. 2004;2(10):820-832. http://www.nature.com/ doifinder/10.1038/nrmicro1004
- [64] Nathan C, Shiloh MU. Reactive oxygen and nitrogen intermediates in the relationship between mammalian hosts and microbial pathogens. PNAS. 2000;97(16):8841-8848
- [65] Lechner M, Lirk P, Rieder J. Inducible nitric oxide synthase (iNOS) in tumour biology: The two sides of the same coin. Seminars in Cancer Biology. 2005;**15**(4):277-289
- [66] Pandit L, Kolodziejska KE, Zeng S, Eissa NT. The physiologic aggresome mediates cellular inactivation of iNOS. Proceedings of the National Academy of Sciences of the United States of America. 2009;106(4):1211-1215. http://www.ncbi.nlm.nih.gov/ pubmed/19139419
- [67] Zumft WG. Cell biology and molecular basis of denitrification. Microbiology and Molecular Biology Reviews. 1997;61(4):533-616. https://www.ncbi.nlm.nih.gov/pmc/articles/ PMC232623/pdf/610533.pdf
- [68] Watmough NJ, Butland G, Cheesman MR, Moir JWB, Richardson DJ, Spiro S. Nitric oxide in bacteria: Synthesis and consumption. Biochimica et Biophysica Acta (BBA)— Bioenergetics. 1999;1411(2-3):456-474. http://linkinghub.elsevier.com/retrieve/pii/S000527 2899000328
- [69] Crane BR. The enzymology of nitric oxide in bacterial pathogenesis and resistance. Biochemical Society Transactions. 2008;**36**(Pt 6):1149-1154. http://www.ncbi.nlm.nih. gov/pubmed/19021514
- [70] Spiro S. Nitric oxide-sensing mechanisms in *Escherichia coli*: Scheme 1. Biochemical Society Transactions. 2006;34(1):200-202. http://biochemsoctrans.org/lookup/doi/10.1042/ BST0340200
- [71] Feld L, Knudsen GM, Gram L. Bactericidal antibiotics do not appear to cause oxidative stress in *Listeria monocytogenes*. Applied and Environmental Microbiology. 2012;78(12):4353-4357. http://aem.asm.org/content/78/12/4353.full.pdf+html
- [72] Seth D, Hausladen A, Wang Y-J, Stamler JS. Endogenous protein S-nitrosylation in *E. coli*: Regulation by OxyR. Science (80-). 2012;**336**(6080):470-473. https://www.ncbi.nlm. nih.gov/pmc/articles/PMC3837355/pdf/nihms519751.pdf

- [73] Heo YJ, Chung IY, Cho WJ, Lee BY, Kim JH, Choi KH, et al. The major catalase gene (*katA*) of *Pseudomonas aeruginosa* PA14 is under both positive and negative control of the global transactivator OxyR in response to hydrogen peroxide. Journal of Bacteriology. 2010; 192(2):381-390. https://www.ncbi.nlm.nih.gov/pmc/articles/PMC2805318/pdf/0980-09.pdf
- [74] van der Veen S, Tang CM. The BER necessities: The repair of DNA damage in humanadapted bacterial pathogens. Nature Reviews. Microbiology. 2015;13(2):83-94. http:// www.nature.com/doifinder/10.1038/nrmicro3391
- [75] Djaman O, Outten FW, Imlay JA. Repair of oxidized iron-sulfur clusters in *Escherichia coli*. The Journal of Biological Chemistry. 2004;279(43):44590-44599. http://www.ncbi. nlm.nih.gov/pubmed/15308657
- [76] Rogers PA, Ding H. L-cysteine-mediated destabilization of dinitrosyl iron complexes in proteins. The Journal of Biological Chemistry. 2001;276(33):30980-30986. http://www.jbc. org/content/276/33/30980.full.pdf
- [77] Schwartz CJ, Giel JL, Patschkowski T, Luther C, Ruzicka FJ, Beinert H, et al. IscR, an Fe-S cluster-containing transcription factor, represses expression of *Escherichia coli* genes encoding Fe-S cluster assembly proteins. Proceedings of the National Academy of Sciences. 2001;98(26):14895-14900. https://www.ncbi.nlm.nih.gov/pmc/articles/PMC 64955/pdf/pq2601014895.pdf
- [78] Husain M, Bourret TJ, McCollister BD, Jones-Carson J, Laughlin J, Vázquez-Torres A. Nitric oxide evokes an adaptive response to oxidative stress by arresting respiration. The Journal of Biological Chemistry. 2008;283(12):7682-7689. http://www.ncbi.nlm.nih. gov/pubmed/18198179
- [79] Appanna VP, Auger C, Thomas SC, Omri A. Fumarate metabolism and ATP production in Pseudomonas fluorescens exposed to nitrosative stress. Antonie van Leeuwenhoek. 2014;106(3):431-438. https://link.springer.com/content/pdf/10.1007/s10482-014-0211-7.pdf
- [80] Auger C, Appanna VD. A novel ATP-generating machinery to counter nitrosative stress is mediated by substrate-level phosphorylation. Biochimica et Biophysica Acta. 2015;1850(1):43-50. http://www.sciencedirect.com/science/article/pii/S0304416514003365
- [81] Iyer LM, Anantharaman V, Aravind L. Ancient conserved domains shared by animal soluble guanylyl cyclases and bacterial signaling proteins. BMC Genomics. 2003;4:8. https://www.ncbi.nlm.nih.gov/pmc/articles/PMC149354/pdf/1471-2164-4-5.pdf
- [82] Plate L, Marletta MA. Nitric oxide-sensing H-NOX proteins govern bacterial communal behavior. Trends in Biochemical Sciences. 2013;38(11):566-575. http://dx.doi.org/ 10.1016/j.tibs.2013.08.008
- [83] Nisbett LM, Boon EM. Nitric oxide regulation of H-NOX signaling pathways in bacteria. Biochemistry. 2016;55(35):4873-4884
- [84] Kondakova T, Catovic C, Barreau M, Nusser M, Brenner-Weiss G, Chevalier S, et al. Response to gaseous NO₂ air pollutant of *P. fluorescens* airborne strain MFAF76a and clinical strain MFN1032. Frontiers in Microbiology. 2016;7:12. http://journal.frontiersin. org/Article/10.3389/fmicb.2016.00379/abstract

- [85] McCollister BD, Hoffman M, Husain M, Vázquez-Torres A. Nitric oxide protects bacteria from aminoglycosides by blocking the energy-dependent phases of drug uptake. Antimicrobial Agents and Chemotherapy. 2011;55(5):2189-2196. http://www.ncbi.nlm. nih.gov/pubmed/21343448
- [86] Huang Z, Fu J, Zhang Y. Nitric oxide donor-based cancer therapy: Advances and prospects. Journal of Medicinal Chemistry. 2017;60(18):7617-7635. http://pubs.acs.org/doi/ 10.1021/acs.jmedchem.6b01672
- [87] Varu VN, Tsihlis ND, Kibbe MR. Basic Science Review: Nitric Oxide-Releasing Prosthetic Materials . Vol. 43, Vascular and Endovascular Surgery. Los Angeles, CA: SAGE Publications Sage CA; 2009. pp. 121-131. http://journals.sagepub.com/doi/10.1177/ 1538574408322752
- [88] Han G, Friedman AJ, Friedman JM. Nitric oxide releasing nanoparticle synthesis and characterization. Methods in Molecular Biology. 2011;704:187-195. http://link.springer. com/10.1007/978-1-61737-964-2_14
- [89] Cobb JP. Nitric oxide synthase inhibition as therapy for sepsis : A decade of promise. Surgical Infections. 2001;2(2):93-101
- [90] Laubach VE, Shesely EG, Smithies O, Sherman PA. Mice lacking inducible nitric oxide synthase are not resistant to lipopolysaccharide-induced death. Proceedings of the National Academy of Sciences of the United States of America. 1995;92:10688-10692. https://www.ncbi.nlm.nih.gov/pmc/articles/PMC40677/pdf/pnas01501-0254.pdf
- [91] Ye S, Yang W, Wang Y, Ou W, Ma Q, Yu C, et al. Cationic liposome-mediated nitric oxide synthase gene therapy enhances the antitumour effects of cisplatin in lung cancer. International Journal of Molecular Medicine. 2013;31(1):33-42
- [92] Ghaffari A, Neil DHH, Ardakani A, Road J, Ghahary A, Miller CCC. A direct nitric oxide gas delivery system for bacterial and mammalian cell cultures. Nitric Oxide. 2005;12(3):129-140. http://linkinghub.elsevier.com/retrieve/pii/S1089860305000054
- [93] Miller CC, Miller MK, Ghaffari A, Kunimoto B. Treatment of chronic nonhealing leg ulceration with gaseous nitric oxide: A case study. Journal of Cutaneous Medicine and Surgery. 2004;8(4):233-238. http://journals.sagepub.com/doi/10.1177/120347540400800406
- [94] Ghaffari A, Miller CC, McMullin B, Ghahary A. Potential application of gaseous nitric oxide as a topical antimicrobial agent. Nitric Oxide—Biological Chemistry. 2006; 14(1):21-29
- [95] Miller C, McMullin B, Ghaffari A, Stenzler A, Pick N, Roscoe D, et al. Gaseous nitric oxide bactericidal activity retained during intermittent high-dose short duration exposure. Nitric Oxide—Biological Chemistry. 2009;20(1):16-23. http://dx.doi.org/10.1016/j. niox.2008.08.002
- [96] Deppisch C, Herrmann G, Graepler-Mainka U, Wirtz H, Heyder S, Engel C, et al. Gaseous nitric oxide to treat antibiotic resistant bacterial and fungal lung infections in patients with cystic fibrosis: A phase I clinical study. Infection. 2016;44(4):513-520

- [97] Petit PC, Fine DH, Vásquez GB, Gamero L, Slaughter MS, Dasse KA. The pathophysiology of nitrogen dioxide during inhaled nitric oxide therapy. ASAIO Journal. 2017;63(1):7-13
- [98] Depayras S, Kondakova T, Merlet-Machour N, Heipieper HJ, Barreau M, Catovic C, et al. Impact of gaseous NO₂ on *P. fluorescens* strain in the membrane adaptation and virulence. International Journal of Environmental Impacts. 2018;1(2):183-192. https:// www.witpress.com/elibrary/ei-volumes/1/2/1806#.WliS_8echfo.mendeley

Multivariate-Assisted Solid Phase Extraction Procedure for Simultaneous Preconcentration and Assessment of UV-Filters in Wastewater Prior to UV-Vis Spectrophotometric Determination

Anele Mpupa and Philiswa N. Nomngongo

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/intechopen.75641

Abstract

Determination of emerging pollutants such as UV-filters in environmental samples is very important because they have been proven to have harmful effects on human and aquatic life. In this study, a simple, fast and inexpensive method combining solid phase extraction (SPE) and UV spectrophotometry was developed for simultaneous preconcentration and determination of benzophenone and sulisobenzone in wastewater samples. The effect of factors affecting the preconcentration of UV-filters was optimized using univariate and multivariate approach. Under optimized conditions, the limits of detection (LOD), limits of quantification (LOQ) and preconcertation factors were in the range of 0.15–0.28 and 0.50–0.93 μ g L⁻¹, 50–55, respectively. The dynamic linear range was up to 250 μ g L⁻¹ for benzophenone and sulisobenzone. In addition, the intra- and inter-day precisions were 3.1–3.3 and 4.5–5.2%, respectively. The developed method was successfully applied to determine UV filters in wastewater samples attaining satisfactory recoveries over the range of 99.3–100.7%. The concentration of the target pollutants in wastewater samples ranged from 6.83 to 85.67 μ g L⁻¹.

Keywords: solid phase extraction, emerging organic pollutants, UV-filters, response surface methodology, UV-Vis spectrophotometry, endocrine disruptors

1. Introduction

IntechOpen

Ultraviolet filters are chemical agents that are used in a variety of cosmetics, specifically those used for sun protection such as sunscreen lotions, creams and sprays [1]. These compounds often contain single or multiple aromatic structures (**Figure 1**), sometimes conjugated with

© 2018 The Author(s). Licensee IntechOpen. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.



Figure 1. Chemical structures of (a) benzophenone and (b) sulisobenzone.

carbon-carbon double bonds and carbonyl moieties [2]. The presence of these functional groups affords UV filters the ability to absorb photons and rapidly return to the ground state by thermally emitting the energy through vibrational relaxation [3]. This makes the compound to be able to mitigate the deleterious effects of UV radiation [1, 2].

After application, UV-filters are washed off and enter the aquatic environmental directly or indirectly via wastewater effluent and recreational water systems [4]. They are also used as sun blocking agents in materials such as plastics, adhesive and rubber, this suggests that these compounds can also leach into the environmental matrices [5–7]. The amount and type of UV-filter used depends on the desired degree of protection, however combined concentrations should not exceed 10% with other organic or inorganic UV-filters [8].

The main concern of the presence of these compounds is their potential toxicity and their effects as xenohormones (effect on reproductive activity) [9, 10]. These effects include estrogenic activity [11], effects on cell proliferation by 4-methylbenzylidene camphor (4-MBC), benzophenones, and octyl methoxycinnamate (OMC) [12]. Several studies have shown hormonal disruption in both in vivo and in vitro test systems in fish and mammals [13–16]. It has also been recently shown that besides estrogens, there are other hormonal targets affected by UV-filters in fish and mammals [17, 18].

A number of detection techniques have been used to quantify UV-filters in environmental water samples. These include techniques high performance liquid chromatography (HPLC) with UV or mass spectrometry detection [19, 20], HPLC-MS/MS [21], gas chromatography-mass spectrometry (GC-MS) and GC-MS/MS [22, 23]. However, the levels of these compounds in environmental waters are usual in the μ g L⁻¹ range. Therefore, sample cleanup/preconcentration techniques such stir-bar sorptive extraction [20], pressurised liquid extraction [1], dispersive liquid-liquid phase extraction [24], solid phase extraction [25], among others, have been used to improve sensitivity, LODs and to remove interferences prior to quantification with the different analytical techniques.

Solid phase extraction is one of the most established preconcentration techniques used for the simultaneous extraction and analysis of organic compounds [26]. Mainly silica bonded phases such as C18 were formerly used for SPE, however recently modified and tuned solid phases can be used to achieve more specificity [27]. Advantages of SPE include the potential of simultaneous extraction, reduced labour and cost. Of the advantages of solid phase extraction, scientists are mostly attracted to the possibility of using reduced amount of organic solvents and the fact that SPE is highly tuneable with regards to the adsorbents used [28].

Thus the aim of this study was to develop a multivariate assisted solid phase extraction method for the simultaneous preconcentration of UV- filters in wastewater samples prior to their spectrophotometric quantification. The main advantage of the method relies on the use of multivariate optimization approach which led to the reduction of the number of experiments and analysis time as well as the use of a simple, fast and cost effective instrumentation. UV-Vis spectrophotometry was chosen due its simplicity and high availability. The factors (such as sample pH, flow rates, eluent and adsorbent types) affecting the preconcentration step method were optimized using univariate and multivariate approach. The developed SPE/spectrophometric method was applied to the preconcentration and determination of two UV filters from the wastewater samples collected from Daspoort wastewater treatment plant (WWTP) in Pretoria, Gauteng, South Africa. According to our literature search there are limited reports on the application of UV-Vis spectrophotometry for quantification of UV filters [29]. In addition, to the best of our knowledge, the application of SPE/UV-Vis spectrophotometry for simultaneous preconcentration and determination of Benzophenone and sulisobenzone has been reported for the first time.

2. Experimental

2.1. Materials and reagents

Benzophenone (Reagent plus, 99%), sulisobenzone (5-benzoyl-4-hydroxy-2-methoxy-benzenesulfonic acid) (HPLC, \geq 97.0%) and acetonitrile (for HPLC Plus) were purchased from Sigma-Aldrich (St. Louis, MO, USA). Ethyl acetate was purchased from Merck (Merck, Darmstadt, Germany), ethanol and methanol were purchased from Associated Chemical Enterprises (Johannesburg, South Africa). Stock solutions of benzophenone and 5-benzoyl-4-hydroxy-2-methoxy-benzenesulfonic acid (10 mg L⁻¹) were prepared in ultra-pure water (Direct-Q® 3UV-R purifier system, Millipore, Merck). Working standards of 100 μ g L⁻¹ were prepared daily by diluting appropriate volumes of the stock solution in ultra-pure water. The ion exchange resins used in this study as packing materials were Dowex 1x8 (Chloride form) (Sigma Aldrich).

2.2. Instrumentation

A Shimadzu UV-2450 high performance single monochromator UV-VIS spectrophotometer (Shimadzu Corporation, Tokyo, Japan) was used for all analysis of the samples. Solid phase extraction (SPE) was carried out in a VacMaster-24 sample SPE station (VacMaster, Biotage, Sweden). The latter was used to control the sample loading and elution flow rate in the range of 1–3.0 mL min⁻¹. An OHAUS starter 2100 pH meter (Pine Brook, NJ, USA) was used for pH adjustments of the reagents and to measure the pH of samples

2.3. Sampling and sample collection

Influent (after sediment removal) and effluent wastewater samples were collected from Daspoort wastewater treatment plant (WWTP, Pretoria, Gauteng, south Africa). The samples were collected in pre-cleaned 500 mL glass bottles. The samples were then refrigerated at 4°C.

Factors	Low level (–1)	Central point (0)	High level (+1)
Sample pH	4	7	10
Sample flow rate (SFR) (mL min ⁻¹)	2	3.5	5
Eluent flow rate (EFR) (mL min ⁻¹)	1	2	3

Table 1. Factors and levels used in 2³ factorial design for extraction and preconcentration of UV filters.

2.4. Solid phase extraction procedure

The columns were prepared according to [30]. Briefly, polyethylene columns of diameter 1.0 cm and 6 cm in height were used for preconcentration. Slurries of 0.5 g of Dowex 1x8 in double distilled deionized water were prepared and packed to columns to heights of about 1 cm. A porous frit was placed at the bottom of the column and at the top of the packing material to hold and confine the adsorbent within the designated capacity/volume. The columns were washed with 6 mL of double distilled deionized water followed by conditioning with 3 mL organic solvent (methanol, ethanol, ethyl acetate or mixture of methanol and acetonitrile) and then 3 mL of double distilled deionized water. Due to the scarcity of reference materials for UV-filters, a commercial sunscreen lotion was used as a reference material for the validation of the SPE/UV method. An appropriate amount of the sunscreen lotion was accurately weighed and dissolved using a small volume of methanol (2 mL) and made to the mark with double distilled deionized water and used for validation experiments.

An aliquot (10 mL) of model solution containing benzophenone and sulisobenzone at a concentration of 100 μ g L⁻¹ was passed through a packed column at a flow rate of 2–5 mL min⁻¹. After percolating the synthetic samples through, the cartridges were washed with 3 mL double distilled deionised water. Then retained analytes were eluted with 2 mL organic solvents. The optimization of the solid phase extraction method was carried out using a 23 full factorial design involving three variables such as pH, sample flow rate (SFR) and eluent flow rate (EFR). Maximum, central point and minimum levels are presented in **Table 1**. The second step of the optimization strategy involved the application of a response surface methodology (RSM) based on a central composite design. All the experiments were carried out in random order and the experimental data was processed by using the Minitab 17 software program.

3. Results and discussion

3.1. Optimization of separation and preconcentration method

3.1.1. Selection of adsorbent and eluent type: univariate approach

Selection of the adsorbent and eluent type was achieved by packing columns using the ion exchange adsorbent (Dowex 1x8). Each experiment was done in triplicates for each solvent type and adsorbent combination. The flow rates and the sample pH were fixed at 2 mL min⁻¹

for both the sample loading and elution and 7, respectively. The results obtained are presented in **Figure 2**.

Selection of the adsorbent and eluent was done using the absorbance of benzophenone and sulisobenzone at the wavelengths of 260 and 220 nm, respectively. Since the absorbance is directly proportional to concentration, a higher absorbance can be related to higher concentration. Thus, from the univariate optimization, the most effective adsorbent and eluent combination was found to be Dowex 1x8 and methanol (**Figure 2**). This combination was further used for the optimization of the solid phase extraction procedure for both benzophenone (UV-01) and sulisobenzone (UV-02).

Due to the overall charge of adsorbents, the extraction and preconcentration of the analytes was possible by means of ionic interactions. Dowex 1x8 has an overall positive charge, thus the negatively charged analytes could interact with the positive charges of the adsorbent. This resulted in higher absorbances and Dowex 1x8 resin was selected as the suitable adsorbent for further studies.

The polarity of the solvent was the contributing factor on the elution of the analytes, the more polar solvent resulted in better elution from the SPE column, and hence higher absorbances for methanol were observed when used in combination with Dowex 1x8. This was a consequence of the methanol having the ability to displace the analytes from the positively charged Dowex 1x8 adsorbent. The 1:1 mixture of acetonitrile and methanol also had promising results, but did not perform better than the methanol alone. Acetonitrile could not be used for this study even though it is more polar than methanol, there are dangers associated with the use of pure acetonitrile like the risk of cyanide poisoning as a result of its decomposition products.

Experimental conditions; mass of adsorbent 0.5 g, sample volume 10 mL at 100 μ g L⁻¹, eluent volume 2 mL, pH 7, flow rates 2 mL min⁻¹. MeOH = methanol, ACN = acetonitrile, EtOH = ethanol.



Figure 2. Selection of suitable adsorbent and eluent combination in model aqueous solution.

3.1.2. Optimization of the SPE operation parameters: Multivariate approach

In order to achieve quantitative preconcentration of analytes by SPE system, the optimization of the most influential parameters, such as sample pH, sample and eluent flow rates, was carried out using 2³ full factorial design and response surface methodology (RSM) based on a central composite design. Two level full factorial design (FFD), involving 11 experiments was used for screening of the significant factors for the extraction and preconcentration of UV filters. **Table 2** presents the factorial design matrix and the analytical responses (expressed as average percentage recovery, %R) obtained in each experiment. Analysis of variance (ANOVA) reproduced in the form of Pareto chart was used to investigate the significance of the effects SPE procedure. The Pareto chart of main effects and their interactions produced are shown in **Figure 3**.

It can be seen from **Figure 3** that for simultaneous preconcentration of UV filters, sample pH and EFR and their interactions were statistically significant. The effect of sample flow rate on the analytical response (%R) was not significant at 95% confidence level. The overall results obtained for the screening step indicated that sample pH and EFR required further optimization. Whereas, sample flow rate was fixed at 3.5 mL min⁻¹.

A central composite design matrix consisting of 14 experiments and analytical response based on each of the experimental runs (**Table 3**) was used for further optimization of the SPE method. The analysis of variance (ANOVA) of the predicted response surface quadratic model for the recoveries of UV filters was obtained. The ANOVA results were analysed using quadratic equations (not included) for the models to illustrate the dependence of the analytical response with respect to the evaluated main effects [31].

The 3D response surface plots (Figure 4) were used to access the interactive relationship between individual variables (sample pH and EFR) and analytical response [31]. Based on

Expt.	pН	SFR (mL/min)	EVR (mL/min)	UV-01 (%R)	UV-02 (%R)
1	4	2	1	34.39	33.7
2	10	2	1	39.63	39.1
3	4	5	1	51.01	53.5
4	10	5	1	72.71	75.4
5	4	2	3	12.22	16.0
6	10	2	3	8.84	9.1
7	4	5	3	11.58	10.6
8	10	5	3	10.24	17.6
9	7	3.5	2	83.55	72.8
10	7	3.5	2	78.88	77.5
11	7	3.5	2	81.08	78.6

Table 2. Two level (2³) full factorial design matrix and analytical response.

Multivariate-Assisted Solid Phase Extraction Procedure for Simultaneous Preconcentration... 51 http://dx.doi.org/10.5772/intechopen.75641



Figure 3. Pareto charts of standardized effects for variables in the preconcentration of (A) benzophenone and (B) sulisobenzone.

quadratic equations and 3D surface response plots, the calculation indicated that pH = 7.5 and $EFR = 2 \text{ mL min}^{-1}$ provided maximum retention and recovery of the studied analytes. Therefore, the results obtained from both designs, illustrated that the optimum conditions that led to quantitative extraction and preconcetration of UV filters were 7.5, 3.5, 2 mL min⁻¹ for sample pH, sample and eluent flow rates, respectively.

Sample	pН	SFR (mL/min)	EFR (mL/min)	UV-01 %R	UV-02 %R
1	4	3.5	1	66.1	63.4
2	10	3.5	1	65.7	44.5
3	4	3.5	3	49.7	16.6
4	10	3.5	3	25.9	19.2
5	7	3.5	2	61.3	51.9
6	7	3.5	2	54.8	59.2
7	7	3.5	2	49.7	49.7
8	2.8	3.5	2	78.8	49.6
9	11.2	3.5	2	42.0	66.0
10	7	3.5	0.5	100.1	103.1
11	7	3.5	3.4	32.2	17.4
12	7	3.5	2	75.0	81.5
13	7	3.5	2	74.8	82.0
14	7	3.5	2	75.1	81.8

Table 3. Central composite design matrix and analytical response.



Figure 4. Response surfaces obtained for (A) benzophenone and (B) sulisobenzone after extraction and preconcentration by SPE.

The effect of pH on the extraction and preconcentration of UV filters can be seen from **Tables 2** and **3**. Acidic pH resulted in lower recoveries. This is because benzophenone and sulisobenzone have pKa values of 7.5 and 7.6, respectively, meaning that in acidic pH they are more likely to accept H^+ ions resulting in lower recoveries as they end up with an overall positive charge. In more alkaline conditions (pH 10), the analytes are easily displaced on the adsorbent, resulting in little or no adsorption.

The central pH (7) showed the highest recoveries for both UV-filters. This is a result of the interaction of the analytes' negative charge with the positive charges of the adsorbent prior to elution with methanol.

The optimum conditions obtained by the multivariate approach were confirmed experimentally. Under these conditions (7.5, 3.5, 2 mL min⁻¹ for sample pH, sample and eluent flow rates, respectively), quantitative recoveries ranging from 96 to 98.6% were obtained. These recoveries were compared with the predicted recoveries values (95.6 and 98.1% for benzophenone and sulisobenzone) obtained using the RSM model. It was then concluded that the results obtained by RSM model were valid since there was no significant difference at a 95% confidence level between the experimental and predicted values.

3.2. Analytical performance

Under the determined optimum experimental conditions, the analytical performances of the developed method for preconcentration and determination of UV-filters were investigated. The calibration curves were obtained after a set of standard solutions (0 to 350 µg L⁻¹) was processed using the described SPE procedure. The concentrations of the analytes in the eluent solutions were quantified with the aid of a UV spectrophotometer. The limits of detection and quantification were calculated using the expressions: $LOD = \frac{3Sd}{b}$ and $LOQ=\frac{10Sd}{b}$, where *Sd* is the standard deviation of 10 replicate measurements at lower concentrations of calibration curves and *b* is the slope of each calibration curves. Dynamic linear ranges (DLR), correlation coefficient (R²), enrichment factor (EF), LOD and LOQ for benzophenone were determined to be LOQ-250 µg L⁻¹, 0.9990, 50, 0.28 µg L⁻¹ and 0.93 µg L⁻¹,

respectively. Whereas for sulisobenzone, the DLR, R², LOD, LOQ, and EF were found to be LOQ-250 μ g L⁻¹, 0.9991, 0.15 μ g L⁻¹, 0.50 μ g L⁻¹ and 55, respectively. Furthermore, the intraday (repeatability; n = 10) and interday (reproducibility; n = 7 working days) precisions of the SPE method, expressed in terms of relative standard deviation (% RSD), ranged 3.1–3.3 and 4.5–5.2%, respectively. The analytical performance of the proposed method was compared to other methods that reported in the literature **Table 4**. It can be seen that the performance of the current method was comparable or better that those reported in the literature. In addition, the LODs obtained using the current method were lower compared to [25, 32].

3.3. Validation and application

The accuracy of the SPE/UV procedure was evaluated using a sunscreen lotion with a sulisobenzone content of 1.75% (w/w). The recovered sulisobenzone was $1.69 \pm 0.07\%$ (w/w) meaning that the percentage recovery was 96.6%. Therefore, the determined values by SPE/UV were in the acceptable range. In addition, the accuracy and matrix effects were investigated by analysing spiked real waste water samples and the results are shown in **Table 5**. From the recoveries shown in **Table 4**, it can be seen that the SPE/UV procedure described was not affected by the matrix effects as the recoveries for both benzophenone and sulisobenzone ranged from 99.3 to 100.7%.

As seen on **Tables 5** and **6**, there was a significant amount of both UV filters on the influent. This can be explained by the fact that the Daspoort waste water treatment plant treats domestic waste water. Therefore as explained by [37], personal care products are usually applied to

Analyte(s)	Matrix	Analytical method	DLDLR (µg L ⁻¹)	LOD (µg L-1)	EF	% RSD	Refs.
Octicrylene	Wastewater	MEPS-GC-MS	0.25–20	0.081	-	7	[33]
Benzophenone-3	Water	CE-ESI-MS	300-20000	150	3400	1.5-6.5	[25]
Benzophenone-3	Water	SBSE-LC-MS/ MS	0.005-0.5	0.0009	-	3–7	[34]
Benzophenone-2	Human serum	DLLME- UPLC-MS/ MS	0.6–40	0.2	-	1.9–13.1	[35]
Benzophenone-3	Sea water	DLLME- GC-MS	0.1–0.5	0.03	262	<15	[25]
Benzophenone	Tap water	DDA-IL- DLLME	0.002–1.5	0.0013	-	3.5–5.3	[32]
Benzophenone	Sunscreen	SPE-GC-MS	10-2000	4.4	25.3	4.6-5.5	[36]
Benzophenone and sulisobenzone	Wastewater	SPE-UV/vis	0.50–250	0.15-0.28	50 and 55	3.1–5.2	Current work

MEPS = microextraction in packed syringe, CE-ESI = capillary electrophoresis-electro spray ionisation, SBSE-LC-MS/ MS = stirbar sorptive extraction-liquid chromatography tandem mass spectroscopy, DLLME-UPLC = dispersive liquid liquid microextraction-ultra pressure liquid chromatography, GC-MS = gas chromatography-mass spectroscopy, SPE = solid phase extraction, DDA-IL =double dispersant assisted-ionic liquid, UV/vis = ultraviolet-visible spectrophotometry.

Table 4. Comparison of the analytical figures of merit of the current method and those reported in the literature.

54 Emerging Pollutants - Some Strategies for the Quality Preservation of Our Environment

Sample	Added (µg/L)	Benzophenone		Sulisobenzone	
		Found (µg/L)	% R	Found (µg/L)	% R
Influent 1	0	85.8 ± 1.3		69.4 ± 1.2	
	50	135.3 ± 2.3	99.3	119.7 ± 3.5	100.7
	100	186.3 ± 2.5	100.6	169.4 ± 4.2	100.0
Effluent 1	0	6.83 ± 0.92		19.8 ± 0.9	
	50	56.7 ± 1.2	99.8	69.6 ± 1.3	99.7
	100	106.9 ± 3.1	100.0	119.1 ± 2.7	99.3

Table 5. Analysis of wastewater samples (influent and effluent) spiked and unspiked from Daspoort (Pretoria, Gauteng, South Africa) wastewater treatment plant.

Samples	Months	Benzophenone		Sulisobenzone	
		SPE/UV	SPE/HPLC	SPE/UV	SPE/HPLC
Influent	August	78.6 ± 1.2	79.4 ± 1.5	56.3 ± 2.1	58.0 ± 1.8
	September	155.3 ± 2.4	157.2 ± 2.2	171.3 ± 3.1	172.2 ± 2.8
	October	327.3 ± 4.3	328.9 ± 4.0	337.3 ± 3.6	337.6 ± 3.6
Effluent	August	18.5 ± 0.7	19.5 ± 0.5	14.5 ± 0.7	15.1 ± 0.5
	September	57.9 ± 1.3	59.0 ± 1.5	23.6 ± 0.9	24.2 ± 0.8
	October	134.5 ± 1.5	135.1 ± 1.2	45.3 ± 0.5	46.0 ± 0.2

 Table 6. Analysis of influent and effluent over from Daspoort (Pretoria, Gauteng, South Africa) wastewater treatment plant a period of 3 months.

the skin and later washed off into drains which are connected to wastewater treatment plants. From wastewater treatment plants, the water is discharged into rivers [3]. This has resulted in the occurrence of UV-filters in surface waters [22], sediments [38, 39], drinking water and even fish [4, 12, 40].

The presence of UV-filter in different water bodies does not only have an effect in humans as suspected endocrine disrupters [18]. Their effects spans into aquatic life, as reported by [3], benzophenone can cause cell membrane impairments in fresh water protozoa. Benzophenone was also found in other fresh water species in percentages ranging from 50 to 80% by [41].

The described method was applied in the analysis of real water samples collected over a period of three months as expressed in **Table 6**. The influent and effluent samples were collected over two seasons, namely winter (one month) and spring (two months). From **Table 6**, it was observed that between the three months, October showed the highest concentrations for both the influent and effluent. This was because it was significantly warmer during spring when compared to August. It is also worthy to note that the concentrations of benzophenone were higher than those of sulisobenzone. This could be the consequence of the degradation of sulisobenzone during the water treatment process. This is possible because benzophenone

forms the backbone of sulisobenzone, this means that when the sulisobenzone losses the methoxy and sulfonic acid side chains, traces of the benzophenone can remain in the water.

The described SPE/UV procedure was compared with a reference method, SPE/HPLC method using the same extraction conditions. The obtained results from the SPE/UV methods were comparable with the SPE coupled with HPLC method. According to the paired student t-test, there was no significant difference between the two methods at 95% confidence level. Therefore, the SPE/UV procedure can be utilised as a rapid, cheap and effective method for the determination of UV-filters in water samples.

4. Conclusions

The combination of SPE and UV-Vis spectrophotometry offers a simple, inexpensive and selective procedure for the assessment of UV-filters in wastewater samples. With the aid of multivariate optimisation of key parameters in the solid phase extraction procedure, it was possible to achieve satisfactory analytical performance. Real sample analysis showed that as explained by [27], wastewater treatment plant processes are not exhaustive enough to completely remove emerging organic pollutants. There were still some traces of benzophenone and sulisobenzone found in the effluent which is released into the nearby river. This means that the flora and fauna in the river is exposed to these potentially toxic compounds. Thus, the next step would be to study the distribution of these UV-filters in the river where the effluent is released into.

Acknowledgements

The authors wish to thank National Research Foundation (NRF, South Africa, Grant No. 99270) for financial assistance.

Author details

Anele Mpupa and Philiswa N. Nomngongo*

*Address all correspondence to: pnnomngongo@uj.ac.za

Department of Applied Chemistry, University of Johannesburg, South Africa

References

 Nieto A, Borrull F, Marcé RM, Pocurull E. Determination of personal care products in sewage sludge by pressurized liquid extraction and ultra high performance liquid chromatography-tandem mass spectrometry. Journal of Chromatography A. 2009; 1216(30):5619-5625

- [2] Giokas DL, Salvador A, Chisvert A. UV filters: From sunscreens to human body and the environment. TrAC Trends in Analytical Chemistry. 2007;26(5):360-374
- [3] Gao L, Yuan T, Zhou C, Cheng P, Bai Q, Ao J, Zhang H. Effects of four commonly used UV filters on the growth, cell viability and oxidative stress responses of the *Tetrahymena thermophila*. Chemosphere. 2013;93(10):2507-2513
- [4] Fent K, Zenker A, Rapp M. Widespread occurrence of estrogenic UV-filters in aquatic ecosystems in Switzerland. Environmental Pollution. 2010;158(5):1817-1824
- [5] Burns JS, Crookes MJ. Using Science to Create a Better Place: UV-Filters in Cosmetics-Prioritisation for Environmental Assessment. Rio House, Waterside Drive, Aztec West, Almondsbury, Bristol: Environment Agency; 2008
- [6] Gackowska A, Przybyłek M, Studziński W, Gaca J. Experimental and theoretical studies on the photodegradation of 2-ethylhexyl 4-methoxycinnamate in the presence of reactive oxygen and chlorine species. Open Chemistry. 2014;12(5):612-623
- [7] Ramos S, Homem V, Alves A, Santos L. A review of organic UV-filters in wastewater treatment plants. Environment International. 2016;86:24-44
- [8] Santos AJM, Crista DM, Miranda MS, Almeida IF, e Silva JPS, Costa PC, Amaral MH, Lobão PA, Lobo JMS, da Silva JCE. Degradation of UV filters 2-ethylhexyl-4-methoxycinnamate and 4-tert-butyl-4'-methoxydibenzoylmethane in chlorinated water. Environmental Chemistry. 2013;10(2):127-134
- [9] Kaiser D, Sieratowicz A, Zielke H, Oetken M, Hollert H, Oehlmann J. Ecotoxicological effect characterisation of widely used organic UV filters. Environmental Pollution. 2012;163:84-90
- [10] Paredes E, Perez S, Rodil R, Quintana JB, Beiras R. Ecotoxicological evaluation of four UV filters using marine organisms from different trophic levels *Isochrysis galbana*, *Mytilus galloprovincialis*, *Paracentrotus lividus*, and *Siriella armata*. Chemosphere. 2014;104:44-50
- [11] Fent K, Kunz PY, Gomez E. UV filters in the aquatic environment induce hormonal effects and affect fertility and reproduction in fish. Chimia International Journal for Chemistry. 2008;62(5):368-375
- [12] Díaz-Cruz MS, Gago-Ferrero P, Llorca M, Barceló D. Analysis of UV filters in tap water and other clean waters in Spain. Analytical and Bioanalytical Chemistry. 2012;402(7): 2325-2333
- [13] Schlumpf M, Schmid P, Durrer S, Conscience M, Maerkel K, Henseler M, Gruetter M, Herzog I, Reolon S, Ceccatelli R, Faass O. Endocrine activity and developmental toxicity of cosmetic UV filters—An update. Toxicology. 2004;205(1):113-122
- [14] Søeborg T, Basse LH, Halling-Sørensen B. Risk assessment of topically applied products. Toxicology. 2007;236(1):140-148
- [15] Christen V, Zucchi S, Fent K. Effects of the UV-filter 2-ethyl-hexyl-4-trimethoxycinnamate (EHMC) on expression of genes involved in hormonal pathways in fathead

minnows (*Pimephales promelas*) and link to vitellogenin induction and histology. Aquatic Toxicology. 2011;**102**(3):167-176

- [16] Kim S, Jung D, Kho Y, Choi K. Effects of benzophenone-3 exposure on endocrine disruption and reproduction of Japanese medaka (*Oryzias latipes*)—A two generation exposure study. Aquatic Toxicology. 2014;155:244-252
- [17] Ponzo OJ, Silvia C. Evidence of reproductive disruption associated with neuroendocrine changes induced by UV-B filters, phtalates and nonylphenol during sexual maturation in rats of both gender. Toxicology. 2013;**311**(1):41-51
- [18] Blüthgen N, Meili N, Chew G, Odermatt A, Fent K. Accumulation and effects of the UV-filter octocrylene in adult and embryonic zebrafish (*Danio rerio*). Science of the Total Environment. 2014;476:207-217
- [19] Vidal L, Chisvert A, Canals A, Salvador A. Ionic liquid-based single-drop microextraction followed by liquid chromatography-ultraviolet spectrophotometry detection to determine typical UV filters in surface water samples. Talanta. 2010;81(1):549-555
- [20] Pedrouzo M, Borrull F, Marcé RM, Pocurull E. Stir-bar-sorptive extraction and ultrahigh-performance liquid chromatography-tandem mass spectrometry for simultaneous analysis of UV filters and antimicrobial agents in water samples. Analytical and Bioanalytical Chemistry. 2010;397(7):2833-2839
- [21] Meinerling M, Daniels M. A validated method for the determination of traces of UV filters in fish using LC–MS/MS. Analytical and Bioanalytical Chemistry. 2006;**386**(5):1465-1473
- [22] Tarazona I, Chisvert A, León Z, Salvador A. Determination of hydroxylated benzophenone UV filters in sea water samples by dispersive liquid–liquid microextraction followed by gas chromatography–mass spectrometry. Journal of Chromatography A. 2010; 1217(29):4771-4778
- [23] Zhang Y, Lee HK. Determination of ultraviolet filters in water samples by vortex-assisted dispersive liquid–liquid microextraction followed by gas chromatography–mass spectrometry. Journal of Chromatography A. 2012;1249:25-31
- [24] Benedé JL, Chisvert A, Salvador A, Sánchez-Quiles D, Tovar-Sánchez A. Determination of UV filters in both soluble and particulate fractions of seawaters by dispersive liquid– liquid microextraction followed by gas chromatography–mass spectrometry. Analytica Chimica Acta. 2014;812:50-58
- [25] Maijó I, Fontanals N, Borrull F, Neusüß C, Calull M, Aguilar C. Determination of UV filters in river water samples by in-line SPE-CE-MS. Electrophoresis. 2013;34(3):374-382
- [26] Dimpe KM, Nomngongo PN. Current sample preparation methodologies for analysis of emerging pollutants in different environmental matrices. TrAC, Trends in Analytical Chemistry. 2016;82:199-207
- [27] Bizkarguenaga E, Ros O, Iparraguirre A, Navarro P, Vallejo A, Usobiaga A, Zuloaga O. Solid-phase extraction combined with large volume injection-programmable temperature vaporization–gas chromatography–mass spectrometry for the multiresidue

determination of priority and emerging organic pollutants in wastewater. Journal of Chromatography A. 2012;**1247**:104-117

- [28] Wells MJM. Essential guides to method development in solid-phase extraction. In: Wilson ID, Adlard ER, Cooke M, Poole CF, editors. Encyclopedia of Separation Science. Vol. 10. London: Academic Press; 2000. pp. 4636-4643
- [29] Sobańska AW, Kałębasiak K, Pyzowski J, Brzezińska E. Quantification of sunscreen benzophenone-4 in hair shampoos by hydrophilic interactions thin-layer chromatography/ densitometry or derivative UV spectrophotometry. Journal of Analytical Methods in Chemistry. 2015;2015:7. Article ID 695658. DOI:10.1155/2015/695658
- [30] Nomngongo PN, Ngila JC, Kamau JN, Msagati TA, Moodley B. Preconcentration of molybdenum, antimony and vanadium in gasolsine samples using Dowex 1-x8 resin and their determination with inductively coupled plasma–optical emission spectrometry. Talanta. 2013;110:153-159
- [31] Nomngongo PN, Ngila JC. Functionalized nanometer-sized alumina supported microsolid phase extraction coupled to inductively coupled plasma mass spectrometry for preconcentration and determination of trace metal ions in gasoline samples. RSC Advances. 2014;4(86):46257-46264
- [32] Ma T, Li Z, Niu Q, Li Y, Zhou W. Double dispersant-assisted ionic liquid dispersive liquid–liquid microextraction coupled with capillary electrophoresis for the determination of benzophenone-type ultraviolet filters in sunscreen cosmetic product. Electrophoresis. 2015;36(20):2530-2537
- [33] Moeder M, Schrader S, Winkler U, Rodil R. At-line microextraction by packed sorbent-gas chromatography-mass spectrometry for the determination of UV filter and polycyclic musk compounds in water samples. Journal of Chromatography A. 2010; 1217(17):2925-2932
- [34] Magi E, Di Carro M, Scapolla C, Nguyen KT. Stir bar sorptive extraction and LC–MS/ MS for trace analysis of UV filters in different water matrices. Chromatographia. 2012; 75(17-18):973-982
- [35] Vela-Soria F, Ballesteros O, Zafra-Gómez A, Ballesteros L, Navalón A. A new method for the determination of benzophenone-UV filters in human serum samples by dispersive liquid–liquid microextraction with liquid chromatography–tandem mass spectrometry. Talanta. 2014;121:97-104
- [36] Kotnik K, Kosjek T, Krajnc U, Heath E. Trace analysis of benzophenone-derived compounds in surface waters and sediments using solid-phase extraction and microwaveassisted extraction followed by gas chromatography–mass spectrometry. Analytical and Bioanalytical Chemistry. 2014;406(13):3179-3190
- [37] Liao C, Kannan K. Widespread occurrence of benzophenone-type UV light filters in personal care products from China and the United States: An assessment of human exposure. Environmental Science & Technology. 2014;48(7):4103-4109

- [38] Kameda Y, Kimura K, Miyazaki M. Occurrence and profiles of organic sun-blocking agents in surface waters and sediments in Japanese rivers and lakes. Environmental Pollution. 2011;**159**(6):1570-1576
- [39] Amine H, Gomez E, Halwani J, Casellas C, Fenet H. UV filters, ethylhexyl methoxycinnamate, octocrylene and ethylhexyl dimethyl PABA from untreated wastewater in sediment from eastern Mediterranean river transition and coastal zones. Marine Pollution Bulletin. 2012;64(11):2435-2442
- [40] Mottaleb MA, Usenko S, O'Donnell JG, Ramirez AJ, Brooks BW, Chambliss CK. Gas chromatography–mass spectrometry screening methods for select UV filters, synthetic musks, alkylphenols, an antimicrobial agent, and an insect repellent in fish. Journal of Chromatography A. 2009;1216(5):815-823
- [41] Langford KH, Reid MJ, Fjeld E, Øxnevad S, Thomas KV. Environmental occurrence and risk of organic UV filters and stabilizers in multiple matrices in Norway. Environment International. 2015;80:1-7
Study on Magnetic Materials for Removal of Water Pollutants

Manoj Sharma, Pankaj Kalita, Kula Kamal Senapati and Ankit Garg

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/intechopen.75700

Abstract

Water is a primary element for all living things, and we need water for each and every dayto-day activity related to agricultural, industrial, and domestic cares and, thus, its quality influences all aspects of human life including energy, food, health, and economy. Safe drinking water is our primary need to protect our life and thus developing efficient and affordable techniques for water treatment to access potable water to the humanity. Water pollution is one of the severe environmental and health problems worldwide. Pollutants in water can be of organic, inorganic, heavy metals, microbial, and radioactive species, which may be in different forms viz. suspended, dissolved, or dispersed materials. The water quality is mainly affected by industrial discharges, agricultural activities, mismanagement of hazardous materials, etc. Nowadays, nanotechnology offers the possibility of an efficient removal of water pollutants including metals, organic dyes, bacteria, parasites, etc. Magnetic nanomaterials like iron oxide (Fe_3O_4) are very promising materials used in water decontamination particularly for heavy metals and dyestuffs because of their ease of separation through external magnet, high surface area, unique morphology as well as their high stability. These materials can be used as adsorbent, photocatalyst, and coagulating agents for water remediation based on their composite materials or surface functionalities.

Keywords: magnetic nanomaterials, water treatment, water pollution

1. Introduction

Water, i.e., 2 mol of hydrogen and 1 mol of oxygen or simply called H_2O plays a vital role in our everyday life. Without water no one can think of life on earth. About 71% of earth is covered with water. Of that 71% water about 97% resides in oceans and only 3% of water is

IntechOpen

© 2018 The Author(s). Licensee IntechOpen. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.

fresh water (**Figure 1a**) and of that 3% about 68.7% of freshwater is locked up in icecaps and glaciers (**Figure 1b**), and it is quite a surprising fact that almost all the remaining fresh water is below the ground. Of all the freshwater on the surface of earth only 0.3% is contained in fresh lakes, and rivers [1].

As the population is increasing day by day the water availability per capita is decreasing. So the challenge of limited amount of freshwater and its decreasing per capita availability is an issue of concern but another major challenge is water pollution that has not only environmental impact but also have a major effect on human health. As per the statistics 783 million people do not have access to clean and safe water worldwide [2]. Around 319 million people in Sub-Saharan Africa are without the access to improved reliable drinking water sources. One in nine people worldwide do not have access to safe and clean drinking water. 443 million school days are lost each year due to water-related diseases [3]. In developing countries, as much as 80% of illnesses are linked to poor water and sanitation conditions [4]. 2.6 billion people in the world lack adequate sanitation and which contributes to about 10% of the global disease burden [5]. Half of the world's hospitals beds are filled with people suffering from a water-related disease [6].



There are various type of illnesses due to water and are summarized in Table 1.

Figure 1. (a) Water distribution on the earth and (b) freshwater distribution on the earth.

Type of diseases	Source	Example	References
Water borne	Human or animal waste	Cholera, typhoid, and dysentery,	[7]
Water washed	Lack of clean water for washing	Skin and eye infections	[8]
Water based	Organism developed in water and then become parasite	Schistosomiasis	[9]
Water-related insect vector	Mosquitoes breed in water	Malaria and dengue	[9]

Table 1. Various types of illness due to water.

As we have seen that the contaminated water has a very bad impact on human health even some heavy metals that if taken for long time it can cause cancer such as arsenic is considered as one the carcinogenic contaminant in water. After understanding the health issues related to contaminant water there is need to understand the source of contamination and major contaminants that pollute the water.

In recent years, magnetic materials have been potentially used for removal of water pollutants, particularly organic contaminates (dyes, chlorinated hydrocarbons, aromatics), pesticides, as well as heavy metals [10]. There are a large number of techniques available for water treatment for safe drinking water including adsorption, precipitation, solvent extraction, ion exchange, reverse osmosis, membrane separation, evaporation, and photocatalysis. The development of nanoscience and nanotechnology shows their potentiality in removing toxic elements from water bodies with better water treatment process. The design and development of nanomaterials which belong to the size range of 1–100 nm exhibiting unique properties as compared to the bulk materials leads to the enormous improvements in many sectors including, health, manufacturing, electronics, environmental remediation as well. The magnetic nanomaterials (paramagnetic or ferromagnetic or superparamagnetic) with tailored surface chemistry have already expanded their scope of application in water treatment. In this chapter, various processes of drinking water treatment and waste water treatment using advanced magnetic materials in removing toxic metal ions, organic and inorganic solutes, bacteria and viruses has been discussed.

1.1. Major sources of water contamination

There are many sources of drinking water and the main sources are ground water, lakes, canals, reservoirs, rain water, fog water and sea water. These sources are contaminated in different ways and broadly the source of contamination can be divided in to two categories:

- Direct sources or point sources
- Indirect sources or non-point sources

Direct sources basically include effluent from industries, treatment plants, refineries, factories, etc. However, indirect sources or non-point sources include the water contamination entering to the water body through a number of processes, e.g., while putting the fertilizers and pesticides to the agricultural field, the elements presents in the chemical percolates down to the groundwater and ultimately pollute the water.

1.2. Different types of water contaminants

There are various types of water contaminants, however in the following subsections organic contaminants, inorganic contaminants and pathogens are briefly discussed.

1.2.1. Organic contaminants

Organic contaminants present in drinking water create severe problem on human health. Pollution by organic chemicals in water bodies occurs by various mechanisms. Industrial waste containing various organic chemical contaminants pollutes the water bodies. Volatile organic compounds (VOCs), pesticides, phenolic compounds, phthalates, and nitrogen-containing compounds, are often detected in polluted water [11]. Many of these compounds have been found to be carcinogenic, even in very low concentrations. WHO Guidelines for drinking water quality, levels are set for 28 organic constituents (i.e., microcystin-LR, chlorinated alkanes, chlorinated benzenes and miscellaneous), 33 pesticides, and 9 disinfectant by-products, due to their health effects on humans [12]. It is noteworthy to mention that, occurrence of pharmaceutical and personal care products and perfluoroalkyl acids in aquatic environment has been recognized as emerging issue in environmental chemistry [13].

1.2.2. Inorganic contaminants

Inorganic contaminants include metals, salts and other compounds that do not contain carbon. Many of them are naturally occurring and should be considered as an integral part of those waters, e.g., calcium carbonate and bicarbonate in hard water. The metal ions such as Hg(II), Pb(II), Cr(III), Cr(VI), Ni(II), Co(II), Cu(II), Cd(II), Ag(I), As(V) and As(III) are toxic from eco toxicological point of view. Besides, the pollution by the radioactive elements is of major concern looking into their long-term hazardous impacts.

1.2.3. Pathogens

Pathogens such as bacteria, viruses and parasites may be present in very low concentration in drinking water; but cause many infectious diseases and are considered as one of the major risk factors with drinking water safety [14]. The pathogenic microorganisms enter in to water body through sewage discharge as a major source or through the wastewater from industries like slaughterhouses. Water-borne pathogens have been the causes of many disease outbreaks such as diarrhea, cholera, gastro-intestinal illness [15]. The recurrence of water-borne pathogens is due to a number of reasons like heavy water contamination, population explosion, change in potable water treatment methods, globalization of commerce and travel. It has been made possible to detect pathogen based water contamination to a large extent owing to the improved methods for detection and source tracking [16, 17]. The most serious health risk is related with ingestion of water which is contaminated with fecal matter and the discharge of wastewater into various ambient water bodies is what contributes to the multiplication of numbers of such pathogens (bacteria, viruses, protozoa and helminthes) [18].

2. Techniques for water treatment

Water treatment is defined as the removal of the above contaminants using some specific process. In most of the water treatment processes, conventional adsorption process with activated carbon is adopted and the adsorption capacity is substantially decreased in presence of high concentration of organic matters in water where the active sites are mostly occupied by these materials. In recent times, there are various technologies have been employed for the removal of water contaminants such as filtration (ceramic, bio sand, membrane, and activated carbon based filtration), heat and UV radiation, chemical treatment (coagulation-flocculation,

chemical disinfection), and desalination (reverse osmosis, distillation). The various techniques in water treatment can categorized into following six classes [19]:

- Adsorption
- Biotechnology
- Catalytic processes
- Membrane processes
- Ionizing radiation processes
- Magnetically assisted processes.

There are specific advantages and disadvantages for a particular process. The nanotech based processes are promising option in current water treatment processes because of their target specificity, ease of separation, high adsorption per unit area, as well as less maintenance.

3. Magnetic materials for water treatment

There has been increased interest in using magnetic materials in water treatment which are basically composed of magnetic core of iron oxides organic compounds, carbon materials, etc. Recently, nanomaterials in different shapes, morphologies, forms, e.g., metal-containing nanoparticles, carbonaceous nanomaterials, zeolites, dendrimers, carbon nanotubes, nanofibers have been used for water purification [20]. However, the difficulty arises in using these materials is the separation of solid materials from liquid and which is more difficult as the particle size decreases in nanoscale. On the other hand, the using of magnetic, particularly the magnetic nanoparticles (MNPs) materials have the advantage of magnetic filtration in separation of solid from liquid and are more efficient [21].

However solid/liquid (S/L) separation is more difficult as the particle size decreases. On the other side, in case of magnetic sorbents based on Fe oxides, the magnetic filtration may be applied for S/L separation. Furthermore, the removal of particles from solution with the use of magnetic fields is more selective and efficient (and often much faster) than centrifugation or filtration (Yauvuz et al.) [21]. Here are the advantages of using MNPs adsorbent for water treatment processes:

- Small size and thus high surface to volume ratio
- Solid/liquid separation through magnetic filtration is selective, faster than centrifugation and filtration techniques
- Reusability
- Greater biocompatibility
- Magnetic separation

3.1. Different types of magnetic materials for water treatment

Different types of magnetic materials have been synthesized and designed for development of advanced materials and applied effectively in widespread uses such as biomedicine, magnetic resonance imaging (MRI), catalysis, spintronics, robotics, engineering, environmental remediation, etc. [22] There are different synthesizing methods viz. co-precipitation, solvothermal, hydrothermal, microemulsion, sonochemical, etc. which determine their particle size, distribution, morphology, surface functionality, and magnetic properties and in turn of their various application [22]. Magnetic materials are made from mixtures of metals of iron, cobalt, nickel, and alloys and their oxides (of the type MFe_2O_4 , where M is a metal). Out of these materials, iron (zero valent iron) and its oxides, i.e., usually γ -Fe₂O₃ (maghemite) and Fe₃O₄ (magnetite) nanoparticles have attained significant interest in recent years and have been used for water treatment processes. The various composite magnetic materials such as Fe₃O₄@C [23, 24], Fe@SiO₂ [25], Fe₃O₄@TiO₂ [26], Fe₃O₄@PPO (poly(propylene oxide)), PEO (poly(ethylene oxide) [27], Fe₃O₄@PDA (polydopamine) [28], Fe₃O₄@PNIPAM (poly(N-isopropylacrylamide)) [29], Fe₃O₄@MIPs (molecularly imprinted polymer-encapsulated particles) [30], Fe₃O₄@CNTs (multiwalled carbon nanotubes) [31], Fe@CS (carbon spheres) [32], Fe/iron oxide-oxyhydroxide/rGO (grapheme) [33], etc. have been used for environmental applications. Singh and his co-workers synthesized a series of magnetic nanocomposites such as CoFe₂O₄–ZnS [34], Fe₃O₄@GTPs (green tea polyphenols) [35], Fe₃O₄-Cr₂O₃ [36], CoFe₂O₄-Cr₂O₃-SiO₂ [37] and applied for wastewater treatment.

In addition to their suitable magnetic properties, i.e., ferrimagnetic, ferromagnetic and superparamagnetic (nanoparticle size less than 10 nm), their synthesis procedure is simple and costeffective and they can be easily functionalized as desired for many applications. The size and shape and magnetism of these magnetic materials can be easily controlled based on their application and thus they can be easily dispersed in liquid medium and their stability can be retained for multiple uses. Moreover, these materials are non-toxic or less toxic, chemically inert, thermally stable as well as biocompatible.

3.2. Use of magnetic materials for clean water technology

Appearance of water pollution as a global threat demands the development of low-cost and reliable materials for effective waste water remediation. The magnetic materials have been used for clean water technology for both in laboratory as well as field scale [38, 39]. In recent years, iron oxide nanomaterials have been used as adsorbent or immobilizing agent and photocatalyst or the both depending on nature of contaminants in water [40].

3.2.1. MNPs for the removal of metals

Heavy metal contamination in water such as cadmium, zinc, lead, chromium, nickel, copper, vanadium, platinum, silver and titanium due to industrial activities is significantly increasing which is detrimental to human beings and animals. Magnetic nanomaterial adsorbents have been potentially used for removal of metallic ions such as Cr(VI), Cu(II), Co(II),Cd(II), As(V), As(III) and Hg(II) in water [41, 42] which are more effective as compared to micron size particles. The magnetic chelating resin based materials have been used for effective removal

of Cu(II), Co(II), and Ni(II) ions [43]. The magnetic hydrogels based on 2-acrylamine-2-methyl-1-propansulfonic acid can be used for removal of many heavy metal ions such Cd(II), Co(II), Fe (II), Pb(II), Ni(II), Cu(II) and Cr(III) from water in repeated cycles [44]. The Cu(II) can also be effectively removed by functionalized mesostructured silica containing magnetite [45]. The acrylate-based polymer composites with magnetite can be used in selective removal of heavy metals from water (selectivity: Cu > Cr > Zn > Ni) [46]. Layered double hydroxide (LDH) prepared from Fe³⁺ and Ni²⁺ shows good adsorption of As and subsequent magnetic separation [47]. The magnetic zeolite composites are used for decontamination of heavy metals from water [48]. The composite materials of mesoporous magnetic MCM-41 with aminopropyls are used for selective removal of As(V), and Cr(VI) in presence of Cu(II) [49].

3.2.2. MNPs for removal of organic contaminants

Magnetic nanoparticles are used as an adsorbent for the removal of various dyes and dyes stuff from aqueous solution. Removal of dyes from waste water has become a serious issue of concern because of its harmful impact on human. Dyes basically can be classified in to two categories, i.e., anionic dyes and cationic dyes.

3.2.3. MNPs for the removal of anionic dyes

Long et al. [50] synthesized Fe₃O₄@catechol/polyethylenimine (PEI) nanoparticles and tested for adsorption of three different kind of anionic dyes, i.e., methyl blue, orange G and amaranth and found the maximum adsorption capacities of 344.8, 192.3 and 146.2 mg/g, respectively. Saksornchai et al. [51] synthesized magnetite (Fe_3O_4) coated with cetyltrimethylammonium bromide (CTAB) and tested for the adsorption of anionic dye Congo red (CR) removal. They found maximum adsorption capacity for CR dye to be 93.46 mg/g. Faraji et al. [52] synthesized triazine-based nitrogen-rich network-modified magnetic nanoparticles were synthesized for the adsorption of methyl orange. Sahraei et al. [53] reported the synthesis of magnetic bio-sorbent hydrogel beads based on modified gum tragacanth/graphene oxide for the removal of heavy metals and dyes from water. They found the adsorption capacity of 101.7 mg/g for Congo red dye. Ge et al. [54] fabricated Fe@MgO magnetic nanocomposites for the removal of heavy metal ions and dye from water. They found that the synthesized nanocomposite showed excellent adsorption capacity of 6947.9 mg/g for methylene orange. Wu et al. [55] fabricated multifunctional magnetic nanoparticle core covered with polyethylenimine (PEI) derived quaternary ammonium compounds (QAC) corona through electrostatic attraction for the removal of dyes and metal ion adsorption. The adsorption results corresponding to synthesized nanoparticle showed the maximum adsorption capacity of 653 mg/g for AF as a representative of dyes. Konicki et al. [56] synthesized Fe@graphite core shell nanocomposite for the removal of anionic dyes from aqueous solution. The synthesized nanoparticles were tested for the adsorption of two anionic dyes namely acid red 88 (AR88) and direct orange 26 (DO26) and the maximum adsorption capacity was found to be 63.7 mg/g and 42.7 for AR88 and DO26, respectively. Zhang et al. [57] synthesized the Fe_3O_4 nanoparticle modified with 3-glycidoxypropyltrimethoxysilane (GPTMS) and poly-lysine (P-Lys). They found that the synthesized MNPs could effectively remove anionic dyes including methyl blue (MB), orange I (OR-I), amaranth (AM) and acid red 18 (AR-18) from water solution.

3.2.4. MNPs for the removal of cationic dyes

Cationic dyes are most toxic because they can easily interact with negatively charged cell membrane surfaces, and also they can enter in to the cells and can concentrate in cytoplasm (Bayramoglu et al.) [58]. Ge et al. [59] have studied the adsorption of cationic dyes such as crystal violet, methylene blue and alkali blue 6B from aqueous solutions by use of polymermodified magnetic nanoparticles. The cationic dyes could be quickly removed from water solution with high efficiency at pH 5–12. More significantly, the MNP showed high efficiency as a reusable adsorbent for fast and convenient removal of cationic dyes from water solution. Yan et al. [60] have synthesized full biodegradable magnetic adsorbent based on glutamic acid modified chitosan and silica coated Fe₃O₄ nanoparticles for removal of three different kinds of cationic dyes, methylene blue, crystal violet and cationic light yellow 7GL, from aqueous solutions. Chen et al. [61] have prepared magnetic adsorbent by fabrication of chitosan/ polyacrylic acid multilayer onto magnetic Fe_3O_4 microspheres for removal of adsorption of two cationic dyes, methylene blue and crystal violet from aqueous solution. Amiri et al. [62] synthesized cobalt ferrite silica magnetic nanocomposite for the adsorption of Malachite green dye and found the adsorption capacity of 75.5 mg/g for that dye. Li et al. [63] synthesized wettable magnetic hypercrosslinked microporous nanoparticle for the water treatment. The synthesized nanoparticle consists of microporous organic polymer which combine sodium acrylate functionalized hypercrosslinked polymer with magnetic Fe₃O₄ nanoparticle to form a hybrid. They tested the synthesized hybrid for the adsorption of Rhodamine B dye and found the maximum adsorption capacity of 216 mg/g. Singh et al. [64] had synthesized the superparamagnetic nanoparticles coated with green tea polyphenol by wet chemical method. They found that the particles have a very high adsorption capacity of (7.25 mg/g) for removal of methylene blue (MB) dye in wastewater treatment. Li et al. [65] synthesized magnetic peach gum bead bio-sorbent for the adsorption of MB dye and found the maximum adsorption capacity of 231.5 mg/g.

3.2.5. MNPs for the removal of pharmaceutical products

The presence of pharmaceuticals such as antibiotics, anticonvulsants, antipyretics drugs, hormones in surface and ground water possesses a major environmental challenge. Their contamination even at trace amount is a serious concern to the aquatic organisms as well.

Attia et al. [66] synthesized magnetic nanoparticles coated zeolite for the adsorption of pharmaceutical compounds from aqueous solution. They found that the synthesized magnetic nanoparticles can remove more that 95% of PPCPs in 10 min. Reddy et al. [67] reviewed spinal ferrite nanoparticles and found that SF and its derivatives can be used for remediation of various pollutants. Nadim et al. [68] Synthesized gallic acid coated magnetic nanoparticles (GA-MNP) and used as a photocatalyst for degradation of meloxicam; a commonly prescribed nonsteroidal anti-inflammatory drug.

Recently, M. Hayasi and his coworker described the use of magnetic poly (styrene-2-acrylamido-2-methyl propanesulfonic acid) (St-AMPS) as adsorbent for removal of the pharmaceuticals viz. ceftriaxone sodium, diclofenac sodium, and atenolol from water [69].

Liu [70] discussed the use of various magnetic nanoparticles (MNPs) in pharmaceutical removal from waste water their recent review which are outlined below (**Table 2**).

3.2.6. MNPs for removal of pathogens

Xu et al. [82] demonstrated that poly-allylamine-hydrochloride (PAAH) stabilized magnetic nanoparticles are powerful tools to remove pathogenic bacteria from drinking water with high efficiency and no significant toxicity was observed in the MNPs treated water. Over 99.5% of the pathogens (four main pathogens viz. *Escherichia, Acinetobacter, Pseudomonas* and *Bacillus*) can be removed when the bacterial count was less than 105 CFU/mL.

Zhang et al. [83] synthesized magnetic nanoparticle coated with Cu doped MgO through a hydrophilic carbon layer (Fe₃O₄@C@MgO-Cu). They found its potential application as disinfectant in water purification by examining the antibacterial activity of the Fe₃O₄@C@MgO-Cu composite toward Gram-negative *Escherichia coli* and Gram-positive *Staphylococcus aureus*.

Zhang et al. [84] synthesized magnetic poly-N,N'-[(4,5-dihydroxy-1,2-phenylene)bis(methylene)]bisacrylamide) (POHABA)-based core-shell nanostructure on the Fe₃O₄ core surface (Fe₃O₄@POHABA). The magnetic nanocomposite, Fe₃O₄@POHABA can be used in domestic water treatment against bacterial pathogens.

Magnetic materials used	Pharmaceuticals present in water	Processes involved for decontamination	References				
Nano zero-valent iron (nZVI)	Carbamazepine	Adsorption, chemical degradation (Fenton oxidation)	[71]				
Nano zero-valent iron (nZVI) and PEG and zeolite supported nZVI	Amoxicillin, ampicillin	Adsorption	[72]				
Zero valent iron (ZVI)	Carbamazepine	Chemical degradation	[73]				
	Diazepam	Chemical degradation	[74]				
Maghemite (Fe ₂ O ₃) core confined in a silica porous layer	Atenolol, gemfibrozil, and sulfamethoxazole	Adsorption	[75]				
MFe_2O_4 (M = Fe, Mn, Co, Zn)	Tetracycline, oxytetracycline, and chlortetracycline	Adsorbent	[76]				
MnFe ₂ O ₄ /(activated carbon)AC	Sulfamethoxazole	Adsorption	[77]				
$MgFe_2O_4/\gamma$ - Fe_2O_3	Minocycline	Adsorption	[78]				
$\rm Fe_3O_4$ MNPs in ultrasound (US)/H_2O_2 system		Sono-degradation	[79]				
$Fe_3O_4@\alpha-MnO_2$ microspheres	Ciprofloxacin	Catalytic degradation	[80]				
Fe-ZnO	Diclofenac	Photocatalytic degradation	[81]				

Rana et al. [85] synthesized ferromagnetic Ni-doped ZnO nanoparticles and applied as an antibacterial agent to control the growth of bacterial pathogens. They found the as synthesized material to be very effective against water related bacteria such as *E. coli* and *V. cholera*.

Table 2. Magnetic nanomaterials used for removal of pharmaceuticals.

Shukla et al. [86] synthesized the iron oxide nanoparticles coated with chitosan oligosaccharide and used for the removal of pathogenic protozoan cysts, entamoeba cyst (which causes amebiasis) from water. They found that *E. histolytica* can be efficiently captured using the magnetic nanoparticles from contaminated water.

Zhang et al. [87] synthesized Fe_3O_4 nanoparticles surrounded with polyethylenimine-derived corona and found to be efficient in capturing the pathogens and heavy metals.

Zhan et al. [88] synthesized the amine-functionalized magnetic nanoparticle (Fe₃O₄-SiO₂-NH₂) and used for rapid removal of pathogenic bacteria and viruses. The magnetic materials can be effectively used to capture a wide range of pathogens including various bacteria such as *S. aureus*, *E. coli* O157:H7, *P. aeuginosa, Salmonella,* and *B. subtilis*.

Park et al. [89] developed a novel magnetic hybrid colloid (MHC) decorated with varying sized Ag nanoparticles. The MHC was prepared as a cluster of superparamagnetic Fe_3O_4 coated with silica shell. The MHC decorated with the Ag nanoparticle of 30 nm size (Ag30@MHC) exhibited the highest antimicrobial efficacy toward *E. coli* CN13 (6-log reduction) and the bacteriophage MS2 (2–3 log reduction).

4. Magnetic water treatment

Magnetic water treatment (MWT) is a new technique which is promising in environmental remediation in addition to its increased application in the area of medicine, agriculture and industrial process. The water molecules acquire unique physicochemical characteristics under the influence of the magnetic fields (MFs). It is a non-chemical method and the water molecules undergo change from their cluster of many loosely bound water molecules into very small, uniform and hexagonally organized clusters of molecules under the magnetic treatment [90]. These features of the magnetized water prevent the polluting agents to enter in to its cluster and also make their easy passage through the cells of plant as well as of animals. Moreover, the MFs have antimicrobial activity on water. Therefore, the magnetized water molecules depict as bio-friendly and eco-friendly compounds to environmental management as well as to plant and animal cells.

5. Sustainability of magnetic nanoparticles

While taking sustainability in to consideration there are various factors over which it depends. Important factors include environmental impact on use of MNPs, toxicity associated with the MNPs, reusability, reactivity, adsorption capacity, biocompatibility, stability, etc. For example, uncoated nanoparticles are associated with some toxicity while coating can help them to make non-toxic. Similarly, their regeneration and reusability are the main factors for making them technically more viable and economically sustainable materials for commercial uses.

6. Conclusions

In this chapter the removal of water pollutants by using magnetic materials of zero valent iron, magnetite (Fe₃O₄), maghemite (γ -Fe₂O₃) as adsorbent, photocatalyst and coagulants have been described. The MNPs have been used in removal of water pollutants through their various surface functionalities (e.g., coating with polyphenols, amino acids, sugars, alkaloids, terpenoids, proteins, carbonyl, carboxyl, carbon, polysaccharides, and semiconductors) with desired size and shapes, and magnetic behavior. Looking in to the fast development of magnetic materials in different technological and scientific fields, magnetic nanomaterials appear to be extremely promising for water and wastewater treatment. The waste water treatment methods using these materials are fast, non-toxic, and eco-friendly as compared to the available physic-chemical treatments which make it attractive for materializing commercially. Their magnetic nature makes them attractive for waste water treatment because of their easy separation from aqueous medium after purification and can be reused in repeated treatment cycles. However, research for bulk production, controlling morphology, optimizing surface functionality and their stability, and biocompatibility should be essentially considered prior to commercial application from laboratory scale. Moreover, further studies needs to be addressed to detail mechanism of magnetic nanomaterials in water treatment. The magnetic nanoparticles and their composites with their high surface to volume ratio offer more surfaces for chemical as well as physical adsorption and thus show high reactivity which gives the prospects of using these materials in large scale removal of emerging water pollutants.

Author details

Manoj Sharma¹, Pankaj Kalita², Kula Kamal Senapati³* and Ankit Garg⁴

*Address all correspondence to: kulasenapati@gmail.com

1 Centre for Rural Technology, Indian Institute of Technology Guwahati, Guwahati, Assam, India

2 Centre for Energy, Indian Institute of Technology Guwahati, Guwahati, Assam, India

3 Central Instruments Facility, Indian Institute of Technology Guwahati, Guwahati, Assam, India

4 Department of Civil and Environmental Engineering, Shantou University, China

References

[1] Okun DA, Wang LK, Shammas NK. Water supply and distribution and wastewater collection. United States of America: John Wiley and Sons; 2010

- [2] WHO, UNICEF. Millennium Development Goals: Progress on Sanitation and Drinking-Water: 2010 Update Report, Geneva: WHO/UNICEF Joint Monitoring Programme for Water Supply; 2010. ISBN: 978 92 4 156395 6
- [3] Corcoran E, editor. Sick Water? The Central Role of Wastewater Management in Sustainable Development: A Rapid Response Assessment. UNEP (United Nations Environment Programme). UN-HABITAT, Nairobi: Kenia publishing; 2010
- [4] United Nations. Statement by Secretary General Koffi Annan. June 2003. Available from: http://www.un.org/News/Press/docs/2003/sgsm8707.doc.htm
- [5] Maral D, Lane J, Scott B, Trouba D. Sanitation and health. PLoS Medicine. 2010;7(11): e1000363
- [6] UNEP/UN-Habitat. Sick Water? The Central Role of Wastewater Management in Sustainable Development. Available from: http://www.grida.no/publications/rr/sickwater/
- [7] Ashbolt NJ. Microbial contamination of drinking water and disease outcomes in developing regions. Toxicology. 2004;198(1):229-238
- [8] Tebbutt THY. Principles of Water Quality Control. Great Britain: Butterworth-Heinemann; 1997
- [9] Gleick PH. Dirty-Water: Estimated Deaths from Water-Related Diseases 2000–2020. Oakland: Pacific Institute for Studies in Development, Environment, and Security; 2002
- [10] Mohammed L, Gomaa HG, Ragab D, Zhu J. Magnetic nanoparticles for environmental and biomedical applications: A review. Particuology. 2017;30:1-14
- [11] Tsuchiya Y. Organical chemicals as contaminants of water bodies and drinking water. Water Quality and Standards – Volume II. 2010;7:150
- [12] World Health Organization. Guidelines for Drinking-water Quality [electronic resource]: incorporating 1st and 2nd addenda, Vol. 1, Recommendations 3rd ed. Geneva; 2008. ISBN: 978 92 4 154761 1 (WEB version)
- [13] Kubota S, Tsuchiya Y, editors, Water Quality and Standard. Vol. II. Encyclopedia of Life Support Systems (EOLSS) Publishing; 2009. p. 151. ISBN: 978-1-84826-031-3 (eBook)
- [14] Park S, Park HH, Ko YS, Lee SJ, Woo K, Ko G. Disinfection of various bacterial pathogens using novel silver nanoparticle-decorated magnetic hybrid colloids. Science of the Total Environment. 2017;609:289-296
- [15] Craun GF, Fraun MF, Calderon RL, Beach MJ. Waterborne outbreaks reported in the United States. Journal of Water and Health. 2006;4:19-30
- [16] Ingerson-Mahar M, Reid A. Microbes in Pipes: The Microbiology of the Water Distribution System A Report on an American Academy of Microbiology Colloquium. Boulder, CO, USA: ASM Academy; 2012. p. 26
- [17] Pandey PK, Kass PH, Soupir ML, Biswas S, Singh VP. Contamination of water resources by pathogenic bacteria. AMB Express. 2014;4:51

- [18] Bitton G. Microbiology of Drinking Water Production and Distribution. 1st ed. Hoboken, NJ, USA: John Wiley & Sons, Inc.; 2014. p. 312
- [19] Ambashta RD, Sillanpaa M. Water purification using magnetic assistance: A review. Journal of Hazardous Materials. 2010;180:38-49
- [20] Tiwari DK, Behari J, Sen P. Application of nanoparticles in waste water treatment. World Applied Sciences Journal. 2008;3(3):417-433
- [21] Yauvuz CT, Mayo JT, Yu WW, Prakash A, Falkner JC, Yean S, Cong L, Shipley HJ, Kan A, Tomson M, Natelson D, Colvin VL. Low-field magnetic separation of monodisperse Fe₃O₄ nanocrystals. Science. 2006;**314**(5801):964-967
- [22] Li X, Xu G, Liu Y, He T. Magnetic Fe₃O₄ nanoparticles: Synthesis and application in water treatment. Nanoscience & Nanotechnology-Asia. 2011;1:14-24
- [23] Bao X, Qiang Z, Chang JH, Ben W, Qu J. Synthesis of carbon-coated magnetic nanocomposite (Fe₃O₄@C) and its application for sulfonamide antibiotics removal from water. Journal of Environmental Sciences. 2014;26:962-969
- [24] Zhou L, Zhang G, Tian J, Wang D, Cai D, Wu Z. Functionalized Fe₃O₄@C nanospheres with adjustable structure for efficient hexavalent chromium removal. ACS Sustainable Chemistry & Engineering. 2017;5(11):11042-11050
- [25] Li Y, Jin Z, Li T, Xiu Z. One-step synthesis and characterization of core-shell Fe@SiO₂ nanocomposite for Cr(VI) reduction. Science of the Total Environment. 421-422, 260-266
- [26] Xin T, Ma M, Zhang H, Gu J, Wang S, Liu M, Zhang Q. A facile approach for the synthesis of magnetic separable Fe₃O₄@TiO₂, core shell nanocomposites as highly recyclable photocatalysts. Applied Surface Science. 2014;288:51-59
- [27] Moeser GD, Roach KA, Green WH, Laibinis PE, Hatton TA. Water-based magnetic fluids as extractants for synthetic organic compounds. Industrial & Engineering Chemistry Research. 2002;41:4739-4749
- [28] Wang Y, Wang S, Niu H, Ma Y, Zeng T, Cai Y, Meng Z. Preparation of polydopamine coated Fe3O4 nanoparticles and their application for enrichment of polycyclic aromatic hydrocarbons from environmental water samples. Journal of Chromatography. A. 2013
- [29] Dionigi C, Pineiro Y, Riminucci A, Banobre M, Rivas J, Dediu V. Regulating the thermal response of PNIPAM hydrogels by controlling the adsorption of magnetite nanoparticles. Applied Physics A. 2014;114:585-590
- [30] Xia X, Lai EPC, Ormeci B. Ultrasonication-assisted synthesis of molecularly imprinted polymer-encapsulated magnetic nanoparticles for rapid and selective removal of 17bestradiol from aqueous environment. Polymer Engineering and Science. 2012;52:1775-1783
- [31] Luo YB, Yu QW, Yuan BF, Feng YQ. Fast microextraction of phthalate acid esters from beverage, environmental water and perfume samples by magnetic multi-walled carbon nanotubes. Talanta. 2012;90:123-131

- [32] Sun H, Zhou G, Liu S, Ang HM, Tad_e MO, Wang S. Nano-Fe⁰ encapsulated in microcarbon spheres: Synthesis, characterization, and environmental applications. ACS Applied Materials & Interfaces. 2012;4:6235-6241
- [33] Karamani AA, Douvalis AP, Stalikas CD. Zero-valent iron/iron oxide-oxyhydroxide/ graphene as a magnetic sorbent for the enrichment of polychlorinated biphenyls, polyaromatic hydrocarbons and phthalates prior to gas chromatography mass spectrometry. Journal of Chromatography A. 2013;1271:1-9
- [34] Senapati KK, Borgohain C, Phukan P. CoFe₂O₄–ZnS nanocomposite: A magnetically recyclable photocatalyst. Catalysis Science & Technology. 2012;2:2361-2366
- [35] Singh KK, Senapati KK, Sarma KC. Synthesis of superparamagnetic Fe₃O₄ nanoparticles coated with green tea polyphenols and their use for removal of dye pollutant from aqueous solution. Journal of Environmental Chemical Engineering. 2017;5:2214-2221
- [36] Singh KK, Senapati KK, Sarma KC. Newly developed Fe₃O₄-Cr₂O₃ magnetic nanocomposite for photocatalytic decomposition of 4-chlorophenol in water. Journal of Environmental Sciences. 2017;52:333-340
- [37] Senapati KK, Borgohain C, Phukan P. Photocatalytic degradation of methylene blue in water using CoFe₂O₄-Cr₂O₃-SiO₂ fluorescent magnetic nanocomposite. Journal of Molecular Catalysis A: Chemical. 2011;346:111-116
- [38] White BR, Stackhouse BT, Holcombe JA. Magnetic γ-Fe₂O₃ nanoparticles coated with poly-L-cysteine for chelation of As(III), Cu(II), Cd(II), Ni(II), Pb(II) and Zn(II). Journal of Hazardous Materials. 2009;161(2–3):848-853
- [39] Girginova PI, Daniel-da-Silva AL, Lopes CB, Figueira P, Otero M, Amaral VS, et al. Silica coated magnetite particles for magnetic removal of Hg²⁺ from water. Journal of Colloid and Interface Science. 2010;345(2):234-240
- [40] Xu P, Zeng GM, Huang DL, Feng CL, Hu S, Zhao MH, Lai C, Wei Z, Huang C, Xie GX, Liu ZF. Use of iron oxide nanomaterials in wastewater treatment: A review. Science of the Total Environment. 2012;424:1-10
- [41] Tuutijarvi T, Lu J, Sillanpaa M, Chen G. As(V) adsorption on maghemite nanoparticles. Journal of Hazardous Materials. 2009;**166**(2–3):1415-1420
- [42] Sharma YC, Srivastava V, Singh VK, Kaul SN, Weng CH. Nano adsorbents for the removal of metallic pollutants from water and wastewater. Environmental Technology. 2009;30(6):583-609
- [43] Monier M, Ayad DM, Wei Y, Sarhan AA. Preparation and characterization of magnetic chelating resin based on chitosan for adsorption of Cu(II), Co(II), and Ni(II) ions. Reactive and Functional Polymers. 2010;70(4):257-266
- [44] Ozay O, Ekici S, Baran Y, Aktas N, Sahiner N. Removal of toxic metal ions with magnetic hydrogels. Water Research. 2009;43(17):4403-4411

- [45] Kim Y, Lee B, Yi J. Preparation of functionalized mesostructured silica containing magnetite (MSM) for the removal of copper ions in aqueous solutions and its magnetic separation. Separation Science and Technology. 2003;**38**(11):2533-2548
- [46] Phanapavudhikul P, Waters JA, de Ortiz ESP. Design and performance of magnetic composite particles for the separation of heavy metals from water. Journal of Environmental Science and Health, Part A. 2003;38(10):2277-2285
- [47] Nakahira A, Kubo T, Murase H. Synthesis of LDH-type clay substituted with Fe and Ni ion for arsenic removal and its application to magnetic separation. IEEE Transactions on Magnetics. 2007;43(6):2442-2444
- [48] Oliveira LCA, Petkowicz DI, Smaniotto A, Pergher SBC. Magnetic zeolites: A new adsorbent for removal of metallic contaminants from water. Water Research. 2004;38:3699-3704
- [49] Chen XQ, Lam KF, Zhang QJ, Pan BC, Arruebo M, Yeung KL. Synthesis of highly selective magnetic mesoporous adsorbent. Journal of Physical Chemistry C. 2009;113(22):9804-9813
- [50] Long Y, Xiao L, Cao Q. Co-polymerization of catechol and polyethylenimine on magnetic nanoparticles for efficient selective removal of anionic dyes from water. Powder Technology. 2017;310:24-34
- [51] Saksornchai E, Kavinchan J, Thongtem S, Thongtem T. Simple wet-chemical synthesis of superparamagnetic CTAB-modified magnetite nanoparticles using as adsorbents for anionic dye Congo red removal. Materials Letters. 2018;213:138-142
- [52] Faraji M, Shabanian M, Aryanasab F. Efficient removal of anionic dyes from aqueous media using newly in situ synthesized triazine-based nitrogen-rich network-modified magnetic nanoparticles. Journal of the Iranian Chemical Society. 2018;15:733-741
- [53] Sahraei R, Pour ZS, Ghaemy M. Novel magnetic bio-sorbent hydrogel beads based on modified gum tragacanth/graphene oxide: Removal of heavy metals and dyes from water. Journal of Cleaner Production. 2017;142:2973-2984
- [54] Ge L, Wang W, Peng Z, Tan F, Wang X, Chen J, Qiao X. Facile fabrication of Fe@MgO magnetic nanocomposites for efficient removal of heavy metal ion and dye from water. In: Powder Technology. 2017
- [55] Wu Y, Chen L, Long X, Zhang X, Pan B, Qian J. Multi-functional magnetic water purifier for disinfection and removal of dyes and metal ions with superior reusability. Journal of hazardous materials. 2018;347:160-167
- [56] Konicki W, Hełminiak A, Arabczyk W, Mijowska E. Removal of anionic dyes using magnetic Fe@graphite core-shell nanocomposite as an adsorbent from aqueous solutions. Journal of Colloid and Interface Science. 2017;497:155-164
- [57] Zhang YR, Su P, Huang J, Wang QR, Zhao BX. A magnetic nanomaterial modified with poly-lysine for efficient removal of anionic dyes from water. Chemical Engineering Journal. 2015;262:313-318

- [58] Bayramoglu G, Altintas B, Arica MY. Adsorption kinetics and thermodynamic parameters of cationic dyes from aqueous solutions by using a new strong cation-exchange resin. Chemical Engineering Journal. 2009;152(2):339-346
- [59] Ge F, Ye H, Li MM, Zhao BX. Efficient removal of cationic dyes from aqueous solution by polymer-modified magnetic nanoparticles. Chemical Engineering Journal. 2012;198:11-17
- [60] Yan H, Li H, Yang H, Li A, Cheng R. Removal of various cationic dyes from aqueous solutions using a kind of fully biodegradable magnetic composite microsphere. Chemical Engineering Journal. 2013;223:402-411
- [61] Chen Y, He F, Ren Y, Peng H, Huang K. Fabrication of chitosan/PAA multilayer onto magnetic microspheres by LbL method for removal of dyes. Chemical Engineering Journal. 2014;249:79-92
- [62] Amiri M, Salavati-Niasari M, Akbari A, Gholami T. Removal of malachite green (a toxic dye) from water by cobalt ferrite silica magnetic nanocomposite: Herbal and green solgel autocombustion synthesis. International Journal of Hydrogen Energy. 2017;42(39): 24846-24860
- [63] Li Q, Zhan Z, Jin S, Tan B. Wettable magnetic hypercrosslinked microporous nanoparticle as an efficient adsorbent for water treatment. Chemical Engineering Journal. 2017;326: 109-116
- [64] Singh KK, Senapati KK, Sarma, KC. Synthesis of superparamagnetic Fe₃O₄ nanoparticles coated with green tea polyphenols and their use for removal of dye pollutant from aqueous solution. Journal of environmental chemical engineering. 2017;5(3):2214-2221
- [65] Li C, Wang X, Meng D, Zhou L. Facile synthesis of low-cost magnetic biosorbent from peach gum polysaccharide for selective and efficient removal of cationic dyes. International journal of biological macromolecules. 2018;107:1871-1878
- [66] Attia TMS, Hu XL, Yin DQ. Synthesized magnetic nanoparticles coated zeolite for the adsorption of pharmaceutical compounds from aqueous solution using batch and column studies. Chemosphere. 2013;93(9):2076-2085
- [67] Reddy DHK, Yun YS. Spinel ferrite magnetic adsorbents: Alternative future materials for water purification? Coordination Chemistry Reviews. 2016;315:90-111
- [68] Nadim AH, Al-Ghobashy MA, Nebsen M, Shehata MA. Gallic acid magnetic nanoparticles for photocatalytic degradation of meloxicam: synthesis, characterization and application to pharmaceutical wastewater treatment. RSC Advances. 2015;5(127):104981-104990
- [69] Hayasi M, Saadatjo N. Preparation of magnetic nanoparticles functionalized with poly (styrene-2-acrylamido-2-methyl propanesulfonic acid) as novel adsorbents for removal of pharmaceuticals from aqueous solutions. Advances in Polymer Technology. 2017:1-13
- [70] Liu WT. Nanoparticles and their biological and environmental applications. Journal of Bioscience and Bioengineering. 2006;102(1):1-7

- [71] Shirazi E, Torabian A, Nabi-Bidhendi G. Carbamazepine removal from groundwater: Effectiveness of the TiO₂/UV, nanoparticulate zero-valent iron, and Fenton (nZVI/H₂O₂) processes. CLEAN – Soil, Air, Water. 2013;**41**:1062-1072
- [72] Ghauch A, Tuqan A, Assi HA. Antibiotic removal from water: Elimination of amoxicillin and ampicillin by microscale and nanoscale iron particles. Environmental Pollution. 2009; 157:1626-1635
- [73] Segura Y, Martinez F, Melero JA. Effective pharmaceutical wastewater degradation by Fenton oxidation with zero-valent iron. Applied Catalysis B: Environmental. 2013;**136**:64-69
- [74] Bautitz IR, Velosa AC, Nogueira RF. Zero valent iron mediated degradation of the pharmaceutical diazepam. Chemosphere. 2012;88:688-692
- [75] Huang YX, Keller AA. Magnetic nanoparticle adsorbents for emerging organic contaminants. ACS Sustainable Chemistry & Engineering. 2013;1:731-736
- [76] Bao X, Qiang Z, Ling W, Chang JH. Sonohydrothermal synthesis of MFe₂O₄ magnetic nanoparticles for adsorptive removal of tetracyclines from water. Separation and Purification Technology. 2013;117:104-110
- [77] Wan J, Deng HP, Shi J, Zhou L, Su T. Synthesized magnetic manganese ferrite nanoparticles on activated carbon for sulfamethoxazole removal. CLEAN – Soil, Air, Water. 2014;42: 1199-1207
- [78] Lu L, Li J, Yu J, Song P, Ng DHL. A hierarchically porous MgFe₂O₄/γ-Fe₂O₃ magnetic microspheres for efficient removals of dye and pharmaceutical from water. Chemical Engineering Journal. 2016;283:524-534
- [79] Wei H, Hu D, Su J, Li KB. Intensification of levofloxacin sono-degradation in a US/H₂O₂ system with Fe₃O₄ magnetic nanoparticles. Chinese Journal of Chemical Engineering. 2015;23:296-302
- [80] Zhao H, Cui H-J, Fu M-L. Synthesis of core-shell structured Fe₃O₄@α-MnO₂ microspheres for efficient catalytic degradation of ciprofloxacin. RSC Advances. 2014;4:39472
- [81] Madhavan J, Kumar PS, Anandan S, Zhou M, Grieser F, Ashokkumar M. Ultrasound assisted photocatalytic degradation of diclofenac in an aqueous environment. Chemosphere. 2010;80:747-752
- [82] Xu Y, Li C, Zhu X, HuangWE, Zhang D. Application of magnetic nanoparticles in drinking water purification. Environmental Engineering and Management Journal (EEMJ). 2014;13(8): 2023-2029
- [83] Zhang X, Wang W, Zhang Y, Zeng T, Jia C, Chang L. Loading Cu-doped magnesium oxide onto surface of magnetic nanoparticles to prepare magnetic disinfectant with enhanced antibacterial activity. Colloids and Surfaces B: Biointerfaces. 2018;161:433-441
- [84] Zhang Z, Xing D, Zhao X, Han X. Controllable synthesis Fe₃O₄@POHABA core-shell nanostructure as high-performance recyclable bifunctional magnetic antimicrobial agent. Environmental Science and Pollution Research. 2017;24(23):19011-19020

- [85] Rana SB, Singh RP. Investigation of structural, optical, magnetic properties and antibacterial activity of Ni-doped zinc oxide nanoparticles. Journal of Materials Science: Materials in Electronics. 2016;27(9):9346-9355
- [86] Shukla S, Arora V, Jadaun A, Kumar J, Singh N, Jain VK. Magnetic removal of Entamoeba cysts from water using chitosan oligosaccharide-coated iron oxide nanoparticles. International Journal of Nanomedicine. 2015;10:4901
- [87] Zhang X, Qian J, Pan B. Fabrication of novel magnetic nanoparticles of multifunctionality for water decontamination. Environmental Science & Technology. 2016;**50**(2):881-889
- [88] Zhan S, Yang Y, Shen Z, Shan J, Li Y, Yang S, Zhu D. Efficient removal of pathogenic bacteria and viruses by multifunctional amine-modified magnetic nanoparticles. Journal of Hazardous Materials. 2014;274:115-123
- [89] Park HH, Park S, Ko G, Woo K. Magnetic hybrid colloids decorated with Ag nanoparticles bite away bacteria and chemisorb viruses. Journal of Materials Chemistry B. 2013;1(21): 2701-2709
- [90] Ali Y, Samaneh R, Zohre R, Mostafa J. Magnetic water treatment in environmental management a review of the recent advances and future perspectives. Current World Environment. 2014;9(3):1008-1016

EPs Antibiotics: Photodecomposition and Biocarbon Adsorption

Nilce Ortiz

Additional information is available at the end of the chapter

http://dx.doi.org/10.5772/intechopen.76893

Abstract

The emerging pollutants (EPs) are considered a global monitoring challenge, present in the environment in very reduced concentrations, and the proper methods for sampling and analysis are still in development. Many published types of research considering the EPs identification only depend on the analytical methodology and a more efficient higher number of EPs. The quantitative determination of the prioritized EPs in water needs advanced and ultra-sensitive instrumental techniques applied in water, water-suspended matter, soil, and biota. The regulatory framework of the water-quality parameters does not often include the microplastics, EPs, and their metabolites; especially, the groundwater water-quality monitoring and control are urgent but not yet achieved. The EPs sources in water are the sewage, industrial, and agricultural waste discard, and the UN estimated that the wastewater produced annually is about 1500 km³, about six times more water than existing in all rivers of the word. In 2015, in China, which has a fast-growing economy, the water is a scarce resource with just 8% of the world's fresh water to meet the water needs of the 22% of the world's population.

Keywords: emerging pollutants, biocarbon, biochar, antibiotics, solar, photodecomposition

1. Introduction

In the world, about 33% of the industrial wastewater and 70% of the household sewage are untreated and directly discharged into rivers and lakes, while 80% of China cities have no sewage treatment facilities and water supplies in 90% of the cities, the water being contaminated [1]. There are 20 classes of categorized EPs accordingly with their origin; the prominent categories are pharmaceuticals (urban and protein production), pesticides (agriculture), disinfection

© 2018 The Author(s). Licensee IntechOpen. This chapter is distributed under the terms of the Creative IntechOpen Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.

by-products (urban and industry), wood preservation, and industrial chemical manufacturing. Different types of EPs show different properties as the organic substances divided in PBT for persistent bioaccumulative and toxic substances as POPs and persistent organic compounds. The EPs with more polarity are the pesticides, pharmaceuticals, and industrial chemicals. The inorganic compounds as poisonous metals are also found in polluted waters and finally the newly developed particulate contaminants as nanoparticles and microplastics [2].

The EPs monitoring and control are a huge problem, and the solution represents a significant challenge in sampling procedures and analytical techniques. The monitoring procedures do not cover all EPs with a potential concern being the highly known hazardous controlled, and the newly discovered contaminants as nanomaterials and microplastics are virtually analytically nonexistent. In the study, the EPs presence and toxicity are studied using bioindication; the most common organism is the *Gammarus pulex* as a model invertebrate from water and sediment.

The properties such as absorption behaviors of pharmaceuticals, for example, can vary vastly in different soil types on ionized and nonionized form, affecting the interaction of soil. Little information is available about the EPs dynamics in the water column, sediments, and the accumulation in the aquatic food chain and the loads from the agro-environment through diffuse pollution or from urban and industrial areas [3, 4].

The EPs decomposition products detection includes enantiomeric distribution of chiral compounds found in the environment, and their possible toxicological differences between enantiomers that are of concern. Such information for the risk assessment analysis is also considered [3].

Worldwide, the regulatory framework is under development to control the production and the discharge of the EPs into water resources, a complex set of regulations governing the production, commercialization, and emission to control the EPs presence in the environment and the drinking water (quality standards and monitoring specification). The European Union (EU) has a regulation plan to register, evaluate, authorize, and restrict the use of almost all EPs substances manufactured or imported to EU.

The agricultural activity is one of the most critical contributors to diffuse pollution in Europe, and such emissions are predicted to increase in the future. The agriculture activity is considered an essential business for regulating the chemical EPs use and emission discharge into the environment. Some research projects are under development, trying to treat, decompose, and remove those pollutants from the water resources.

The emerging pollutants are considered the potentially toxic chemicals present in low concentrations and many environmental compartments. They include pesticides, biocides, pharmaceuticals, industrial chemicals, and personal care products. The common entrance of these compounds in surface water resources is via untreated sewage discharge, the effluents of wastewater treatment plants (WWTPs), and from agricultural, urban, and street runoff. The organic pollutant water inputs usually occur continuously in low dosages or as peaks trigged by emission or runoff events. Such a behavior is particularly harmful to antibiotics environmental contamination, providing the optimized conditions for microorganism adaptation and increase in resistance.

The concept of chemical activity helps to understand the EPs environmental fate, distribution, quantification, and prediction of the ecological partitioning theory of the chemicals in aquatic

systems as water column, sediment (mostly organic matter), and biota (lipids and proteins). It addresses single compound or mixtures reaching the equilibrium. The chemical activity relies on partition coefficients which are only available as experimental values. There are some models calculating partition coefficients from octanol–water systems (Kow-based models), and they were used for decades to estimate the concentration in different compartments.

Nowadays, the relationship of the polyparameters linear-free energy is used to calculate the partition coefficients of biological and nonbiological matrices including lipids and different proteins [4, 5]. The model helps to describe the diffusion and partitioning processes and estimate when an environmental compartment acts as an EPs source or a sink for a single contaminant or a multi-compartment system.

2. EPs photodecomposition

Two of the main topics of growing concern in analytical chemistry are the development of green water treatment. TiO_2 has emerged as a promising photocatalyst for environmental cleanup applications; they have efficiently decomposed and removed a variety of pollutants promoting the generation of OH radicals using oxidation reactions with in situ active oxygen generated upon light irradiation. In water purification, photodegradation of contaminants in real water samples has become an important topic of research in the recent years.

There are some studies using electrocoagulation process to decompose EPs compounds. The process is an electrochemical introducing coagulants and removing suspended solids, colloidal material, and metals as well as other dissolved solids from water and wastewater eliminating pollutants, pesticides, and radionuclides. A direct current is applied, and one electrode is soluble into a solution which finally precipitates as oxides or hydroxides.

The environmental chemistry is the base for many treatment technologies of these pollutants, and the application of the adsorption process is one of the most used techniques. The results and comparison of different treatment technologies usually consider the initial concentration and the final concentration. The adsorbent materials are graphene oxide, clay mineral and biochar, nanocrystalline mineral, and arsenite using an enhanced coagulation process [6, 7]. The pharmaceuticals used iron chemical reduction reaction, and the advanced oxidation performed by ozone/UV also was used. The pharmaceuticals uses iron chemical reduction reaction, and the advanced oxidation performed by ozone /UV. The study of micropollutants biodegradation uses also a membrane.

The most common material used for EPs adsorption is the activated carbon (AC) with high porosity and surface area, and the use of AC shows the removal percentage higher than 90% for a wide variety of compounds bringing the residual concentration below the regulation limit. Other materials need more contact time for the same results. The advanced wastewater reclamation plant often uses the AC [5]. Many authors point out the high importance of the AC origin, depending on the initial crystalline structure of the biomass, the AC obtained from wood, vine, and olive waste, and coal showed the removal percentage always higher than 80% for antibiotics.

There are some adsorbents used for pharmaceutical removal as biochar, clay minerals, zeolites, Fe-Mn-binary oxide, graphene oxide, alumina, nanoscale iron, molecularly imprinted polymer, and carbon nanotubes [5, 8]. The phase extraction uses organic phase to remove organic contaminants, and the use of membrane technology is for pharmaceuticals removal from polluted waters. The bases of the membrane technology are the hydrostatic pressure to remove suspended solids, and solutes with a high molecular weight also classified as ultrafiltration, nanofiltration, microfiltration, forward osmosis, and reverse osmosis. The high removal percentages are obtained for forward and reverse osmosis with the removal percentages usually more than 95%.

The biological processes in conventional activated sludge decompose only the natural pharmaceutical compounds like caffeine, dichlorofenac, and trimethoprim. The advanced oxidation processes provide higher removal percentages associated with the hydroxyl radical production with removal percentages always higher than 96% including the sonochemical decomposition. The solar photo Fenton process obtains the removal percentage of 95–97.5% in just 20 min of reaction.

Many published results indicate that the degradation of EPs using a single treatment is not likely the best approach to treat and remove EPs from water, and the use of a combined technology can overcome deficiencies of individual technologies and be able to ply in complex mixtures of contaminants. The advanced oxidation processes are at present the most efficient degradation processed for EPs contamination.

The results of the kinetics studies of photodecomposition and biocarbon sorption provide valuable insights about the kinetics models: pseudo-first-order (Eq. (1)), pseudo-second-order (Eq. (2)), and intraparticle (Eq. (3)) with the determination of photodecomposition and adsorption rates on pseudo-first-order equation [9, 10].

$$\log(q_{e} - q_{t}) = \log(q_{e}) - \frac{K_{1}}{2.303}t$$
(1)

where K_1 is the pseudo-first-order rate (min⁻¹) and qe (mgg⁻¹) refers to the experimental adsorbed mass at equilibrium. The plotting of the calculated values of ln (qe-qt) for t (time) and the calculation of K_1 were used the slope values of the line equation.

Pseudo-second order equation:

$$\frac{t}{q_{t}} = \frac{1}{K_{2}} + \frac{1}{q_{c}}t$$
(2)

where K_2 (g.mg⁻¹.min⁻¹) is the kinetics adsorption rate, plotting the t/qt for t (min), and the calculation predicted the adsorption capacity qe (mg g⁻¹) and the integrated adsorption rate K_2 with the slope and the intercept of the line equation, respectively.

Intraparticle equation:

$$\log(q_t) = \log(K_{id}) + a\log(t)$$
(3)

The use of the experimental results allows performing the kinetics calculations [11–13], using Eqs. (1)–(3). The pseudo-first-order equation represents a logarithm of the reactant species

and the reaction time; larger K_1 indicates a fast reactant consumption and small time to complete the reaction; the R² values obtained for the pseudo-first equation indicate a lower correspondence between the results and the theory. Some published results show the solar photodecomposition processes with goethite as pseudo-first-order kinetics with K = 0.26 × 10^{-2} min⁻¹. Usually, the photodecomposition experimental results indicated a lower correlation with pseudo-first-order, sometimes just one K₁ value; all results showed a better correspondence with the pseudo-second order.

Considering the pseudo-second-order reaction, the sum of the exponents in the equation rate is equal to two for the plotted reactant concentrations with time. The pseudo-second-order response depends on the initial content, of the two different reactants A and B combining in a single elementary step. Before the rate, where A decreases, they can be expressed using the differential equation, and the linear equation can be rearranged, integrated, and followed where the slope value is K_2 . The pseudo-second order showed better correspondence with the experimental results corroborating with many published results for biocarbon adsorption and amoxicillin (AMX) removal treatments. The interparticle reaction usually points out the slow step of the adsorption reaction.

The use of integrated processes as solar/TiO₂ photodecomposition followed by adsorption has many advantages as an excellent potential for photocatalysis with the application of solar treatment chambers and possible self-cleaning surfaces. However, the practical applications and continuous use demand solutions to kinetics problems, and they may rise as the adsorbent-reduced surface area, TiO₂ oxidation surface, and solid low stability due to long-term use and the potential oxide mass production. The amoxicillin degradation with solar/TiO₂ anatase proceeds about three times faster than with ultraviolet (UV) lamp [14, 15]. The explanation of the disproportional improvement oxidation rates is the difference between the small spectrum irradiance of UV band and the broad spectrum of visible solar light. The intensity of radiation spectrum grows with an increasing wavelength from 300 to 500 nm. The combination of solar photodecomposition and the adsorption process is efficient and low cost.

3. The antibiotics photodecomposition products

The electrospray ionization mass spectrometry (ESI-MS) analytical technique measures the EPs methylene blue (MB) photodecomposition. Before the photodecomposition reaction, the methylene blue compound was m/z 284 (**Figure 1(a**)), and after 1 day of photodecomposition, there are several peaks (**Figure 1(b**)). Those peaks were MB fragment degradation compounds with m/z values of 109, 129, and 165. Those peaks have the relative intensity of 37.3, 44.2, and 40.5% considering the original MB peak of 100% with m/z 284 (**Figure 1(a**)).

The EPs dye photosensitization process involves the dye initial electronic excitation D to D^{*} induced by hv incident radiation energy which ejects one electron in the semiconductor (SC) conduction band [8, 16, 17]. The emitted electron reacts with the environment oxygen oxidizing the radial D^{*}, and the total process results in colorless products, Eq. (6).



Figure 1. ESI-MS spectra of methylene blue photodecomposition (a) before and (b) after degradation of 1 day (adapted from Kobzi et al. [27]) [8].

The study provides some example of dye sensitization, and the photosensitized bleaching occurring under visible light and an aerated aqueous dispersion of TiO_2 . The light absorbed by the dye alone was able to promote its rapid and complete photobleaching. The photonic efficiency η for the visible photobleaching of the MB is already known, and it is a function of the incident radiation wavelength. The spectrum η versus λ plot was similar in shape to the diffuse reflectance spectrum (DRS) of the MB adsorbed on TiO₂ with a peak at 665 nm which is the wavelength of the maximum adsorption for MB.

By contrast, the dye photolysis involves the electronically excited state of the dye which is either unstable and quenched by ambient O_2 to produce singlet oxygen which oxidizes the dye. Considering only the equation, it is not surprising if the rate of dye bleaching due to photolysis is often unchanged in the absence of the semiconductor. When the photolysis and photosensitization are examined, the presence of SC enhances the production of the bleached product.

$$D + hv'' \rightarrow D^* (+O_2) \rightarrow Bleached products$$
 (4)

The rapid and irreversible reduction of the photogenerated holes by glycerol followed the reduction reaction and the molar absorptivity of 4.7000 M⁻¹ cm⁻¹ at 620 nm. The glycerol behaves as an SED which is also present in the ink film. The MB photocatalytic process runs as follows:

The use of photocatalyst indicator (paiis) helps to measure the electron generation efficiency of the TiO₂ changing from blue redox dye Resazurin (Rz) to pink Resorufin (Rf). The Rz as

the MB adsorbs little in the UVA region. The elimination of the photogenerated holes by the glycerol, which acts as an SED, followed the reduction reaction.

Photocatalyzed MB redox reaction:

The colorless species identified as Leuco methylene blue (LMB) is readily re-oxidized to MB by ambient O_2 and has to be produced and maintained in anaerobic conditions to be stable, **Figure 2**. This environmental condition also ensures that the traditional dye-sensitized bleaching reaction cannot occur [19, 20].

The MB is excited under the red region of the light source and received electrons to the excited MB which produces an anion which abstracts a proton from water to provide LMB; the LMB is very readily re-oxidized back to MB if there is any oxygen in the solution [9, 10, 21]. The CG results showed no change in the MB concentration for the first few minutes of reaction, but if the reaction progresses, efficiently all molecules of MB will be converted to LMB. Subsequently, the visual observation indicates that the intense blue color of the solution starts to disappear as the reaction progresses as an effect of the photoreduction of the MB at 365 nm in the absence, and the presence of glycerol electron donor occurs, and the MB rapidly photobleached via an oxidative route with LMB primary production in 15 min. The air presence also quickly reduced the LMB to MB, gaining 92% of the original color.



Figure 2. Scheme for photodecomposition mechanism of MB and the Leuco MB (LMB) production [9, 10].

The photocatalytic dye bleaching process using the $\lambda = 617$ nm is much slower than the reaction mediated by TiO₂ and $\lambda = 365$ nm light. The MB/glycerol has the potential as the dye-based test for UV presence, and visible absorbing photocatalytic materials provided by photocatalyst test.

The redox potential for the reduction of MB at pH 7 to LMB is 0.011 v, whereas for oxygen it depends on electron transference and how many are present. For 4e-, O_2/H_2O (0.815 v) and for 1 electron O_2/O_2 - (-0.33 v). It is not obvious that an SC is capable of reducing MB to LMB and also will be able to reduce O_2 . The oxi-reduction indicator dyes are important to visualize the redox chemistry and measure the dye environmental oxidation/reduction.

The environmental antibiotics resistance to widely used medical and veterinary medicine is a serious problem and poses a significant threat to the health of humans and livestock infected with resistant bacterial strains. The alarming fact is that resistant genes can be mobilized between various environmental compartments and transferred into the food chain. The conventional processes associated with sewage treatment, hydrolysis, biodegradation, and sorption are ineffective for the removal of many antibiotics, and thus the photodegradation may be the predominant transformation pathway for antibiotics deactivation in the environment.

The UV spectra of amoxicillin (AMX) and the amoxicillin decomposition products (ADPs) as ADP1,2, ADP4,5, and ADP8,9 consisted of two peaks at λ = 230 and 275 nm, similar with AMX. This effect is due to the para-substituted phenolic group, which do not change in the AMX and ADPs skeleton (**Figures 3** and **4**) and is the primary contributor to the observed UV spectra. Those peaks enable a quantitative calculation of these ADPs in the environment based on the assumption that the UV RF is relatively similar for AMX and its ADPs at λ = 230 nm.



Figure 3. Suggested photodegradation pathway of amoxicillin in an aqueous medium (adapted from Gozlan et al., 2010) [7].



Figure 4. UV spectra of (a) amoxicillin and degradation products ADP1,2, ADP4,5, and ADP8,9 and (b) the degradation product ADP6 [11].

On the other hand, the compound ADP^{"6} has conjugated double bonds (**Figure 3**), and thus, its UV spectrum consists of two different peaks at λ = 240 and 250 nm. The calculation and quantification of ADP could not be in the environment due to the incompatibility of the AMX and ADP6 UV spectrum at λ = 230 nm.

The confirmation of the proposed AMX DP structures was by comparison of the H NMR spectra of the AMX and those of purified ADP. **Figure 4** presents the degradation schematics of the AMX to its ADPs, the products ADP1,2, ADP4,5, and ADP8,9 are epimers, and preparative purification was carried out separately for each.

The penicilloic acid epimers (ADP4,5) are rapidly isomerized and become stable at an isomer ratio of 3:2, so the NMR spectra could not be obtained separately, **Table 1**. The isomers spectrum showed that H in the AMX spectrum is δ = 5.44 ppm, consisting of two hydrogen atoms, which shift to a higher field and become H-6a and H6b δ = 5.18 and 4.46 ppm, the ADP3 and ADP 8,9.

In recent years, the water-quality monitoring results indicated the presence of antibiotics and antibiotics residues in aquatic environments in many countries, including Europe, North America, and Asia. In addition, the antibiotic-resistant bacteria have become a serious problem worldwide, caused by the excessive use and incorrect discharge of antibiotics into the environment.

The use of ampicillin (AMP) is worldwide, as an essential antibiotic but this organic molecule rapidly decomposes in water containing bivalent cations as Ca^{2+} and Mg^{2+} and their

Hydrogen	AMX	ADP1	ADP2	ADP3	ADP4	ADP5	ADP6	ADP7	ADP8
CH ₃	136	0.95	1.33	1.20	1.37	1.26		131	1.28
	1.37	1.53	1.39	1.57	1.53	1.62		1.65	1.66
H-4	4.09	3.97	3.67	4.31	4.00	4.09		3.69	3.69
H-17	5.07	5.18	5.10	4.99	5.09	5.09		4.91	4.95
H-2	5.44	5.17	5.37	5.25	4.95	5.05		5.17	5.15
H-6a	5.44	5.18	4.46	5.90	3.54	3.70		3.90	4.39
H-6b					3.87	3.83			
H-21,23	6.92	6.96	6.96	6.94	6.96	6.96	6.81	6.82	6.82
H-20,24	7.31	7.40	7.40	7.31	7.35	7.35	8.13	7.18	7.18
ADP6							7.24		
ADP6							7.45		

Table 1. The RMN results for amoxicillin photodecomposition and byproducts [11].

detection in the environment is difficult. The development of analytical technique allows the identification and quantification of the ampicillin and ampicillin degradation by-products as 2-hydroxy-3-phenylpyrazine (HPP) in the environment. The results indicate their presence in 42–79% of the monitored rivers and household ponds but they were not detected in aquaculture ponds. In these locations, the HPP concentrations were in the range of 1.3–413.3 ngL⁻¹.The research results indicate the HPP presence as a promising marker for AMP presence and other β -lactam antibiotics with AMP substructure into the environment [22, 24, 25] (**Figures 5** and **6**).



Figure 5. Amoxicillin chemical structure.

The β -lactams antibiotics such as ampicillin, penicillins, cephalosporins, and oxicephalosporins inhibit biosynthesis of the bacterial cell wall by acylating and thereby inactivating transpeptidases. The antibacterial activity depends on the β -lactams rings, and some complex



Figure 6. The photodecomposition scheme for ampicillin and its by-products.



Figure 7. The complex formation with Cu-ampicillin chemical structure as a complex ligand.

compounds are responsible for their deactivation. There are several possible sites of the coordination for the metal ions on penicillin. The nitrogen and the double oxygen bonding are the most probable [13, 24, 26].

The amoxicillin and many antibiotics can act as a complex ligand with several possible sites for metal ions like Cu (II), Zn (II), and Cd(II) for coordination on penicillin. In the study, the kinetics and thermodynamics calculations are easy to find and how the neighboring group affects the probable complexation sites, of metals and the C=O of the antibiotic β -lactamic group and the NH amido penicillin group. The two most probable structures are shown in **Figure 7**. They are not excluding each other. For cephalexin and Cu²⁺, the interaction takes place through the side chain, the same as expected for ampicillin structure.

The metal ligand binary constants at temperature 37° C and ionic strength for ampicillin and Co^{2+} are = 3.12, for Ni²⁺ = 3.66, for Cu²⁺ = 4.79 and for Zn²⁺ = 2.98. Usually, the complex constants with Cooper are more stable [14, 23].

4. Binary and ternary antibiotics mixture

The TiO₂ is steady, inexpensive photocatalyst, and widely applied for removal and decompose organic pollutants. However, the poor selectivity of the TiO₂ is not conducive to remove high toxic contaminants from mixed solution, mostly in the presence of other pollutants [15, 16]. Some studies use tetracycline (TC) as a molecular template and the TiO₂ fly ash cenospheres as a supporter, the synthesis of the molecularly imprinted photocatalyst (MIP). Such a material possesses the specific recognition ability toward tetracycline (TC) using surface-imprinting technology and the photo-induced method. The cenospheres hollow spherical structure has the diameter from 90 to 120 μ m, and the degradation process of photocatalytic activity of MIP with 20 mgL⁻¹ of TC under visible light radiation reached the photodegradation of TC in the ternary solutions containing TC, oxytetracycline (Oxy), and ciprofloxacin (CIP). The coefficients of selectivity of degradation from TC, Oxy, and CIP were 1.67 and 1.25, respectively. The photodegradation mechanism of TC analyzed by mass spectrum (MS) indicated the TC decomposition step by step, resulting in CO₂, H₂O, and other gases.

A large number of antibiotics and their residues lead to the environmental emergence as a threat to indigenous microbial populations. The tetracycline (TC) ranks second in the global production and use. In spite of its consumption, environmental TC residues are very low (μ gL⁻¹ or nanogram L⁻¹), but they are resilient, and the TC residues may cause a series of ecological environmental and human health effects, such as promoting the resistant bacteria. The photocatalytic with solar energy is green technology and capable of decomposing the organic pollutant to a nontoxic compound.

The molecular imprinting is a versatile and straightforward method for the preparation of robust materials which can recognize the specific target in secondary and tertiary systems. Then, stability, the ease of development, and low cost make the molecularly imprinted particularly

attractive. Two or more antibiotics always coexist in the polluted aquatic environment. Thus, it is essential and meaningful to discuss the multicomponent antibiotics solution.

Published works related a variety of antibiotics in polluted soil and water environment. Therefore, it is of great significance to explore the ecological risk of the combined exposure to various antibiotics. The mixture of different types of antibiotics may lead to varying joint effects on the bacteria's, synergistic, additive, and some antagonistic effects.

Some published works indicate that antibiotics mixtures present synergetic effects and others antagonistic effects. The presence of sulfonamides (SA), as potentiator effects (SAP), and tetracycline's (TC) was investigated for binary and tertiary mixture toxicity. The mixtures of SA-SAP and SA-SAP-TC presented a synergetic impact on bacteria tests, while SA-TC and SAP-TC showed antagonistic effects. The TC presence in ternary mixtures altered the toxic ratio of SA and SAP, which lead to the various joint effects of the ternary mixtures on different bacteria populations.

5. The biocarbon: adsorbent uses

Since the beginning, the powder biocarbon in Brazil was considered a worthless material, always related to airborne pollution, infant and slavery condition work, and work-related disease. Nowadays, a group of researchers is trying to restore the biocarbon industrial use as an essential additive for agroindustry on soil amendment with possibility of enhancing the organic matter and water content retention.

The biocarbon (biochar) has been used to increase the amount of organic matter in the agricultural soil. There are many benefits in this application, resulting in improved soil fertility, nutrient content, water retention, better physical structure, and improved microbial activity. Intense agricultural operation tends to reduce the amount of organic matter present in the soil.

The application of biocarbon can be decisive in the semi-arid region of the northeast region with low rainwater retention, and such water source is scarce and used in a short period of the year. Soon, the agriculturists of these areas will be able to produce the own bio-carbon utilizing the agriculture biomass.

The biocarbon organic matter composition allows its use as a slow-release fertilizer such as adsorbent properties promoting the addition, retaining and the release of nutrients. The addition of some plastic agents like stack gel and bentonite clay helps the nutrient releasing rate control during the pellets formation. The biocarbon is a renewable organic matter source and provides phosphate and nitrate from wastewater adsorption treatment with nutrient retention for further agriculture use as slow-release fertilizer.

Biocarbon produced in a high-temperature pyrolysis application on heavy metals retention such as cadmium and zinc is studied. In some cases, it has been used in the recovery of soils with different levels of cadmium, zinc, and leads contaminated by the mining industry, reducing the concentration of toxic metals in plants such as beans. Biocarbon as an alternative adsorption matrix for water treatment and emerging pollutants removal has been confirmed. The biocarbon usually shows better environmental results in comparison with activated carbon, and also in many cases, it provides better potential energy supplied for lignocellulosic materials.

The biochar has various environmental applications like pollutant removal, carbon sequestration, and soil amendment. It has unique properties which makes it an efficient, cost-effective, and environmentally friendly material for contaminant removal. The different physical-chemical properties of the surface are microporosity and pH that can maximize its efficiency to various environmental applications. The research updates related to the pollutants interaction with surface functional groups of biochar and the effect of the parameters variability in biochar attribute to specific pollutants removal, involved mechanisms, and efficiency for these removals.

Emerging pollutants (EPs) include agrochemicals, antibiotics, polycyclic aromatic hydrocarbons (PAHS), polychlorinated biphenyls (PCBs), volatile organic compound (VOC), aromatic dyes, toxic metals, ammonia, nitrate, phosphate, sulfite from aqueous, gaseous, and solid phases. There is also the possibility of the biochar-tailoring properties to improve their removal efficiency for organic–inorganic contaminants [20]. The soil application not only remediates but improves soil properties as water-holding capacity, O2 content, and moisture level.

The removal of aromatic dyes, for example, rhodamine, methyl violet, and methyl blue by anionic biochar, is mainly involved in electrostatic attraction/repulsion interactions [17, 18, 25]. In these sorption mechanisms, highly polar biochar pyrolyzed at = <400°C contained aromatic Pi-configuration and electron donor and acceptor functional groups [20]. These π electron-rich biochar functional groups (-Ve) have electron donor-acceptor interactions with π electron-deficient (+Ve) organic dyes. Hence, these interactions have resulted in an electrostatic repulsion and promoted pollutant adsorption via H-bonding between biochar and apolar dyes [26].

Crop residues as peanut, canola, soybean straw, rice hull, and so on derived biochar at a temperature of <400°C contained more O- and H-bearing functional groups. Thus, such a material exhibited a higher sorption capacity for apolar methyl violet and methylene blue due to pH change through biochar amendment [17, 24]. At higher pH, the net negative charge on biochar surface (due to dissociation of phenolic OH groups) increased the electrostatic interactions with methyl violet, whereas at lower pH, the π - π electron donor-acceptor interactions increased, thus improving the H-bonding for methylene blue sorption.

At zero point charge (ZPC), biochar does not have any surface charge, and it develops surface charge dependent on pH. The pH below the biochar ZPC (8.17, 8.52 and 8.79) comprised positive charge and sorbs less methylene blue. Whereas the pH higher of the biochars ZPC comprise negative surface charge increasing the methylene blue adsorption due high electrostatic attractions.

Likewise, the sorption of polar antibiotics sulfamethazine (SMZ) by hardwood/softwoodderived biochars (produced at 300–700°C has pH-dependent interactions. At higher pH, H bonding occurs between anionic SMZ and COOH or OH group biochar. At lower and neutral pH, the π - π electron donor-acceptor interactions and cation exchange are dominant mechanisms between biochar and SMZ [23, 26]. These studies confirmed that pH is the most crucial factor for biochar interactions with polar organic pollutants.

The integrated process of photodecomposition followed by adsorption study includes the adsorption isotherms, performing the calculations of Langmuir, Freundlich, and Redlich-Peterson (R-P) isotherms, Eqs. (7)–(9), respectively. The Langmuir isotherm adsorption assumes an ideal solid surface composed by a series of distinct sites capable of binding the adsorbate in a molecular coverage; the chemical reaction between the adsorbate molecule and the surface is a pseudo-second-order reaction. The Freundlich isotherm is empirical but widely used, and the value of n is a measure of the adsorption intensity higher than 1, where the adsorption processes are more favorable. The Redlich-Peterson (R-P) is more accurate than the Langmuir and Freundlich due the "g" value equal to 1. Usually, the R-P is by Langmuir and Freundlich isotherm equations; such observed behaviors were also studied. The error calculation will help to point out better isotherm adjustment

$$Ce/qe = 1/Q_0 b + Ce/Q_{0'}$$
 (7)

$$\log qe = \log Kf + 1/n \log Ce$$
(8)

$$\ln (Ce/qe) = g \ln Ce - \ln Kr$$
(9)

where Ce = equilibrium concentration (mgL⁻¹), qe = the amount adsorbed at equilibrium (mg.g⁻¹), Q₀ and b are Langmuir constants, Q₀ indicates the adsorption capacity of the material, and b indicates the energy of adsorption. Kf and n are Freundlich constants. Kf indicates the adsorption capacity of the material and n indicates the efficiency of adsorption. Kr and g are Redlich-Peterson constants; Kr indicates the adsorption capacity and "g" is the exponent between 0 and 1.

The RL values were in the interval from 0 to 1, with favorable adsorption accordingly with Langmuir isotherm. The Freundlich isotherm constant n was also in the range of 2 < n < 10; the indication of the agreement with Freundlich model with equal adsorption heating and Redlich-Peterson parameters were also promising.

6. Conclusion

The emerging pollutants are considered potentially toxic chemicals present in low concentrations and many environmental compartments. They include pesticides, biocides, pharmaceuticals, industrial chemicals, and personal care products. The common entrance of these compounds in surface water resources is via untreated sewage discharge, the effluents of wastewater treatment plants (WWTPs), and from agricultural, urban, and street runoff. The organic pollutant water inputs usually occur continuously in low dosages or as peaks trigged by emission or runoff events. Such a behavior is particularly harmful to antibiotics environmental contamination, providing the optimized conditions for microorganism adaptation and increase in resistance. The concept of chemical activity helps to understand the EPs environmental fate, distribution, quantification, and prediction of the ecological partitioning theory of the chemicals in aquatic systems as water column, sediment (mostly organic matter), and biota (lipids and proteins) address single compound or mixtures reaching the equilibrium. The chemical activity relies on partition coefficients which are only available as experimental values. TiO, has emerged as a promising photocatalyst for environmental cleanup applications; they have efficiently decomposed and removed a variety of pollutants, promoting the generation of OH radicals using oxidation reactions with in situ active oxygen generated upon light irradiation. In water purification, photodegradation of contaminants in real water samples has become an important topic of research in recent years. The results of the kinetics studies of photodecomposition and carbon sorption provide valuable insights about the kinetics models: pseudo-first-order (Eq. (1)), pseudo-second-order (Eq. (2)), and intraparticle with the determination of photodecomposition and adsorption rates. The ESI-MS analytical technique allows the measurement of the EPs methylene blue photodecomposition. Before the photodecomposition reaction, the methylene blue compound was only m/z 284, and after 1 day of photodecomposition, there are several peaks.

The UV spectra of amoxicillin (AMX) and the amoxicillin decomposition products (ADPs) as ADP1,2, ADP4,5, and ADP8,9 consisted of two peaks at λ = 230 and 275 nm, similar with AMX. This effect is due to the para-substituted phenolic group, which do not change in the AMX and ADPs skeleton and is the primary contributor to the observed UV spectra. The development of analytical technique allows the identification and quantification of the ampicillin and ampicillin degradation by-products as 2-hydroxy-3-phenylpyrazine (HPP) in the environment. The results indicate their presence in 42-79% of the monitored rivers and household ponds. In these locations, the HPP concentrations were in the range of 1.3–413.3 ngL⁻¹. The amoxicillin and many antibiotics can act as a complex ligand with several possible sites for metal ions like Cu (II), Zn (II), and Cd (II) for coordination on penicillin. The kinetics and thermodynamics calculations indicate neighboring group effects of the probable complexation sites, of metals and the C=O of the antibiotic β -lactamic group and the NH amido penicillin group. Published works related to a variety of antibiotics in polluted soil and water environment. Therefore, it is of great significance to explore the ecological risk of the combined exposure to various antibiotics. The mixture of different types of antibiotics may lead to varying joint effects on the bacteria, synergistic, additive, and some antagonistic effects. The biocarbon acts as an alternative adsorption matrix for water treatment and emerging pollutants removal, usually showing better environmental results in comparison with activated carbon, and also in many cases, it provides better potential energy supplied for lignocellulosic materials.

Acknowledgements

FAPESP (São Paulo Research Foundation) and CNPq (National Council of Technological and Scientific Development).

Author details

Nilce Ortiz

Address all correspondence to: nortizbr@gmail.com

Institute for Nuclear and Energy Researches-IPEN, São Paulo, Brazil

References

- Geissen V, Mol H, Klumpp E, Umlauf G, Nadal M, Ploeg MV, Zee SEATM, Ritsema CJ. Emerging pollutants in the environment: A challenge for water resource management. International Soil and Water Conservation Research. 2015;3:57-65 Available from: http://creativecommons.org/licenses/by-nc-nd/4.0/
- [2] UN W WAP. Water for People, Water for Life (2003): 3rd World Water Forum in Kyoto, Japan. 2003. Available from: www.orman-network.net
- [3] Petrie B, Barden R, Kasprzyk-Hordern B. A review on emerging contaminants in wastewaters and the environment: Current knowledge, understudied areas, and recommendations for future monitoring. Water Research. 2015;72:3-27. DOI: 101016/j.watres.2014. 08.053
- [4] Shihomatu HM, Martins EAJ, Cotrim MEB, Lebre DT, Ortiz N, Pires MAF. Guarapiranga reservoirs – Pharmaceuticals and historical urban occupation in a water resource. Journal of Geoscience and Environmental Protection. 2017;5:1-17
- [5] Inostroza PA, Massei R, Wild R, Krauss M, Brack W. Chemical activity and distribution of emerging pollutants: Insights from a multi-compartment analysis of the freshwater system. Environmental Pollution. 2017;231:339-347. DOI: 10.1016/j.envpol.2017.08.015
- [6] Ray SK, Dhakal D, Kshetri YK, Lee SW. Cu-alfa-NiMoO₄ photocatalyst for degradation of methylene blue with pathways and antibacterial performance. Journal of Photochemistry and Photobiology A: Chemistry. 2017;**348**:18-32
- [7] Gozlan I, Rotstein A, Avisar D. Amoxicillin degradation products formed under controlled environmental conditions: Identification and determination in the aquatic environment. Chemosphere. 2013;**91**:985-992. DOI: 10.1016/j.chemosphere.2013.01095
- [8] Qiu Y, Zheng Z, Zhou Z, Sheng D. Effectiveness and mechanism of dye adsorption on straw-based biochar. Bioresource Technology. 2009;100:5348-5351. DOI: 10.1016/j.biortech. 2009.05.054
- [9] Lee S, Mills A, Weels N. Assessing photocatalytic activity sing methylene blue without dye sensitization. Catalysis Today, 2017. https://doi.org/10.1016/j.cattod.2017.11.011

- [10] Izadifard M, Langford CH, Achari G. Photocatalytic dechlorination of PCB 138 using leuco-methylene blue and visible light – Reaction conditions and mechanisms. Journal of Hazardous Materials. 2010;181:393-398. DOI: 10.1016/j.jhazmat.2010.05.023
- [11] Oliveira FR, Patel AK, Jaisi DP, Adhikari S, Lu H, Khanal SK. Environmental applications of biochar: Current status and perspectives. Bioresource Technology. 2017;246:110-122. DOI: 101016/j.biortech.2017.08.122
- [12] Ortiz N, Silva A, Lima GNS, hyppolito FP. Using solar-TiO₂ and biocarbon to decompose and adsorb amoxicillin from polluted waters. International Journal of Chemistry. 2018;10:131-136
- [13] Sy NV, Harada K, Asayama M, Warisaya M, Dung LH, Sumimura Y, Diep KT, Há LV, Thang NN, Hoa TTT, Phu TM, Khai PN, Phuuong NT, Tuyen LD, Yamamoto Y, Hirata K. Residues of 2-hydroxy-3-phenylpyrazine, a degradation product of some β-lactam antibiotics, in environmental water in Vietnam. Chemosphere. 2017;172:355-362. DOI: 10.1016/J.Chemosphere.2016.12.156
- [14] Lozano MJ, Borrás J. Antibiotic as ligant. Coordinating behavior of the cephalexin towards Zn II and Cd II ions. Journal of Inorganic Biochemistry. 1987;31:187-195
- [15] Mukherjee G, Ghosh T. Metal ion interaction with penicillins- part VII: Mixed ligand complex formation of cobalt(II), nickel(II), copper(II), and zinc(II) with ampicillin and nucleic bases. Journal of Inorganic Biochemistry. 1995;59:827-833
- [16] Lu Z, Huo P, Luo Y, Liu X, Wu D, Gao X, Li C, Yan Y. Performance of molecularly imprinted photocatalysts based on fly ash cenospheres for selective photodegradation of single ternary antibiotics solution. Journal of Molecular Catalysis A: Chemical. 2013; 378:91-98
- [17] Li G, Zhu W, Zang C, Zhang S, Liu L, Zhu L, Zhao W. Effect of a magnetic field on the adsorptive removal of methylene blue onto wheat straw biochar. Bioresource Technology. 2016:16-22. http://dx.doi.org/10.1016/j.biortech.2015.12.087
- [18] Xu R, Xiao S, Yuan J, Zhao A. Adsorption of methyl violet from aqueous solutions by the biochars derived from crop residues. Bioresource Technology. 2011;102:10293-10298. DOI: 10.1016/j.biortec.2011.08.089
- [19] Nunes RM, Costa D, Ortiz N. The use of eucalyptus activated biocarbon for water treatment-adsorption processes. American Journal of Analytical Chemistry. 2017;8:515-522. DOI: 10.4236/ajac.2017.88037
- [20] Teixido M, Pignatello JJ, Beltran JL, Granados M, Peccia J. Speciation of the ionizable antibiotic sulfamethazine on black carbon (biochar). Environmental Science Technology. 2011;45:10020-10027. DOI: 10.1021/es202487h
- [21] Guilhen SN, Ortiz N, Fungaro DA, Masek O. Pyrolitic temperature evaluation of macauba biochar for uranium adsorption from aqueous solutions. In: Proceedings of Engineering Conferences International – Biochar: Production, Characterization and Applications. 2017 ECI Digital Archives: http://dc.engconfintl.org/biochar/ consulted on May 8th, 2018
- [22] Sun K, Keiluweit M, Kleber M, Pan Z, Xing B. Sorption of fluorinated herbicides to plant biomass derived biochars as a function of molecular structure. Bioresource Technology. 2011:9897-9903
- [23] Rodriguez-Narvaez OM, Peralta-Hernandez JM, Goonetilleke A, Bandala ER. Treatment technologies for emerging contaminants in water: A review. Chemical Engineering Journal. 2017;323:361-380. DOI: 10.1016/j.cej.2017.04.106
- [24] Ahmed MB, Zhou JL, Ngo HH, Guo W, Chen M. Progress in the preparation and application of modified biochar for improved contaminant removal from water and wastewater. Bioresource Technology. 2016;214:836-851. DOI: 10.1016/j.biortech.2016.05.057
- [25] Vithanage M, Rajapaksha AU, Tang X, Thiele-Bruhn S, Kim KH, Lee S, Ok YS. Sorption and transport of sulfamethazine in agricultural soils amended with invasive–plantderived biochar. Journal of Environmental Management. 141:95-103. DOI: 10.1016/j. jenvman.2014.02.030
- [26] Wang D, Wu X, Lin Z, Ding Y. A comparative study of the binary and ternary mixture toxicity of antibiotics towards three bacteria based on QSAR investigation. Environmental Research. 2018;162:127-134. DOI: 10.1016/j.envres.2017.12.015
- [27] Kobzi B, Watanabe Y, Akiyama K, Kuzmann E, Homonnay Z, Krehula S, Ristic M, Nishida T, Kubuki S. 57 Fe Mossbauer study and methylene blue decomposition effect of nanoparticle mixtures composed of metallic iron and maghemite. Journal of Alloys and Compounds. 2017;722:94-100. DOI: 10.1016/j.jallcom.2017.06.083



Edited by Sonia Soloneski and Marcelo L. Larramendy

This edited book, *Emerging Pollutants - Some Strategies for the Quality Preservation of Our Environment*, contains a series of chapters providing some strategies for the preservation of our environmental quality focusing on the different categories of environmental pollutants and their negative consequences on living organisms.

Published in London, UK © 2018 IntechOpen © Paul-Daniel Florea / iStock

IntechOpen



